ORAL ARGUMENT NOT YET SCHEDULED

No. 12-1100 (and consolidated cases)

IN THE UNITED STATES COURT OF APPEALS FOR THE DISTRICT OF COLUMBIA CIRCUIT

WHITE STALLION ENERGY CENTER, LLC, et al., Petitioners,

V.

UNITED STATES ENVIRONMENTAL PROTECTION AGENCY, Respondent.

On Petition for Review of Final Agency Action 77 Fed. Reg. 9304 (Feb. 16, 2012)

JOINT BRIEF OF INDUSTRY INTERVENORS IN RESPONSE TO ENVIRONMENTAL PETITIONERS

F. William Brownell
Lauren E. Freeman
Lee B. Zeugin
Elizabeth L. Horner
HUNTON & WILLIAMS LLP
2200 Pennsylvania Avenue, N.W.
Washington, D.C. 20037
(202) 955-1500
bbrownell@hunton.com
lfreeman@hunton.com
lzeugin@hunton.com
ehorner@hunton.com

Counsel for Utility Air Regulatory Group

Dated: February 21, 2013

Peter S. Glaser George Y. Sugiyama TROUTMAN SANDERS LLP 401 Ninth Street, N.W., Suite 1000 Washington, D.C. 20001 (202) 274-2998 peter.glaser@troutmansanders.com

Counsel for National Mining
Association, Peabody Energy
Corporation, and Sunflower Electric
Power Corporation

Additional counsel listed on following pages

Jeremy C. Marwell VINSON & ELKINS LLP 2200 Pennsylvania Avenue, N.W. Suite 500 West Washington, D.C. 20037 (202) 639-6507 jmarwell@yelaw.com

Eric A. Groten VINSON & ELKINS LLP 2801 Via Fortuna, Suite 100 Austin, TX 78746-7568 (512) 542-8709 egroten@yelaw.com

Counsel for White Stallion Energy Center, LLC

Jeffrey R. Holmstead Sandra Y. Snyder BRACEWELL & GIULIANI LLP 2000 K Street, N.W., Suite 500 Washington, D.C. 20006-1872 (202) 828-5800 Jeffrey.Holmstead@bgllp.com Sandra.Snyder@bgllp.com

Counsel for Tri-State Generation and Transmission Association, Inc.

David B. Rivkin, Jr.
Lee A. Casey
Mark W. DeLaquil
Andrew M. Grossman
BAKER & HOSTETLER LLP
Washington Square, Suite 1100
1050 Connecticut Avenue, N.W.
Washington, D.C. 20036-5304
(202) 861-1731
drivkin@bakerlaw.com

Counsel for National Black Chamber of Commerce and Institute for Liberty

CERTIFICATE AS TO PARTIES, RULINGS, AND RELATED CASES

Pursuant to Circuit Rule 28(a)(1), Industry Intervenor-Respondents state as follows:

A. Parties, Intervenors, and Amici

With the following exceptions, the parties, intervenors and amici in this action are those set forth in the certificate filed October 23, 2012 with the Joint Brief of State, Industry, and Labor Petitioners (Doc. No. 1401252). First, on December 6, 2012, the Court granted the motion of EcoPower Solutions (USA) Corporation to dismiss its petition for review (No. 12-1170) (Doc. No. 1408727). Second, on January 16, 2013, the Court granted the motion of the Utility Air Regulatory Group and Oak Grove Management Company LLC to intervene as respondents in Nos. 12-1174 and 12-1194 (Doc. No. 1415430). Third, on February 11, 2013, the Court granted the motion of the American Thoracic Society, American College of Preventive Medicine, American College of Occupational and Environmental Medicine, National Association for the Medical Direction of Respiratory Care, American College of Chest Physicians, William W. Buzbee, Jody Freeman, Oliver A. Houck, Richard J. Lazarus, Robert V. Percival, and Zygmunt J.B. Plater to participate as *amici curiae* in support of respondent in No. 12-1100 (Doc. No. 1419810).

B. Rulings Under Review

These petitions challenge EPA's final rule, "National Emission Standards for Hazardous Air Pollutants From Coal- and Oil-Fired Electric Utility Steam Generating Units," 77 Fed. Reg. 9304 (Feb. 16, 2012).

C. Related Cases

Each of the petitions for review consolidated under No. 12-1100 is related. These cases consist of Nos. 12-1101, 12-1102, 12-1147, 12-1172, 12-1173, 12-1174, 12-1175, 12-1176, 12-1177, 12-1178, 12-1180, 12-1181, 12-1182, 12-1183, 12-1184, 12-1185, 12-1186, 12-1187, 12-1188, 12-1189, 12-1190, 12-1191, 12-1192, 12-1193, 12-1194, 12-1195, and 12-1196. The consolidated cases on review have not previously been reviewed by this or any other Court.

No. 12-1272—which focuses on two issues of the rule involving new units—was severed from the cases consolidated under No. 12-1100 on June 28, 2012. *See* Order Severing New Source Issues (Doc. No. 1381112). Briefing in that case is currently being held in abeyance pending administrative reconsideration proceedings. *See* Order Holding Case in Abeyance (D.C. Cir. Sept. 12, 2012) (Doc. No. 1394140).

No. 12-1166, which challenges the New Source Performance Standards ("NSPS") issued in the same *Federal Register* notice as the rule under review in this case, was deconsolidated from No. 12-1100. *See* Order Deconsolidating NSPS

Issues (D.C. Cir. Aug. 24, 2012) (Doc. No. 1391295). Additionally, the NSPS issues in Nos. 12-1170 and 12-1185 were severed and assigned to a new docket, No. 12-1366, and consolidated with No. 12-1166. *Id*.

CORPORATE DISCLOSURE STATEMENTS

Industry Intervenor-Respondents submit the following statements pursuant to Rule 26.1 of the Federal Rules of Appellate Procedure and Circuit Rule 26.1:

Institute for Liberty ("IFL") is a non-profit and nonpartisan organization dedicated to defending the rights of individuals and businesses against undue encroachments by government that impair economic and civil liberties. It produces academic research on health, economic, and regulatory policy and, through its Center for American Regulatory Engagement, helps ordinary Americans participate in the regulatory process to ensure that their views are represented. IFL has no parent company, subsidiary, or affiliate that has issued shares or debt securities to the public.

National Black Chamber of Commerce ("NBCC") is a non-profit, nonpartisan, nonsectarian organization dedicated to the economic empowerment of African American communities through entrepreneurship. Incorporated in 1993, it represents nearly 100,000 African American-owned businesses, and advocates on behalf of the one million Black-owned businesses in the United States. The Chamber has 190 affiliated chapters located throughout the nation. Members of the NBCC include companies that are substantial consumers of electricity and whose economic viability depends on affordable electric service. NBCC has no

parent company, subsidiary, or affiliate that has issued shares or debt securities to the public.

National Mining Association ("NMA") is a non-profit, incorporated national trade association whose members include the producers of most of America's coal, metals, and industrial and agricultural minerals; manufacturers of mining and mineral processing machinery, equipment, and supplies; and engineering and consulting firms that serve the mining industry. NMA has no parent companies, subsidiaries, or affiliates that have issued shares or debt securities to the public, although NMA's individual members have done so.

Peabody Energy Corporation ("Peabody") is a publicly-traded company on the New York Stock Exchange ("NYSE") under the symbol "BTU." No public corporation owns more than 10% of Peabody's stock, with the exception of BlackRock, Inc. (NYSE: BLK), a publicly-held corporation which reported that as of December 31, 2011, it owned approximately 11.1% of Peabody's outstanding common stock. Peabody owns and operates several coal mines across the United States, and its coal production fuels approximately 10% of the nation's power generation.

Sunflower Electric Power Corporation ("Sunflower") is a not-for-profit, wholesale, electric generation and transmission utility, which is owned and governed by six member distribution cooperatives, serving customers in central

and western Kansas. Neither Sunflower, nor its member cooperatives issue stock, and therefore no publicly-traded company owns 10% or more of their stock.

Tri-State Generation & Transmission Association, Inc. ("Tri-State") is a wholesale electric power supply cooperative which operates on a not-for-profit basis and is owned by 1.5 million member-owners and 44 distribution cooperatives. Tri-State issues no stock and has no parent corporation. Accordingly, no publicly held corporation owns 10% or more of its stock.

Utility Air Regulatory Group ("UARG") is a not-for-profit association of individual electric generating companies and national trade associations that participates on behalf of its members collectively in administrative proceedings under the Clean Air Act, and in litigation arising from those proceedings, that affect electric generators. UARG has no outstanding shares or debt securities in the hands of the public and has no parent company. No publicly held company has a 10% or greater ownership interest in UARG.

White Stallion Energy Center, LLC ("White Stallion") is a limited liability company organized under the laws of the State of Texas engaged in the business of energy development and production. White Stallion has no parent companies, and no publicly-held corporation has a 10% or greater ownership interest in it.

TABLE OF CONTENTS

CER	CTIFIC	ATE AS TO PARTIES, RULINGS, AND RELATED CASES	1
COF	RPOR A	ATE DISCLOSURE STATEMENTS	iv
TAE	BLE O	F CONTENTS	vii
TAE	BLE O	F AUTHORITIES	viii
GLC	SSAR	Y	xi
STA	TUTE	S AND REGULATIONS	1
ISSU	JES PI	RESENTED	1
SUN	/MAR	Y OF ARGUMENT	1
ARC	GUME	NT	2
I.	EPA	's Averaging Alternative Is Lawful	2
II.	The Stack Testing Option for Non-Mercury Metals Is Lawful and Necessary		
	A.	EPA Reasonably Determined That Continuous Monitoring Should Not Be Required For the Non-Mercury Metals or the Surrogate PM Limit	7
	B.	The Periodic Stack Testing Option Provides a Reasonable Assurance of Compliance	10
CON	NCLUS	SION	16

TABLE OF AUTHORITIES

Document #1421812

CASES

40 C.F.R. § 63.503	2
40 C.F.R. § 63.652	2
40 C.F.R. § 63.846	
40 C.F.R. § 63.2240(c)	2
40 C.F.R. § 63.8050	2
40 C.F.R. § 63.9982	
40 C.F.R. § 63.10000(b)	
40 C.F.R. § 63.10007(a)(2)	
40 C.F.R. § 63.10009	
40 C.F.R. § 63.10009(b)	
40 C.F.R. § 63.10010(j)	
40 C.F.R. § 63.10030(e)(7)(i)	
40 C.F.R. § 63.10031(c)(3)	
40 C.F.R. § 63.10031(g)	
40 C.F.R. § 63.10032(d)(3)	
40 C.F.R. § 63.10032(g)	
40 C.F.R. § 63.10032(h)	14
40 C.F.R. Part 64 (2012)	11
40 C.F.R. § 64.2	13
40 C.F.R. § 64.3	
40 C.F.R. Part 70 (2012)	
40 C.F.R. § 70.5(d)	
40 C.F.R. § 70.6(c)(5)	15
FEDERAL REGISTER	
62 Fed. Reg. 54,900 (Oct. 22, 1997)	14, 15
64 Fed. Reg. 52,828 (Sept. 30, 1999)	6
76 Fed. Reg. 24,976 (May 3, 2011)	7, 10
77 Fed. Reg. 9304 (Feb. 16, 2012)	12, 13, 14, 15

MISCELLANEOUS

EPA, Memorandum from Jeffrey Cole, RTI International, to Bill Maxwell, EPA, National Emission Standards for Hazardous Air Pollutants (NESHAP) Maximum Achievable Control Technology (MACT) Floor Analysis for Coal- and Oil-fired Electric Utility Steam Generating Units for Final Rule (Dec. 16, 2011), EPA-HQ-OAR-2009-0234-20132 ("MACT Floor Analysis")
EPA, Responses to Public Comments on EPA's National Emission
Standards for Hazardous Air Pollutants from Coal- and Oil-Fired
Electric Utility Steam Generating Units (Dec. 2011), EPA-HQ-OAR-
2009-0234-20126 ("RTC")
UARG, Comments of the Utility Air Regulatory Group on National
Emission Standards for Hazardous Air Pollutants from Coal- and Oil-
Fired Electric Utility Steam Generating Units (Aug. 4, 2011), EPA-
HO-OAR-2009-0234-17775 ("UARG MATS Rule Comments")8. 9. 10

GLOSSARY

Page 13 of 122

Act Clean Air Act

Agency United States Environmental Protection Agency

CAA Clean Air Act

CAM Compliance Assurance Monitoring

CEMS Continuous Emissions Monitoring System

CPMS Continuous Parameter Monitoring System

EGU Electric Utility Generating Unit

EPA United States Environmental Protection Agency

HAP Hazardous Air Pollutant

ICR Information Collection Request

JA Joint Appendix

lb/TBtu Pounds per Trillion British Thermal Units

MACT Maximum Achievable Control Technology

MATS Mercury and Air Toxics Standards

NESHAP National Emission Standard for Hazardous Air Pollutants

NSPS New Source Performance Standards

PM Particulate Matter

PM CEMS Particulate Matter Continuous Emissions Monitoring System

PM CPMS Particulate Matter Continuous Parameter Monitoring System

RTC Response to Comments

SIP State Implementation Plan

UARG Utility Air Regulatory Group

UPL Upper Prediction Limit

STATUTES AND REGULATIONS

All applicable statutes and regulations are contained in the briefs of Environmental Petitioners and Respondent U.S. Environmental Protection Agency ("EPA" or the "Agency") or in the attached Addendum.

ISSUES PRESENTED

- Whether EPA lawfully allowed certain electric generating units
 ("EGUs") to demonstrate compliance with the Mercury and Air Toxics Standards
 Rule ("MATS rule") at the source level.
- 2. Whether EPA reasonably determined that periodic stack testing combined with other Clean Air Act ("CAA" or "the Act") requirements would provide a reasonable assurance of compliance with the applicable non-mercury metals and particulate matter ("PM") standard.

SUMMARY OF ARGUMENT

EPA's averaging alternative is lawful and consistent with the averaging alternatives included in many other maximum achievable control technology ("MACT") standards. The emission limits in the MATS rule were derived by EPA using a mixture of individual unit and averaged emissions. Allowing EGUs to comply with those limits by combining units' emissions neither extends the averaging period, nor allows emission of a greater quantity of a pollutant than through individual unit compliance.

EPA's stack testing compliance option for non-mercury metals and PM also is reasonable. Having established PM and non-mercury emission limits using only stack test data, EPA had good reason not to mandate use of PM continuous emissions monitoring systems ("CEMS"), which rely on an entirely different and sometimes difficult measurement technique. To determine compliance, EPA established a reasonable frequency for stack testing and made sure that EGUs are required to take steps to assure compliance in between tests and to certify their compliance status.

ARGUMENT

EPA's Averaging Alternative Is Lawful I.

Emissions averaging has been a long-time staple of EPA's CAA § 112 regulations. EPA has included emissions averaging as a compliance option in many national emission standards for hazardous air pollutants ("NESHAPs"). In the MATS rule, EPA identified the "affected source" as each individual or group of two or more new, reconstructed or existing sources, and included emissions averaging as a compliance option. 77 Fed. Reg. 9304, 9465, 9473-76 (Feb. 16, 2012) (40 C.F.R. §§ 63.9982, 63.10009).

¹ See, e.g., 40 C.F.R. §§ 63.503 (Group I polymers and resins), 63.652 (petroleum refineries), 63.846 (primary aluminum reduction plants), 63.2240(c) (plywood and composite wood products), and 63.8050 (miscellaneous coating manufacturing).

As a factual matter, emissions averaging played a significant role in the emission limits EPA set in the MATS rule. EPA's December 2009 information collection request ("ICR") required well-controlled EGUs to conduct stack testing for one or more hazardous air pollutant ("HAP") groupings.² EPA selected EGUs required to test based primarily on the control equipment installed at those units.³ Whether an EGU had its own stack or shared a common stack with other EGUs was not a factor in EPA's selection of units for ICR testing. Consequently, units sharing a common stack conducted a single emissions test and reported the results to EPA. EPA then attributed those emissions to each unit that shared the stack. In essence, EPA "averaged" the emissions of those units because it is impossible to accurately partition the measured emissions in the common stack between the two or more units that share the stack.⁴ Units with shared stacks comprise a large percentage of the "best performing" units in the pools EPA used to calculate the MACT floors. For example, 16 of the 47 "best performing" units in the mercury pool have a shared stack; 31 of the 130 units in the metals pool have a shared

² The testing groups were mercury, non-mercury metals, acid gases, organics and dioxins.

³ For units equipped with the same control technology, EPA used the date of installation as a secondary selection factor.

⁴ Even identical sister units burning the same coal and having the same design and control equipment do not have identical emissions. MACT Floor Analysis, Attachment a4, *excerpt* from spreadsheet tab *Coal fPM Data*, rows 3 & 5 (Dec. 16, 2011), EPA-HQ-OAR-2009-0234-20132 (JA).

stack.⁵ Thus, the emission limits in the MATS rule were derived using a mixture of individual stack and "averaged" emissions.

Environmental Petitioners do not challenge EPA's legal authority to include an emissions averaging option in the MATS rule. Instead, they offer a series of false claims to argue that the averaging alternative specific to the MATS rule violates the MACT standard setting provisions of CAA §§ 112(d)(2) and (d)(3). Central to Environmental Petitioners' argument is their claim that the averaging alternative relaxes the minimum floor for a MACT standard because "it increases the standards' averaging period." Environmental Petitioners assert that multiplication of the normal 30-boiler operating day compliance period by the number of EGUs being averaged provides the EGUs with a longer compliance period than they would have through the individual-unit compliance option.

That assertion misperceives how the averaging provision works. The same 30-boiler operating day compliance period applies regardless of whether compliance is demonstrated by individual units or by emissions averaging.⁷ EPA's

⁵ Shared stack units can be identified as those units at a given facility with identical emission values. MACT Floor Analysis, Attachment a2, *excerpt* from tab *Hg_Data_*>8300_Btulb & Attachment a4, *excerpt* from tab *Coal fPM Data* (JA ,).

⁶ Brief of Environmental Petitioners ("Env. Br.") at 16.

⁷ HAP metal and acid gas emissions have a 30-boiler operating day compliance period for both individual unit compliance and averaging. A 90-boiler operating day compliance period applies to mercury emissions when units are

Filed: 02/21/2013 Page 19 of 122

averaging formula requires the facility operator to sum, over the 30-group boiler operating day compliance period, the hourly mass emissions of the EGUs in the averaging group as well as either the hourly heat input or gross electrical output.⁸ See 77 Fed. Reg. at 9473-75 (40 C.F.R. § 63.10009(b)). Total mass emissions are then divided by total heat input or electrical output to produce a weighted average emissions rate over the 30-boiler operating day compliance period. That rate is then compared to the MACT limit. The standards' operating period does not change as a result of averaging.

Environmental Petitioners also dispute EPA's conclusion that the MATS emission limits are equally stringent regardless of whether compliance is demonstrated on an individual unit basis or by emissions averaging. See 77 Fed. Reg. at 9385. They contend there is a distinction between the "total quantity" of HAPs emitted under averaging and the "rate of emissions." See Env. Br. at 18. This argument fails because it again rests on the erroneous claim that the averaging

averaged. To assure a comparable stringency for mercury, EPA set a lower numeric limit for that averaging alternative—1.0 lb/TBtu for 90-day averaging period versus a 1.2 lb/TBtu for 30-day compliance period.

⁸ The MATS rule gives existing sources the choice of demonstrating compliance with either a heat input based limit or an electrical output based limit.

alternative extends the compliance period for individual units.⁹ The averaging alternative does not lessen the stringency of the emission limits in the MATS rule.

Finally, Environmental Petitioners claim that EPA violated the beyond-the-floor requirements of CAA § 112(d)(2) because it did not apply a discount factor to the averaging alternative. Env. Br. at 21-22. Nothing in CAA § 112 requires EPA to apply a discount factor to an averaging alternative when that option permits the same level of emissions as a standard applied only to individual units. EPA has offered a reasoned basis for its decision not to apply a discount factor in the case of EGUs. ¹⁰ EPA's decision should stand.

II. The Stack Testing Option for Non-Mercury Metals Is Lawful and Necessary

A rule satisfies the CAA § 114(a)(3) "enhanced monitoring" requirement if, in the Agency's "informed discretion," it provides "a reasonable assurance of compliance." *Sierra Club v. EPA*, 353 F.3d 976, 990-91 (D.C. Cir. 2004) (quoting *Nat'l Lime Ass'n v. EPA*, 233 F.3d 625, 635 (D.C. Cir. 2000) (internal quotation

⁹ Environmental Petitioners cite EPA's statement that the stringency of an emissions standard "is a function of both the numerical value of the standard and the averaging period." Env. Br. at 18 (quoting 64 Fed. Reg. 52,828, 52,930-31 (Sept. 30, 1999)). For all HAPs emission limits, except the more stringent alternative mercury limit, the numerical limit and the averaging period are the same whether compliance is demonstrated for individual units or by emissions averaging. Thus, the stringency of those limits is the same.

¹⁰ 77 Fed. Reg. at 9386; Response to Comment ("RTC"), Vol. 2 at 361-63 (Dec. 2011), EPA-HQ-OAR-2009-0234-20126 (JA -).

marks omitted) and Natural Res. Def. Council, Inc. v. EPA, 194 F.3d 130, 136 (D.C. Cir. 1999)). Nothing in CAA § 114(a)(3) requires that the specified monitoring definitively establish a source's compliance or noncompliance. Natural Res. Def. Council, 194 F.3d at 137. To justify the specified monitoring, EPA need only "consider relevant factors" (like those presented in comments on the proposed rule) and articulate a "rational connection between the facts and its choices." *Id.* at 136. EPA adopted the stack testing option for non-mercury metals and PM using appropriate CAA procedures, and provided the CAA-mandated level of compliance assurance.

EPA Reasonably Determined That Continuous Monitoring A. Should Not Be Required For the Non-Mercury Metals or the **Surrogate PM Limit**

Environmental Petitioners begin by criticizing EPA for failing to require CEMS. Env. Br. at 11. However, nothing in the CAA requires the use of CEMS, even when such monitoring is feasible or practicable. *Natural Res. Def. Council*, 194 F.3d at 136-37. And, EPA had good reason not to require CEMS for PM or non-mercury metals.

In the proposed rule, EPA requested comment on "application of PM CEMS" to EGUs, and the use of data from such systems for compliance determinations." 76 Fed. Reg. 24,976, 25,052 (May 3, 2011). In response, EPA received detailed objections citing, among other things, the errors associated with PM CEMS

measurements, and the difficulty of developing adequate site-specific correlations of PM CEMS to the applicable stack test method at EGUs using the type of controls (like wet scrubbers and fabric filters) required under the rule. *See, e.g.,* UARG MATS Rule Comments at 141-50 (Aug. 4, 2011), EPA-HQ-OAR-2009-0234-1775 (JA______). Because PM CEMS do not measure PM directly, the monitor's response must be correlated to the applicable stack test at each source in a series of tests conducted under the EGU's full range of operations. The lower the emission limit, the more difficult the correlation criteria are to pass. To pass the statistical requirements and obtain a robust correlation, most EGUs must detune or even disable certain emission controls, creating other operational problems and potential liability. However, even a properly correlated PM CEMS is allowed to differ from the applicable stack test by as much as 25 percent. *Id.*

None of the issues commenters raised with PM CEMS were considered in the standard setting process. EPA based the rule's numeric emission limits for non-mercury metals and PM solely on data obtained from EGUs during short term stack tests conducted under full load, steady-state operating conditions. RTC, Vol. 1 at 460, 517, 551 (JA____, ____, ____). EPA's prediction of those units' highest short term emissions using the upper prediction limit ("UPL") statistic addressed only the variability *in those stack test data*. *Id*. As a result, commenters argued that to mandate PM CEMS, EPA at a minimum *would have to examine PM CEMS*

data and consider the impact of the difference in measurement techniques on the numeric emission limit. UARG MATS Rule Comments at 143 (JA___). As Environmental Petitioners argue in their challenge to the rule's option for use of a PM continuous parameter monitoring system ("CPMS"), "[t]his Court has remanded particulate matter standards before when test methods used to set the standard conflicted with those used to measure compliance." Env. Br. at 24 (citing Portland Cement Ass'n v. Ruckelshaus, 486 F.2d 375, 396-97 (D.C. Cir. 1973) and Clean Air Implementation Project v. EPA, 150 F.3d 1200, 1203 (D.C. Cir. 1998)).

Persuaded by commenters not to require PM CEMS, EPA provided it as an option instead. 77 Fed. Reg. at 9384; RTC, Vol. 2 at 89-90, 200 (citing the difficulties of certifying a PM CEMS for direct compliance) (JA___-__, ___). To compensate for "regular fuel and process variability" not otherwise accounted for in the numeric emission limit, EPA allowed EGUs that opt to use PM CEMS to comply on a 30-boiler operating day rolling average basis. RTC, Vol. 1 at 460, 463 (JA___, ___). However, neither the UPL-adjusted numeric emission limit, nor the 30-boiler operating day rolling average, address measurement error associated with other types of data. As a result, EGUs that opt to use PM CEMS do so at their own risk. 11

¹¹ Because CEMS that measure non-mercury metals have not been validated by EPA, EPA did not propose (or solicit comment on) their use. However, EPA in the final rule acknowledged the ability of EGUs to petition for approval of such

B. The Periodic Stack Testing Option Provides a Reasonable Assurance of Compliance

Having determined that PM CEMS should not be required, EPA identified a reasonable frequency for conducting PM or non-mercury metals stack tests and explained why it was not necessary to mandate additional monitoring to assure compliance with the rule. 77 Fed. Reg. at 9384, RTC, Vol. 2 at 93 (JA___). In its proposed rule, EPA solicited comment on requiring EGUs that opt to use stack testing also to comply with site-specific control device operating limits established during performance testing. The concept EPA proposed was similar to that adopted in the final rule with respect to PM CPMS, except that the focus was on operating parameters for control devices used to comply with the applicable limit. 76 Fed. Reg. at 25,030.

Commenters objecting to the proposed operating limits provided detailed information demonstrating the lack of direct correlation between the enforceable control device operating limits and the applicable PM or non-mercury emission limit. *See, e.g.,* UARG MATS Rule Comments at 202-211 (JA___-__); RTC, Vol. 2 at 73-85 (JA__-__). Commenters advocated that rather than specify

[&]quot;HAP metals CEMS" as an alternative test method by performing their own validation testing and submitting the results to EPA under 40 C.F.R. § 63.7(f). 77 Fed. Reg. at 9478 (40 C.F.R. § 63.10010(j)); RTC, Vol. 2 at 175 (JA).

Filed: 02/21/2013 Page :

operating limits, EPA allow EGUs to employ principles like those required under the Compliance Assurance Monitoring ("CAM") rule at 40 C.F.R. Part 64.¹² *Id*.

EPA adopted the CAM rule to provide "enhanced monitoring" at major stationary sources, like EGUs. The CAM rule applies to major sources that rely on a control device to comply with certain emission standards, but that are not required to perform continuous monitoring in the units of the standard. The rule requires such sources to develop a plan that uses control device or other operating parameters to trigger requirements for investigation and corrective action.

Although the monitoring in the CAM rule does not identify the precise level of emissions, it does identify parameter levels correlated to proper operation of the control based on a number of factors including the potential failure rate of the control device. 40 C.F.R. § 64.3. This Court previously upheld the monitoring provisions in the CAM rule as sufficient to satisfy both the "enhanced monitoring"

¹² Because the PM CPMS option requires establishment of an enforceable operating limit, rather than a requirement for investigation and corrective action like the CAM rule, few EGUs with significant compliance margin are likely to choose it. The enforceable operating limit unreasonably restricts operations for those EGUs.

However, as with PM CEMS, as long as EPA continues to provide a stack testing option that is consistent with the manner in which EPA set the emission limits, the additional option is not unlawful. Environmental Petitioners' objections to the use of the highest value during the stack test, rather than the average, are based on unrealistic assumptions regarding variability of emissions during a single steady-state stack test, and the misconception that stack tests generate the same type of hourly average emissions values generated by a PM CPMS.

requirements in CAA § 114(a)(3) and the requirements for alternative monitoring in CAA § 504(b). *Natural Res. Def. Council*, 194 F.3d at 135-37.

In the final rule, EPA agreed in part with commenters and eliminated the additional control device operating limits associated with the stack testing option for PM. 13 Specifically, "after considering other programs in place to ensure proper operation of controls at EGUs," EPA concluded that those requirements in combination with frequent periodic stack testing constituted enhanced monitoring. 77 Fed. Reg. at 9384. The requirements to which EPA referred included the CAM rule, "New Source Review permit conditions, and other SIP and NSPS requirements for operating and maintaining equipment in accordance with good air pollution control practices." *Id.*

EPA's conclusion regarding the contribution of other applicable rules to the MATS rule compliance demonstration is reasonable. All EGUs are subject to one or more filterable PM standards under a state implementation plan ("SIP"), New Source Review permit, or New Source Performance Standard ("NSPS"). With the exception of NSPS proposed after November 15, 1990, all of those PM standards are subject to the CAM rule and, as a result, EGUs already are monitoring parameters relevant to operation of their PM emission controls and taking

¹³ EPA reached a different conclusion with respect to acid gases at liquid oilfired EGUs, for which EPA did impose additional requirements. 77 Fed. Reg. at 9384.

PM limits in the NSPS that are exempt from the CAM rule include their own CAM-like monitoring requirements that apply to any EGU that does not use PM CEMS. 40 C.F.R. §§ 60.42Da(c) and (d), 60.48Da(o). Although EPA could have duplicated the CAM and NSPS PM monitoring requirements in the MATS rule, given that the monitoring in those rules is statutorily required, EPA reasonably concluded that those rules could be relied upon, and that including similar requirements in the MATS rule would be redundant. RTC, Vol. 2 at 75 (JA___).

All EGUs also are subject to one or more requirements to operate and maintain all of their emission control devices consistent with good air pollution control practices for minimizing emissions "at all times." The NSPS and Subpart UUUUU contain such requirements, as do many SIPs. 40 C.F.R. §§ 60.11(d); 77 Fed. Reg. at 9466 (40 C.F.R. § 63.10000(b)). In other words, EGUs have a duty to operate controls consistent with their emission limits regardless of the type of monitoring being conducted. EPA is authorized to conduct inspections and use any available information, including operating and maintenance procedures and records, to determine compliance with this requirement. *Id.* The CAM rule satisfied the CAA requirement for "enhanced monitoring" by adding to these good air pollution control practice provisions enforceable requirements to investigate

changes in control device parameters and to take corrective action. 62 Fed. Reg. 54,900, 54,918-19 (Oct. 22, 1997).

The MATS rule's testing, recordkeeping, and reporting requirements supplement the requirements in other rules by ensuring that EGUs do not change operating conditions (e.g., fuel) without re-testing, and that they keep records of their responses to control device operating problems. Stack tests must be conducted under "maximum normal operating load conditions," and EGUs that switch fuel "type" (e.g., coal rank) must identify a test based on that fuel. 77 Fed. Reg. at 9472, 9482 (40 C.F.R. §§ 63.10007(a)(2), 63.10031(c)(3)). In between stack tests, EGUs must keep records of "the occurrence and duration of each malfunction of an operation (i.e., the process equipment) or the air pollution control and monitoring equipment," and "actions taken during periods of malfunction to minimize emissions ... including corrective actions to restore malfunctioning process and air pollution control and monitoring equipment to its normal or usual manner of operation." Id. at 9484 (40 C.F.R. § 63.10032(g) and (h)). EGUs also must report each malfunction "which caused or may have caused any applicable emission limitation to be exceeded." *Id.* at 9483 (40 C.F.R. § 63.10031(g)).

Low-emitting EGUs, which are allowed after three years to reduce stack testing frequency, are subject to additional requirements to ensure emission

controls are properly operated. Specifically, low-emitting EGUs must submit an initial notification of compliance status identifying whether "there have been any operational changes since the last stack test that could increase emissions," and thereafter must keep records documenting that "there was no change in source operations including fuel composition and operation of air pollution control equipment that would cause emissions of the pollutant to increase within the past year." *Id.* at 9482-83 (40 C.F.R. §§ 63.10030(e)(7)(i), 63.10032(d)(3)).

Finally, consistent with the "compliance certification" requirement in CAA § 114(a)(3), all of the information collected and recorded by EGUs under the MATS rule or elsewhere must be considered in annual reports certifying whether the EGU's compliance with its applicable requirements, including CAA § 112 standards like the non-mercury metals and PM limits, was "continuous" or "intermittent." 40 C.F.R. § 70.6(c)(5) (implementing the compliance certification requirement in the federal operating permit program). Upon promulgation of that requirement, EPA made clear that CAA § 113(c)(2) and 18 U.S.C. § 1001 (prohibiting false certifications or omission of material information) prevent responsible officials from ignoring relevant information regarding a source's compliance status in that compliance certification, even if the source would be in compliance based solely on the compliance method specified in the applicable rule (e.g., in this case the stack test data). 62 Fed. Reg. at 54,937. The responsible

official also must certify the truth, accuracy, and completeness of that submission. 40 C.F.R. § 70.5(d).

In sum, although the MATS rule does not itself mandate additional monitoring during the period in between stack tests, it does require (1) operation of PM controls consistent with the applicable PM emission limit, and (2) recording of incidents of control device malfunction and corrective action. Further, all EGUs are subject to CAM rule requirements to monitor and respond to PM control device operating parameters, and operating permit requirements to certify "continuous" or "intermittent" compliance with the applicable MATS rule PM or non-mercury metals standard based on all material information. In light of the manner in which EPA set the PM and non-mercury metals standards, and the absence of other reasonable monitoring options for EGUs, EPA reasonably determined in its "informed discretion" that stack testing combined with these and other requirements are sufficient to provide a reasonable assurance of compliance with those MATS rule standards.

CONCLUSION

For the reasons set out above and in EPA's brief, the emissions averaging and non-mercury metals stack testing compliance demonstration options should be upheld.

DATED: February 21, 2013

/s/ Lauren E. Freeman
F. William Brownell
Lauren E. Freeman
Lee B. Zeugin
Elizabeth L. Horner
HUNTON & WILLIAMS LLP
2200 Pennsylvania Avenue, N.W.
Washington, D.C. 20037
(202) 955-1500
bbrowell@hunton.com
lfreeman@hunton.com
lzeugin@hunton.com
ehorner@hunton.com

Counsel for Utility Air Regulatory Group

/s/ Jeremy C. Marwell
Jeremy C. Marwell
VINSON & ELKINS LLP
2200 Pennsylvania Avenue, N.W.
Suite 500 West
Washington, D.C. 20037
(202) 639-6507
jmarwell@velaw.com

Eric A. Groten
VINSON & ELKINS LLP
2801 Via Fortuna, Suite 100
Austin, TX 78746-7568
(512) 542-8709
egroten@velaw.com

Counsel for White Stallion Energy Center, LLC Respectfully submitted,

Filed: 02/21/2013

/s/ Peter S. Glaser
Peter S. Glaser
George Y. Sugiyama
TROUTMAN SANDERS LLP
401 Ninth Street, N.W., Suite 1000
Washington, D.C. 20001
(202) 274-2998
peter.glaser@troutmansanders.com

Counsel for National Mining Association, Peabody Energy Corporation, and Sunflower Electric Power Corporation

/s/ David B. Rivkin, Jr.
David B. Rivkin, Jr.
Lee A. Casey
Mark W. DeLaquil
Andrew M. Grossman
BAKER & HOSTETLER LLP
Washington Square, Suite 1100
1050 Connecticut Avenue, N.W.
Washington, D.C. 20036-5304
(202) 861-1731
drivkin@bakerlaw.com

Counsel for National Black Chamber of Commerce and Institute for Liberty

/s/ Jeffry R. Holmstead
Jeffrey R. Holmstead
Sandra Y. Snyder
BRACEWELL & GIULIANI LLP
2000 K Street, N.W., Suite 500
Washington, D.C. 20006-1872
(202) 828-5800
Jeffrey.Holmstead@bgllp.com
Sandra.Snyder@bgllp.com

Counsel for Tri-State Generation and Transmission Association, Inc.

CERTIFICATE OF COMPLIANCE

Pursuant to Rule 32(a)(7)(C) of the Federal Rules of Appellate Procedure and Circuit Rules 32(a)(1) and 32(a)(2)(C), I hereby certify that the foregoing Joint Brief of Industry Intervenors in Response to Environmental Petitioners contains 3723 words, as counted by a word processing system that includes headings, footnotes, quotations, and citations in the count, and therefore is within the word limit set by the Court.

Dated: February 21, 2013 /s/ Lauren E. Freeman

Statutory and Regulatory Addendum

TABLE OF CONTENTS

18 U.S.C. § 1001	Addendum - 001
Clean Air Act, 42 U.S.C. § 7413(c)(2), CAA § 113(c)(2)	Addendum - 002
40 C.F.R. Part 60 (2012)	
40 C.F.R. § 60.11(d)	Addendum - 003
40 C.F.R. § 60.48Da(o)	Addendum - 004
40 C.F.R. Part 63 (2012)	
40 C.F.R. § 63.7(f)	Addendum - 008
40 C.F.R. § 63.503	Addendum - 010
40 C.F.R. § 63.652	Addendum - 028
40 C.F.R. § 63.846	Addendum - 047
40 C.F.R. § 63.2240(c)	Addendum - 049
40 C.F.R. § 63.8050	Addendum - 051
40 C.F.R. § 63.9982	Addendum - 054
40 C.F.R. § 63.10000(b)	Addendum - 056
40 C.F.R. § 63.10007(a)(2)	Addendum - 057
40 C.F.R. § 63.10009	Addendum - 058
40 C.F.R. § 63.10010(j)	Addendum - 064
40 C.F.R. § 63.10030(e)(7)(i)	Addendum - 066
40 C.F.R. § 63.10031(c)(3)	Addendum - 067
40 C.F.R. § 63.10031(g)	Addendum - 069
40 C.F.R. § 63.10032(d)(3)	Addendum - 070

40 C.F.R. § 63.10032(g)	Addendum - 070
40 C.F.R. § 63.10032(h)	Addendum - 070
40 C.F.R. Part 64 (2012)	Addendum - 071
40 C.F.R. Part 70 (2012)	
40 C.F.R. § 70.5(d)	Addendum - 083
40 C.F.R. § 70.6(c)(5)	Addendum - 084

1990—Pub. L. 101-647, title XXV, \$2501(b), title XXXV, \$3532, Nov. 29, 1990, 104 Stat. 4860, 4925, inserted a period after "1031" and added item 1032.

1989—Pub. L. 101-73, title IX, §§961(g)(2), 962(a)(4), Aug. 9, 1989, 103 Stat. 500, 502, struck out item 1008 "Federal Savings and Loan Insurance Corporation transactions" and item 1009 "Rumors regarding Federal Savings and Loan Insurance Corporation".

 $1988 — Pub. L. 100-700, \S 2(c), Nov. 19, 1988, 102 Stat. 4632, added item 1031.$

1984—Pub. L. 98–473, title II, §§ 1602(b), 2102(b), Oct. 12, 1984, 98 Stat. 2184, 2192, added items 1029 and 1030.

1982—Pub. L. 97–398, §3, Dec. 31, 1982, 96 Stat. 2010, added item 1028.

1974—Pub. L. 93–406, title I, §111(a)(2)(B)(iii), Sept. 2, 1974, 88 Stat. 852, substituted "Employee Retirement Income Security Act of 1974" for "Welfare and Pension Plans Disclosure Act" in item 1027.

1967—Pub. L. 90–19, §24(e), May 25, 1967, 81 Stat. 28, included "Department of Housing and Urban Development" in item 1010, and substituted the same for "Public Housing Administration" in item 1012.

1962—Pub. L. 87–420, §17(d), Mar. 20, 1962, 76 Stat. 42, added item 1027.

1951—Act Oct. 31, 1951, ch. 655, §25, 65 Stat. 720, substituted "Public Housing Administration" for "United States Housing Authority" in item 1012.
1949—Act May 24, 1949, ch. 139, §§18, 19, 63 Stat. 92,

1949—Act May 24, 1949, ch. 139, §§18, 19, 63 Stat. 92, corrected spelling in item 1012 and substituted "officers" for "offices" in item 1019.

18 U.S.C. § 1001 § 1001. Statements or entries generally

- (a) Except as otherwise provided in this section, whoever, in any matter within the jurisdiction of the executive, legislative, or judicial branch of the Government of the United States, knowingly and willfully—
 - (1) falsifies, conceals, or covers up by any trick, scheme, or device a material fact;
 - (2) makes any materially false, fictitious, or fraudulent statement or representation; or
 - (3) makes or uses any false writing or document knowing the same to contain any materially false, fictitious, or fraudulent statement or entry;

shall be fined under this title, imprisoned not more than 5 years or, if the offense involves international or domestic terrorism (as defined in section 2331), imprisoned not more than 8 years, or both. If the matter relates to an offense under chapter 109A, 109B, 110, or 117, or section 1591, then the term of imprisonment imposed under this section shall be not more than 8 years.

- (b) Subsection (a) does not apply to a party to a judicial proceeding, or that party's counsel, for statements, representations, writings or documents submitted by such party or counsel to a judge or magistrate in that proceeding.
- (c) With respect to any matter within the jurisdiction of the legislative branch, subsection (a) shall apply only to—
 - (1) administrative matters, including a claim for payment, a matter related to the procurement of property or services, personnel or employment practices, or support services, or a document required by law, rule, or regulation to be submitted to the Congress or any office or officer within the legislative branch; or
 - (2) any investigation or review, conducted pursuant to the authority of any committee, subcommittee, commission or office of the Congress, consistent with applicable rules of the House or Senate.

(June 25, 1948, ch. 645, 62 Stat. 749; Pub. L. 103-322, title XXXIII, §330016(1)(L), Sept. 13, 1994, 108 Stat. 2147; Pub. L. 104-292, §2, Oct. 11, 1996, 110 Stat. 3459; Pub. L. 108-458, title VI, §6703(a), Dec. 17, 2004, 118 Stat. 3766; Pub. L. 109-248, title I, §141(c), July 27, 2006, 120 Stat. 603.)

Page 37 of 122

HISTORICAL AND REVISION NOTES

Based on title 18, U.S.C., 1940 ed., § 80 (Mar. 4, 1909, ch. 321, § 35, 35 Stat. 1095; Oct. 23, 1918, ch. 194, 40 Stat. 1015; June 18, 1934, ch. 587, 48 Stat. 996; Apr. 4, 1938, ch. 69, 52 Stat. 197)

Section 80 of title 18, U.S.C., 1940 ed., was divided into two parts.

The provision relating to false claims was incorporated in section 287 of this title.

Reference to persons causing or procuring was omitted as unnecessary in view of definition of "principal" in section 2 of this title.

Words "or any corporation in which the United States of America is a stockholder" in said section 80 were omitted as unnecessary in view of definition of "agency" in section 6 of this title.

In addition to minor changes of phraseology, the

In addition to minor changes of phraseology, the maximum term of imprisonment was changed from 10 to 5 years to be consistent with comparable sections. (See reviser's note under section 287 of this title.)

AMENDMENTS

2006—Subsec. (a). Pub. L. 109-248 inserted last sentence in concluding provisions

tence in concluding provisions. 2004—Subsec. (a). Pub. L. 108-458 substituted "be fined under this title, imprisoned not more than 5 years or, if the offense involves international or domestic terrorism (as defined in section 2331), imprisoned not more than 8 years, or both" for "be fined under this title or imprisoned not more than 5 years, or both" in concluding provisions.

1996—Pub. L. 104–292 reenacted section catchline without change and amended text generally. Prior to amendment, text read as follows: "Whoever, in any matter within the jurisdiction of any department or agency of the United States knowingly and willfully falsifies, conceals or covers up by any trick, scheme, or device a material fact, or makes any false, fictitious or fraudulent statements or representations, or makes or uses any false writing or document knowing the same to contain any false, fictitious or fraudulent statement or entry, shall be fined under this title or imprisoned not more than five years or both."

not more than five years, or both."
1994—Pub. L. 103-322 substituted "fined under this title" for "fined not more than \$10,000".

CHANGE OF NAME

Reference to United States magistrate or to magistrate deemed to refer to United States magistrate judge pursuant to section 321 of Pub. L. 101–650, set out as a note under section 631 of Title 28, Judiciary and Judicial Procedure.

SHORT TITLE OF 2004 AMENDMENT

Pub. L. 108-275, §1, July 15, 2004, 118 Stat. 831, provided that: "This Act [enacting section 1028A of this title, amending sections 641 and 1028 of this title, and enacting provisions listed in a table relating to sentencing guidelines set out as a note under section 994 of Title 28, Judiciary and Judicial Procedure] may be cited as the 'Identity Theft Penalty Enhancement Act'."

SHORT TITLE OF 2003 AMENDMENT

Pub. L. 108–21, title VI, §607(a), Apr. 30, 2003, 117 Stat. 689, provided that: "'This section [amending section 1028 of this title] may be cited as the 'Secure Authentication Feature and Enhanced Identification Defense Act of 2003' or 'SAFE ID Act'."

SHORT TITLE OF 2000 AMENDMENT

Pub. L. 106-578, §1, Dec. 28, 2000, 114 Stat. 3075, provided that: "This Act [amending section 1028 of this

(relating to preconstruction requirements), an order under section 7477 of this title (relating to preconstruction requirements), an order under section 7603 of this title (relating to emergency orders), section 7661a(a) or 7661b(c) of this title (relating to permits), or any requirement or prohibition of subchapter IV-A of this chapter (relating to acid deposition control), or subchapter VI of this chapter (relating to stratospheric ozone control), including a requirement of any rule, order, waiver, or permit promulgated or approved under such sections or subchapters, and including any requirement for the payment of any fee owed the United States under this chapter (other than subchapter II of this chapter) shall, upon conviction, be punished by a fine pursuant to title 18 or by imprisonment for not to exceed 5 years, or both. If a conviction of any person under this paragraph is for a violation committed after a first conviction of such person under this paragraph, the maximum punishment shall be doubled with respect to both the fine and imprisonment.

CAA § 113(c)(2)

- (2) Any person who knowingly—
- (A) makes any false material statement, representation, or certification in, or omits material information from, or knowingly alters, conceals, or fails to file or maintain any notice, application, record, report, plan, or other document required pursuant to this chapter to be either filed or maintained (whether with respect to the requirements imposed by the Administrator or by a State);
- (B) fails to notify or report as required under this chapter: or
- (C) falsifies, tampers with, renders inaccurate, or fails to install any monitoring device or method required to be maintained or followed under this chapter²

shall, upon conviction, be punished by a fine pursuant to title 18 or by imprisonment for not more than 2 years, or both. If a conviction of any person under this paragraph is for a violation committed after a first conviction of such person under this paragraph, the maximum punishment shall be doubled with respect to both the fine and imprisonment.

(3) Any person who knowingly fails to pay any fee owed the United States under this subchapter, subchapter III, IV-A, V, or VI of this chapter shall, upon conviction, be punished by a fine pursuant to title 18 or by imprisonment for not more than 1 year, or both. If a conviction of any person under this paragraph is for a violation committed after a first conviction of such person under this paragraph, the maximum punishment shall be doubled with respect to both the fine and imprisonment.

(4) Any person who negligently releases into the ambient air any hazardous air pollutant listed pursuant to section 7412 of this title or any extremely hazardous substance listed pursuant to section 11002(a)(2) of this title that is not listed in section 7412 of this title, and who at the time negligently places another person in imminent danger of death or serious bodily injury shall, upon conviction, be punished by a fine under title 18 or by imprisonment for not more

than 1 year, or both. If a conviction of any person under this paragraph is for a violation committed after a first conviction of such person under this paragraph, the maximum punishment shall be doubled with respect to both the fine and imprisonment.

(5)(A) Any person who knowingly releases into the ambient air any hazardous air pollutant listed pursuant to section 7412 of this title or any extremely hazardous substance listed pursuant to section 11002(a)(2) of this title that is not listed in section 7412 of this title, and who knows at the time that he thereby places another person in imminent danger of death or serious bodily injury shall, upon conviction, be punished by a fine under title 18 or by imprisonment of not more than 15 years, or both. Any person committing such violation which is an organization shall, upon conviction under this paragraph, be subject to a fine of not more than \$1,000,000 for each violation. If a conviction of any person under this paragraph is for a violation committed after a first conviction of such person under this paragraph, the maximum punishment shall be doubled with respect to both the fine and imprisonment. For any air pollutant for which the Administrator has set an emissions standard or for any source for which a permit has been issued under subchapter V of this chapter, a release of such pollutant in accordance with that standard or permit shall not constitute a violation of this paragraph or paragraph (4).

(B) In determining whether a defendant who is an individual knew that the violation placed another person in imminent danger of death or serious bodily injury-

(i) the defendant is responsible only for actual awareness or actual belief possessed; and

(ii) knowledge possessed by a person other than the defendant, but not by the defendant, may not be attributed to the defendant:

except that in proving a defendant's possession of actual knowledge, circumstantial evidence may be used, including evidence that the defendant took affirmative steps to be shielded from relevant information.

(C) It is an affirmative defense to a prosecution that the conduct charged was freely consented to by the person endangered and that the danger and conduct charged were reasonably foreseeable hazards of-

(i) an occupation, a business, or a profession;

(ii) medical treatment or medical or scientific experimentation conducted by professionally approved methods and such other person had been made aware of the risks involved prior to giving consent.

The defendant may establish an affirmative defense under this subparagraph by a preponderance of the evidence.

(D) All general defenses, affirmative defenses, and bars to prosecution that may apply with respect to other Federal criminal offenses may apply under subparagraph (A) of this paragraph and shall be determined by the courts of the United States according to the principles of common law as they may be interpreted in the light of reason and experience. Concepts of justification and excuse applicable under this sec-

²So in original. Probably should be followed by a comma.

§ 60.9

(iii) Providing standards for use by the voluntary consensus standard body to approve the accrediting body that will accredit the audit sample providers.

(4) The technical criteria documents for the accredited sample providers and the accrediting body shall be developed through a public process guided by a voluntary consensus standards body (VCSB). The VCSB shall operate in accordance with the procedures and requirements in the Office of Management and Budget Circular A-119. A copy of Circular A-119 is available upon request by writing the Office of Information and Regulatory Affairs, Office of Management and Budget, 725 17th Street, NW., Washington, DC 20503, by calling (202) 395-6880 or downloading online http://standards.gov/ at standards gov/a119.cfm. The VCSB shall approve all accrediting bodies. The Administrator will review all technical criteria documents. If the technical criteria documents do not meet the minimum technical requirements in paragraphs (g)(2) through (4)of this section, the technical criteria documents are not acceptable and the proposed audit sample program is not capable of producing audit samples of sufficient quality to be used in a compliance test. All acceptable technical criteria documents shall be posted on the EPA Web site at the following URL, http:// www.epa.gov/ttn/emc.

[36 FR 24877, Dec. 23, 1971, as amended at 39 FR 9314, Mar. 8, 1974; 42 FR 57126, Nov. 1, 1977; 44 FR 33612, June 11, 1979; 54 FR 6662, Feb. 14, 1989; 54 FR 21344, May 17, 1989; 64 FR 7463, Feb. 12, 1999; 72 FR 27442, May 16, 2007; 75 FR 55646, Sept. 13, 2010]

§ 60.9 Availability of information.

The availability to the public of information provided to, or otherwise obtained by, the Administrator under this part shall be governed by part 2 of this chapter. (Information submitted voluntarily to the Administrator for the purposes of §§ 60.5 and 60.6 is governed by §§ 2.201 through 2.213 of this chapter and not by §2.301 of this chapter.)

§ 60.10 State authority.

The provisions of this part shall not be construed in any manner to preclude

any State or political subdivision thereof from:

- (a) Adopting and enforcing any emission standard or limitation applicable to an affected facility, provided that such emission standard or limitation is not less stringent than the standard applicable to such facility.
- (b) Requiring the owner or operator of an affected facility to obtain permits, licenses, or approvals prior to initiating construction, modification, or operation of such facility.

§ 60.11 Compliance with standards and maintenance requirements.

- (a) Compliance with standards in this part, other than opacity standards, shall be determined in accordance with performance tests established by §60.8, unless otherwise specified in the applicable standard.
- (b) Compliance with opacity standards in this part shall be determined by conducting observations in accordance with Method 9 in appendix A of this part, any alternative method that is approved by the Administrator, or as provided in paragraph (e)(5) of this section. For purposes of determining initial compliance, the minimum total time of observations shall be 3 hours (30 6-minute averages) for the performance test or other set of observations (meaning those fugitive-type emission sources subject only to an opacity standard).
- (c) The opacity standards set forth in this part shall apply at all times except during periods of startup, shutdown, malfunction, and as otherwise provided in the applicable standard.
- (d) At all times, including periods of startup, shutdown, and malfunction, owners and operators shall, to the extent practicable, maintain and operate any affected facility including associated air pollution control equipment in a manner consistent with good air pollution control practice for minimizing emissions. Determination of whether acceptable operating and maintenance procedures are being used will be based on information available to the Administrator which may include, but is not limited to, monitoring results, opacity observations, review of operating and maintenance procedures, and inspection of the source.

40 C.F.R. § 60.11(d)

§60.49Da(m)), and dividing by the average hourly gross energy output (measured according to the provisions of §60.49Da(k)) or the average hourly net

energy output, as applicable.

40 C.F.R. § 60.48Da(o)

- (o) Compliance provisions for sources subject to $\S 60.42Da(c)(2)$ or (d). Except as provided for in paragraph (p) of this section, the owner or operator of an affected facility for which construction, reconstruction, or modification commenced after February 28, 2005, shall demonstrate compliance with each applicable emission limit according to the requirements in paragraphs (o)(1) through (o)(5) of this section.
- (1) You must conduct a performance test to demonstrate initial compliance with the applicable PM emissions limit in §60.42Da(c)(2) or (d) by the applicable date specified in §60.8(a). Thereafter, you must conduct each subsequent performance test within 12 calendar months following the date the previous performance test was required to be conducted. You must conduct each performance test according to the requirements in §60.8 using the test methods and procedures in §60.50Da. The owner or operator of an affected facility that has not operated for 60 consecutive calendar days prior to the date that the subsequent performance test would have been required had the unit been operating is not required to perform the subsequent performance test until 30 calendar days after the next boiler operating day. Requests for additional 30 day extensions shall be granted by the relevant air division or office director of the appropriate Regional Office of the U.S. EPA.
- (2) You must monitor the performance of each electrostatic precipitator or fabric filter (baghouse) operated to comply with the applicable PM emissions limit in §60.42Da(c)(2) or (d) using a continuous opacity monitoring system (COMS) according to the requirements in paragraphs (o)(2)(i) through (vi) unless you elect to comply with one of the alternatives provided in paragraphs (o)(3) and (o)(4) of this section, as applicable to your control device.
- (i) Each COMS must meet Performance Specification 1 in 40 CFR part 60, appendix B.

- (ii) You must comply with the quality assurance requirements in paragraphs (o)(2)(ii)(A) through (E) of this section.
- (A) You must automatically (intrinsic to the opacity monitor) check the zero and upscale (span) calibration drifts at least once daily. For a particular COMS, the acceptable range of zero and upscale calibration materials is as defined in the applicable version of Performance Specification 1 in 40 CFR part 60, appendix B.
- (B) You must adjust the zero and span whenever the 24-hour zero drift or 24-hour span drift exceeds 4 percent opacity. The COMS must allow for the amount of excess zero and span drift measured at the 24-hour interval checks to be recorded and quantified. The optical surfaces exposed to the effluent gases must be cleaned prior to performing the zero and span drift adjustments, except for systems using automatic zero adjustments. For systems using automatic zero adjustments, the optical surfaces must be cleaned when the cumulative automatic zero compensation exceeds 4 percent opacity.
- (C) You must apply a method for producing a simulated zero opacity condition and an upscale (span) opacity condition using a certified neutral density filter or other related technique to produce a known obscuration of the light beam. All procedures applied must provide a system check of the analyzer internal optical surfaces and all electronic circuitry including the lamp and photodetector assembly.
- (D) Except during periods of system breakdowns, repairs, calibration checks, and zero and span adjustments, the COMS must be in continuous operation and must complete a minimum of one cycle of sampling and analyzing for each successive 10 second period and one cycle of data recording for each successive 6-minute period.
- (E) You must reduce all data from the COMS to 6-minute averages. Sixminute opacity averages must be calculated from 36 or more data points equally spaced over each 6-minute period. Data recorded during periods of system breakdowns, repairs, calibration checks, and zero and span adjustments must not be included in the data

Page 41 of 122

40 CFR Ch. I (7-1-12 Edition)

averages. An arithmetic or integrated average of all data may be used.

(iii) During each performance test conducted according to paragraph (o)(1) of this section, you must establish an opacity baseline level. The value of the opacity baseline level is determined by averaging all of the 6minute average opacity values (reported to the nearest 0.1 percent opacity) from the COMS measurements recorded during each of the test run intervals conducted for the performance test, and then adding 2.5 percent opacity to your calculated average opacity value for all of the test runs. If your opacity baseline level is less than 5.0 percent, then the opacity baseline level is set at 5.0 percent.

(iv) You must evaluate the preceding 24-hour average opacity level measured by the COMS each boiler operating day excluding periods of affected facility startup, shutdown, or malfunction. If the measured 24-hour average opacity emission level is greater than the baseline opacity level determined in paragraph (o)(2)(iii) of this section, you must initiate investigation of the relevant equipment and control systems within 24 hours of the first discovery of the high opacity incident and take the appropriate corrective action as soon as practicable to adjust control settings or repair equipment to reduce the measured 24-hour average opacity to a level below the baseline opacity level. In cases when a wet scrubber is used in combination with another PM control device that serves as the primary PM control device, the wet scrubber must be maintained and operated.

(v) You must record the opacity measurements, calculations performed, and any corrective actions taken. The record of corrective action taken must include the date and time during which the measured 24-hour average opacity was greater than baseline opacity level, and the date, time, and description of the corrective action.

(vi) If the measured 24-hour average opacity for your affected facility remains at a level greater than the opacity baseline level after 7 boiler operating days, then you must conduct a new PM performance test according to paragraph (o)(1) of this section and establish a new opacity baseline value

according to paragraph (o)(2) of this section. This new performance test must be conducted within 60 days of the date that the measured 24-hour average opacity was first determined to exceed the baseline opacity level unless a waiver is granted by the permitting authority.

(3) As an alternative to complying with the requirements of paragraph (0)(2) of this section, an owner or operator may elect to monitor the performance of an electrostatic precipitator (ESP) operated to comply with the applicable PM emissions limit in \$60.42Da(c)(2) or (d) using an ESP predictive model developed in accordance with the requirements in paragraphs (0)(3)(i) through (v) of this section.

(i) You must calibrate the ESP predictive model with each PM control device used to comply with the applicable PM emissions limit in §60.42Da(c)(2) or (d) operating under normal conditions. In cases when a wet scrubber is used in combination with an ESP to comply with the PM emissions limit, the wet scrubber must be maintained and operated.

(ii) You must develop a site-specific monitoring plan that includes a description of the ESP predictive model used, the model input parameters, and the procedures and criteria for establishing monitoring parameter baseline levels indicative of compliance with the PM emissions limit. You must submit the site-specific monitoring plan for approval by the permitting authority. For reference purposes in preparing the monitoring plan, see the OAQPS "Compliance Assurance Monitoring (CAM) Protocol for an Electrostatic Precipitator (ESP) Controlling Particulate Matter (PM) Emissions from a Coal-Fired Boiler." This document is available from the U.S. Environmental Protection Agency (U.S. EPA); Office of Air Quality Planning and Standards; Sector Policies and Programs Division; Measurement Policy Group (D243-02), Research Triangle Park, NC 27711. This document is also available on the Technology Transfer Network (TTN) under Emission Measurement Center Continuous Emission Monitoring.

(iii) You must run the ESP predictive model using the applicable input data

§60.48Da

each boiler operating day and evaluate the model output for the preceding boiler operating day excluding periods of affected facility startup, shutdown, or malfunction. If the values for one or more of the model parameters exceed the applicable baseline levels determined according to your approved sitespecific monitoring plan, you must initiate investigation of the relevant equipment and control systems within 24 hours of the first discovery of a model parameter deviation and, take the appropriate corrective action as soon as practicable to adjust control settings or repair equipment to return the model output to within the applicable baseline levels.

- (iv) You must record the ESP predictive model inputs and outputs and any corrective actions taken. The record of corrective action taken must include the date and time during which the model output values exceeded the applicable baseline levels, and the date, time, and description of the corrective
- (v) If after 7 consecutive days a model parameter continues to exceed the applicable baseline level, then you must conduct a new PM performance test according to paragraph (o)(1) of this section. This new performance test must be conducted within 60 calendar days of the date that the model parameter was first determined to exceed its baseline level unless a waiver is granted by the permitting authority.
- (4) As an alternative to complying with the requirements of paragraph (o)(2) of this section, an owner or operator may elect to monitor the performance of a fabric filter (baghouse) operated to comply with the applicable PM emissions limit in §60.42Da(c)(2) or (d) by using a bag leak detection system according to the requirements in paragraphs (o)(4)(i) through (v) of this section.
- (i) Each bag leak detection system must meet the specifications and requirements in paragraphs (o)(4)(i)(A)through (H) of this section.
- (A) The bag leak detection system must be certified by the manufacturer to be capable of detecting PM emissions at concentrations of 1 milligram per actual cubic meter (0.00044 grains per actual cubic foot) or less.

(B) The bag leak detection system sensor must provide output of relative PM loadings. The owner or operator must continuously record the output from the bag leak detection system using electronic or other means (e.g., using a strip chart recorder or a data logger.)

Filed: 02/21/2013

- (C) The bag leak detection system must be equipped with an alarm system that will react when the system detects an increase in relative particulate loading over the alarm set point established according to paragraph (o)(4)(i)(D) of this section, and the alarm must be located such that it can be noticed by the appropriate plant personnel.
- (D) In the initial adjustment of the bag leak detection system, you must establish, at a minimum, the baseline output by adjusting the sensitivity (range) and the averaging period of the device, the alarm set points, and the alarm delay time.
- (E) Following initial adjustment, you must not adjust the averaging period, alarm set point, or alarm delay time without approval from the permitting authority except as provided in paragraph (d)(1)(vi) of this section.
- (F) Once per quarter, you may adjust the sensitivity of the bag leak detection system to account for seasonal effects, including temperature and humidity, according to the procedures identified in the site-specific monitoring plan required by paragraph (o)(4)(ii) of this section.
- (G) You must install the bag leak detection sensor downstream of the fabric filter and upstream of any wet scrubber.
- (H) Where multiple detectors are required, the system's instrumentation and alarm may be shared among detec-
- (ii) You must develop and submit to the permitting authority for approval a site-specific monitoring plan for each bag leak detection system. You must operate and maintain the bag leak detection system according to the sitespecific monitoring plan at all times. Each monitoring plan must describe the items in paragraphs (o)(4)(ii)(A) through (F) of this section.
- (A) Installation of the bag leak detection system;

§60.48Da

40 CFR Ch. I (7-1-12 Edition)

(B) Initial and periodic adjustment of the bag leak detection system, including how the alarm set-point will be established:

Document #1421812

- (C) Operation of the bag leak detection system, including quality assurance procedures:
- (D) How the bag leak detection system will be maintained, including a routine maintenance schedule and spare parts inventory list;
- (E) How the bag leak detection system output will be recorded and stored;
- (F) Corrective action procedures as specified in paragraph (o)(4)(iii) of this section. In approving the site-specific monitoring plan, the permitting authority may allow owners and operators more than 3 hours to alleviate a specific condition that causes an alarm if the owner or operator identifies in the monitoring plan this specific condition as one that could lead to an alarm, adequately explains why it is not feasible to alleviate this condition within 3 hours of the time the alarm occurs. and demonstrates that the requested time will ensure alleviation of this condition as expeditiously as prac-
- (iii) For each bag leak detection system, you must initiate procedures to determine the cause of every alarm within 1 hour of the alarm. Except as provided in paragraph (o)(4)(ii)(F) of this section, you must alleviate the cause of the alarm within 3 hours of the alarm by taking whatever corrective action(s) are necessary. Corrective actions may include, but are not limited to the following:
- (A) Inspecting the fabric filter for air leaks, torn or broken bags or filter media, or any other condition that may cause an increase in particulate emissions:
- (B) Sealing off defective bags or filter
- (C) Replacing defective bags or filter media or otherwise repairing the control device:
- (D) Sealing off a defective fabric filter compartment:
- (E) Cleaning the bag leak detection system probe or otherwise repairing the bag leak detection system; or
- (F) Shutting down the process producing the particulate emissions.

- (iv) You must maintain records of the information specified in paragraphs (o)(4)(iv)(A) through (C) of this section for each bag leak detection system.
- (A) Records of the bag leak detection system output;
- (B) Records of bag leak detection system adjustments, including the date and time of the adjustment, the initial bag leak detection system settings, and the final bag leak detection system settings; and
- (C) The date and time of all bag leak detection system alarms, the time that procedures to determine the cause of the alarm were initiated, if procedures were initiated within 1 hour of the alarm, the cause of the alarm, an explanation of the actions taken, the date and time the cause of the alarm was alleviated, and if the alarm was alleviated within 3 hours of the alarm.
- (v) If after any period composed of 30 boiler operating days during which the alarm rate exceeds 5 percent of the process operating time (excluding control device or process startup, shutdown, and malfunction), then you must conduct a new PM performance test according to paragraph (o)(1) of this section. This new performance test must be conducted within 60 calendar days of the date that the alarm rate was first determined to exceed 5 percent limit unless a waiver is granted by the permitting authority.
- (5) An owner or operator of a modified affected facility electing to meet the emission limitations in §60.42Da(d) shall determine the percent reduction in PM by using the emission rate for PM determined by the performance test conducted according to the requirements in paragraph (o)(1) of this section and the ash content on a mass basis of the fuel burned during each performance test run as determined by analysis of the fuel as fired.
- (p) As an alternative to meeting the compliance provisions specified in paragraph (o) of this section, an owner or operator may elect to install, evaluate, maintain, and operate a CEMS measuring PM emissions discharged from the affected facility to the atmosphere and record the output of the system as specified in paragraphs (p)(1) through (p)(8) of this section.

40 CFR Ch. I (7-1-12 Edition)

- (2) Safe sampling platform(s);
- (3) Safe access to sampling platform(s);
- (4) Utilities for sampling and testing equipment; and
- (5) Any other facilities that the Administrator deems necessary for safe and adequate testing of a source.
- (e) Conduct of performance tests. (1) Performance tests shall be conducted under such conditions as the Administrator specifies to the owner or operator based on representative performance (i.e., performance based on normal operating conditions) of the affected source. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a performance test, nor shall emissions in excess of the level of the relevant standard during periods of startup, shutdown, and malfunction be considered a violation of the relevant standard unless otherwise specified in the relevant standard or a determination of noncompliance is made under §63.6(e). Upon request, the owner or operator shall make available to the Administrator such records as may be necessary to determine the conditions of performance tests.
- (2) Performance tests shall be conducted and data shall be reduced in accordance with the test methods and procedures set forth in this section, in each relevant standard, and, if required, in applicable appendices of parts 51, 60, 61, and 63 of this chapter unless the Administrator—
- (i) Specifies or approves, in specific cases, the use of a test method with minor changes in methodology (see definition in §63.90(a)). Such changes may be approved in conjunction with approval of the site-specific test plan (see paragraph (c) of this section); or
- (ii) Approves the use of an intermediate or major change or alternative to a test method (see definitions in §63.90(a)), the results of which the Administrator has determined to be adequate for indicating whether a specific affected source is in compliance; or
- (iii) Approves shorter sampling times or smaller sample volumes when necessitated by process variables or other factors; or

- (iv) Waives the requirement for performance tests because the owner or operator of an affected source has demonstrated by other means to the Administrator's satisfaction that the affected source is in compliance with the relevant standard.
- (3) Unless otherwise specified in a relevant standard or test method, each performance test shall consist of three separate runs using the applicable test method. Each run shall be conducted for the time and under the conditions specified in the relevant standard. For the purpose of determining compliance with a relevant standard, the arithmetic mean of the results of the three runs shall apply. Upon receiving approval from the Administrator, results of a test run may be replaced with results of an additional test run in the event that—
- (i) A sample is accidentally lost after the testing team leaves the site; or
- (ii) Conditions occur in which one of the three runs must be discontinued because of forced shutdown; or
- (iii) Extreme meteorological conditions occur; or
- (iv) Other circumstances occur that are beyond the owner or operator's control
- (4) Nothing in paragraphs (e)(1) through (e)(3) of this section shall be construed to abrogate the Administrator's authority to require testing under section 114 of the Act.
- (f) Use of an alternative test method—(1)General. Until authorized to use an intermediate or major change or alternative to a test method, the owner or operator of an affected source remains subject to the requirements of this section and the relevant standard.
- (2) The owner or operator of an affected source required to do performance testing by a relevant standard may use an alternative test method from that specified in the standard provided that the owner or operator—
- (i) Notifies the Administrator of his or her intention to use an alternative test method at least 60 days before the performance test is scheduled to begin;
- (ii) Uses Method 301 in appendix A of this part to validate the alternative test method. This may include the use of specific procedures of Method 301 if use of such procedures are sufficient to

40 C.F.R. § 63.7(f)

§ 63.7

validate the alternative test method; and

(iii) Submits the results of the Method 301 validation process along with thnotification of intention and the justification for not using the specified test method. The owner or operator may submit the information required in this paragraph well in advance of the deadline specified in paragraph (f)(2)(i) of this section to ensure a timely review by the Administrator in order to meet the performance test date specified in this section or the relevant standard.

(3) The Administrator will determine whether the owner or operator's validation of the proposed alternative test method is adequate and issue an approval or disapproval of the alternative test method. If the owner or operator intends to demonstrate compliance by using an alternative to any test method specified in the relevant standard, the owner or operator is authorized to conduct the performance test using an alternative test method after the Administrator approves the use of the alternative method. However, the owner or operator is authorized to conduct the performance test using an alternative method in the absence of notification of approval/disapproval 45 days after submission of the request to use an alternative method and the request satisfies the requirements in paragraph (f)(2) of this section. The owner or operator is authorized to conduct the performance test within 60 calendar days after he/she is authorized to demonstrate compliance using an alternative test method. Notwithstanding the requirements in the preceding three sentences, the owner or operator may proceed to conduct the performance test as required in this section (without the Administrator's prior approval of the site-specific test plan) if he/she subsequently chooses to use the specified testing and monitoring methods instead of an alternative.

(4) If the Administrator finds reasonable grounds to dispute the results obtained by an alternative test method for the purposes of demonstrating compliance with a relevant standard, the Administrator may require the use of a test method specified in a relevant standard.

- (5) If the owner or operator uses an alternative test method for an affected source during a required performance test, the owner or operator of such source shall continue to use the alternative test method for subsequent performance tests at that affected source until he or she receives approval from the Administrator to use another test method as allowed under §63.7(f).
- (6) Neither the validation and approval process nor the failure to validate an alternative test method shall abrogate the owner or operator's responsibility to comply with the requirements of this part.
- (g) Data analysis, recordkeeping, and reporting. (1) Unless otherwise specified in a relevant standard or test method. or as otherwise approved by the Administrator in writing, results of a performance test shall include the analysis of samples, determination of emissions, and raw data. A performance test is "completed" when field sample collection is terminated. The owner or operator of an affected source shall report the results of the performance test to the Administrator before the close of business on the 60th day following the completion of the performance test, unless specified otherwise in a relevant standard or as approved otherwise in writing by the Administrator (see §63.9(i)). The results of the performance test shall be submitted as part of the notification of compliance status required under §63.9(h). Before a title V permit has been issued to the owner or operator of an affected source, the owner or operator shall send the results of the performance test to the Administrator. After a title V permit has been issued to the owner or operator of an affected source, the owner or operator shall send the results of the performance test to the appropriate permitting authority.
 - (2) [Reserved]
- (3) For a minimum of 5 years after a performance test is conducted, the owner or operator shall retain and make available, upon request, for inspection by the Administrator the records or results of such performance test and other data needed to determine emissions from an affected source.

40 CFR Ch. I (7-1-12 Edition)

use of Method 25A, 40 CFR part 60, appendix A shall conform with the requirements in paragraphs (j)(1) and (j)(2) of this section.

- (1) The organic HAP used as the calibration gas for Method 25A, 40 CFR part 60, appendix A shall be the single organic HAP representing the largest percent by volume of emissions.
- (2) The use of Method 25A, 40 CFR part 63, appendix A is acceptable if the response from the high-level calibration gas is at least 20 times the standard deviation of the response from the zero calibration gas when the instrument is zeroed on the most sensitive scale.
- (k) An owner or operator using a flare to comply with the requirements of this section shall conduct a compliance demonstration as specified in §63.504(c).
- (1) When the term "equipment" is used in subpart H of this part, the definition of this term in §63.482(b) shall apply for the purposes of this subpart.
- (m) The phrase "the provisions of subparts F, I, or U of this part" shall apply instead of the phrase "the provisions of subpart F or I of this part" throughout §§ 63.163 and 63.168, for the purposes of this subpart. In addition, the phrase "subparts F, I, and U" shall apply instead of the phrase "subparts F and I" in §63.174(c)(2)(iii), for the purposes of this subpart.
- (n) Heat exchange system provisions. The owner or operator of each affected source shall comply with the requirements of §63.104 for heat exchange systems, with the exceptions noted in paragraphs (n)(1) through (n)(5) of this
- (1) When the term "chemical manufacturing process unit" is used in §63.104, the term "elastomer product process unit" (or EPPU) shall apply for the purposes of this subpart, with the exception noted in paragraph (n)(2) of this section.
- (2) When the phrase "a chemical manufacturing process unit meeting the conditions of §63.100(b)(1) through (b)(3) of this subpart, except for chemical manufacturing process units meeting the condition specified in §63.100(c) of this subpart" is used in §63.104(a), the term "an EPPU, except for EPPUs meeting the condition specified in

§63.480(b)" shall apply for the purposes of this subpart.

Filed: 02/21/2013

- (3) When §63.104 refers to Table 4 of subpart F of this part or Table 9 of subpart G of this part, the owner or operator is only required to consider organic HAP listed on Table 5 of this sub-
- (4) When $\S63.104(c)(3)$ specifies the monitoring plan retention requirements, and when §63.104(f)(1) refers to the record retention requirements in §63.103(c)(1), the requirements in §63.506(a) and §63.506(h) shall apply, for the purposes of this subpart.
- (5) When §63.104(f)(2) requires information to be reported in the Periodic Reports required by §63.152(c), the owner or operator shall instead report information specified $\S63.104(f)(2)$ in the Periodic Reports required by §63.506(e)(6), for the purposes of this subpart.
- (6) The compliance date for heat exchange systems subject to the provisions of this section is specified in §63.481(d)(6).

[62 FR 46925, Sept. 5, 1996, as amended at 62 FR 37722, July 15, 1997; 65 FR 38070, June 19, 2000; 76 FR 22592, Apr. 21, 2011]

§63.503 Emissions averaging provisions.

- (a) This section applies to owners or operators of existing affected sources who seek to comply with §63.483(b) by using emissions averaging rather than following the provisions of §§ 63.484, 63.485, 63.486, 63.494, and 63.501.
- (1) The following emission point limitations apply to the use of these provisions:
- (i) All emission points included in an emissions average shall be from the same affected source. There may be an emissions average for each individual affected source located at a plant site.
- (ii)(A) If a plant site has only one affected source for which emissions averaging is being used to demonstrate compliance, the number of emission points allowed to be included in the emission average is limited to twenty. This number may be increased by up to five additional points if pollution prevention measures are used to control five or more of the emission points included in the emissions average.

§ 63.503

- (B) If a plant site has two or more affected sources for which emissions averaging is being used to demonstrate compliance, the number of emission points allowed in the emissions average for those affected sources is limited to twenty. This number may be increased by up to five additional emission points if pollution prevention measures are used to control five or more of the emission points included in the emissions averages.
- (2) Compliance with the provisions of this section may be based on either organic HAP or TOC.
- (3) For the purposes of the provisions in this section, whenever Method 18, 40 CFR part 60, appendix A, is specified within the paragraphs of this section or is specified by reference through provisions outside this section, Method 18 or Method 25A, 40 CFR part 60, appendix A, may be used. The use of Method 25A, 40 CFR part 60, appendix A, shall conform with the requirements in paragraphs (a)(3)(i) and (a)(3)(ii) of this section.
- (i) The organic HAP used as the calibration gas for Method 25A, 40 CFR part 60, appendix A, shall be the single organic HAP representing the largest percent by volume of the emissions.
- (ii) The use of Method 25A, 40 CFR part 60, appendix A, is acceptable if the response from the high-level calibration gas is at least 20 times the standard deviation of the response from the zero calibration gas when the instrument is zeroed on the most sensitive scale.
- (b) Unless an operating permit application has been submitted, the owner or operator shall develop and submit for approval an Emissions Averaging Plan containing all of the information required in §63.506(e)(4) for all emission points to be included in an emissions average.
- (c) Paragraphs (c)(1) through (c)(4) of this section describe the emission points that may be used to generate emissions averaging credits if control was applied after November 15, 1990 and if sufficient information is available to determine the appropriate value of credits for the emission point. Paragraph (c)(5) of this section discusses the use of pollution prevention in generating emissions averaging credits.

- (1) Storage vessels, batch front-end process vents, aggregate batch vent streams, continuous front-end process vents, and process wastewater streams that are determined to be Group 2 emission points.
- (2) Storage vessels, continuous frontend process vents, and process wastewater steams that are determined to be Group 1 emission points and that are controlled by a technology that the Administrator or permitting authority agrees has a higher nominal efficiency than the reference control technology. Information on the nominal efficiencies for such technologies shall be submitted and approved as provided in paragraph (i) of this section.
- (3) Batch front-end process vents and aggregate batch vent streams that are determined to be Group 1 emission points and that are controlled to a level more stringent than the applicable standard.
- (4) Back-end process operations that are controlled such that organic HAP emissions from the back-end process operation are less than would be achieved by meeting the residual organic HAP limits in §63.494. For the purposes of the emission averaging provisions in this section, all back-end process operations at an affected facility shall be considered a single emission point.
- (5) The percent reduction for any storage vessel, batch front-end process vent, aggregate batch vent stream, continuous front-end process vent, and process wastewater stream shall be determined using the procedures specified in paragraph (j) of this section.
- (i) For a Group 1 storage vessel, batch front-end process vent, aggregate batch vent stream, continuous front-end process vent, or process waste-water stream, the pollution prevention measure shall reduce emissions more than if the reference control technology or standard had been applied to the emission point instead of the pollution prevention measure, except as provided in paragraph (c)(5)(ii) of this section.
- (ii) If a pollution prevention measure is used in conjunction with other controls for a Group 1 storage vessel, batch front-end process vent, aggregate batch vent stream, continuous front-

40 CFR Ch. I (7-1-12 Edition)

end process vent, or process waste-water stream, the pollution prevention measure alone does not have to reduce emissions more than the reference control technology or standard, but the combination of the pollution prevention measure and other controls shall reduce emissions more than if the applicable reference control technology or standard had been applied instead of the pollution prevention measure.

- (d) The following emission points cannot be used to generate emissions averaging credits:
- (1) Emission points already controlled on or before November 15, 1990 cannot be used to generate credits unless the level of control was increased after November 15, 1990. In this case, credit will be allowed only for the increase in control after November 15, 1990.
- (2) Group 1 emission points, identified in paragraph (c)(2) of this section, that are controlled by a reference control technology cannot be used to generate credits unless the reference control technology has been approved for use in a different manner and a higher nominal efficiency has been assigned according to the procedures in paragraph (i) of this section.
- (3) Emission points on nonoperating EPPU cannot be used to generate credits. EPPU that are shutdown cannot be used to generate credits or debits.
- (4) Maintenance wastewater cannot be used to generate credits. Wastewater streams treated in biological treatment units cannot be used to generate credits. These two types of wastewater cannot be used to generate credits or debits. For the purposes of this section, the terms wastewater and wastewater stream are used to mean process wastewater.
- (5) Emission points controlled to comply with a State or Federal rule other than this subpart cannot be used to generate credits, unless the level of control has been increased after November 15, 1990 to a level above what is required by the other State or Federal rule. Only the control above what is required by the other State or Federal rule will be credited. However, if an emission point has been used to generate emissions averaging credit in an approved emissions average, and the

emission point is subsequently made subject to a State or Federal rule other than this subpart, the emission point may continue to generate emissions averaging credit for the purpose of complying with the previously approved emissions average.

- (e) For all emission points included in an emissions average, the owner or operator shall perform the following tasks:
- (1) Calculate and record monthly debits for all Group 1 emission points that are controlled to a level less stringent than the reference control technology or standard for those emission points. The Group 1 emission points are identified in paragraphs (c)(2) through (c)(4) of this section. Equations in paragraph (g) of this section shall be used to calculate debits.
- (2) Calculate and record monthly credits for all Group 1 and Group 2 emission points that are overcontrolled to compensate for the debits. Equations in paragraph (h) of this section shall be used to calculate credits. Emission points and controls that meet the criteria of paragraph (c) of this section may be included in the credit calculation, whereas those described in paragraph (d) of this section shall not be included.
- (3) Demonstrate that annual credits calculated according to paragraph (h) of this section are greater than or equal to debits calculated for the same annual compliance period according to paragraph (g) of this section.
- (i) The owner or operator may choose to include more than the required number of credit-generating emission points in an emissions average in order to increase the likelihood of being in compliance.
- (ii) The initial demonstration in the Emissions Averaging Plan or operating permit application that credit-generating emission points will be capable of generating sufficient credits to offset the debits from the debit-generating emission points shall be made under representative operating conditions. After the compliance date, actual operating data shall be used for all debit and credit calculations.
- (4) Demonstrate that debits calculated for a quarterly (3-month) period according to paragraph (g) of this

§ 63.503

section are not more than 1.30 times the credits for the same period calculated according to paragraph (h) of this section. Compliance for the quarter shall be determined based on the ratio of credits and debits from that quarter, with 30 percent more debits than credits allowed on a quarterly basis.

- (5) Record and report quarterly and annual credits and debits in the Peri-Reports as specified odic in §63.506(e)(6). Every fourth Periodic Report shall include a certification of compliance with the emissions averaging provisions as required 63.506(e)(6)(x)(C)(2).
- (f) Debits and credits shall be calculated in accordance with the methods and procedures specified in paragraphs (g) and (h) of this section, respectively, and shall not include emissions during the following periods:
 - (1) [Reserved]
- (2) Emissions during periods of monitoring excursions, as defined in §63.505 (g) or (h). For these periods, the cal-

culation of monthly credits and debits shall be adjusted as specified in paragraphs (f)(2)(i) through (f)(2)(iii) of this section.

- (i) No credits would be assigned to the credit-generating emission point.
- (ii) Maximum debits would be assigned to the debit-generating emission point.
- (iii) The owner or operator may demonstrate to the Administrator that full or partial credits or debits should be assigned using the procedures in paragraph (1) of this section.
- (g) Debits are generated by the difference between the actual emissions from a Group 1 emission point that is uncontrolled or is controlled to a level less stringent than the applicable reference control technology or standard and the emissions allowed for the Group 1 emission point. Debits shall be calculated as follows:
- (1) Source-wide debits shall be calculated using Equation 33. Debits and all terms of the equation are in units of megagrams per month (Mg/month):

$$\begin{split} \text{Debits} &= \sum_{i=1}^{n} \left(\text{ECFEPV}_{\text{iACTUAL}} - (0.02) \text{ ECFEPV}_{\text{iu}} \right) + \sum_{i=1}^{n} \left(\text{ES}_{\text{iACTUAL}} - (0.05) \text{ ES}_{\text{iu}} \right) \\ &+ \left(\text{EBEP}_{\text{ACTUAL}} - \text{EBEP}_{\text{c}} \right) + \sum_{i=1}^{n} \left(\text{EWW}_{\text{iACTUAL}} - \text{EWW}_{\text{ic}} \right) \\ &+ \sum_{i=1}^{n} \left(\text{EBFEPV}_{\text{iACTUAL}} - (0.1) \text{ EBFEPV}_{\text{iu}} \right) + \sum_{i=1}^{n} \left(\text{EABV}_{\text{iACTUAL}} - (0.1) \text{ EABV}_{\text{iu}} \right) \end{split} \quad \text{[Eq. 33]}$$

Where:

 $ECFEPV_{iACTUAL}$ = Emissions from each Group 1 continuous front-end process vent i that is uncontrolled or is controlled to a level less stringent than the applicable reference control technology. ECFEPViACTUAL is calculated according to paragraph (g)(2)(iii) of this section.

(0.02)ECFEPV_{iu} = Emissions from each Group 1 continuous front-end process vent i if the applicable reference control technology had been applied to the uncontrolled emissions. ECFEPV_{in} is calculated according to paragraph (g)(2)(ii) of this section.

 $ES_{iACTUAL}$ = Emissions from each Group 1 storage vessel i that is uncontrolled or is controlled to a level less stringent than the applicable reference control technology or standard. ESiactual is calculated according to paragraph (g)(3) of this section.

 $(0.05)ES_{iu}$ = Emissions from each Group 1 storage vessel i if the applicable reference control technology or standard had been applied to the uncontrolled emissions. ESiu is calculated according to paragraph (g)(3) of this section.

EBEP_{ACTUAL} = Emissions from back-end process operations that do not meet the residorganic HAP limits in §63.494. EBEPACTUAL is calculated according to paragraph (g)(4)(i) of this section.

EBEP_c = Emissions from back-end process operations if the residual organic HAP limits in §63.494(a) were met. EBEPc is calculated according to paragraph (g)(4)(ii) of this section.

 $EWW_{iACTUAL}$ = Emissions from each Group 1 wastewater stream i that is uncontrolled

40 CFR Ch. I (7-1-12 Edition)

or is controlled to a level less stringent than the applicable reference control technology. $EWW_{iACTUAL}$ is calculated according to paragraph (g)(5) of this section.

 EWW_{ic} = Emissions from each Group 1 wastewater stream i if the reference control technology had been applied to the uncontrolled emissions. EWWic is calculated according to paragraph (g)(5) of this section.

- $EBFEPV_{iACTUAL}$ = Emissions from each Group 1 batch front-end process vent stream i that is uncontrolled or is controlled to a level less stringent than the applicable standard. EBFEPV_{iACTUAL} is calculated according to paragraph (g)(6)(ii) of this section.
- (0.1) EBFEPV_{iu} = Emissions from each Group 1 batch front-end process vent i if the applicable standard had been applied to the uncontrolled emissions. EBFEPViu is calculated according to paragraph (g)(6)(i) of this section.
- $EABV_{iACTUAL}$ = Emissions from each Group 1 aggregate batch vent stream i that is uncontrolled or is controlled to a level less stringent than the applicable standard. EABViactual is calculated according to paragraph (g)(7)(iii) of this section.
- (0.1) EABV_{iu} = Emissions from each Group 1 aggregate batch vent stream i if the appli-

cable standard had been applied to the uncontrolled emissions. EABV_{in} is calculated according to paragraph (g)(7)(ii) of this sec-

- n = The number of emission points being included in the emissions average.
- (2) Emissions from continuous frontend process vents shall be calculated as follows:
- (i) For purposes of determining continuous front-end process vent stream flow rate, organic HAP concentrations, and temperature, the sampling site shall be after the final product recovery device, if any recovery devices are present; before any control device (for continuous front-end process vents, recovery devices shall not be considered control devices); and before discharge to the atmosphere. Method 1 or 1A of 40 CFR part 60, appendix A, shall be used for selection of the sampling site.
- (ii) ECFEPV_{iu} for each continuous front-end process vent i shall be calculated using Equation 34.

ECFEPV_{iu} =
$$(2.494 \times 10^{-9})$$
Qh $\left(\sum_{i=1}^{n} C_{j} M_{j}\right)$ [Eq. 34]

where:

ECFEPViu=Uncontrolled continuous frontend process vent emission rate from continuous front-end process vent i, Mg/ month.

Q=Vent stream flow rate, dry standard m³/ min, measured using Method 2, 2A, 2C, or 2D of 40 CFR part 60, appendix A, as appro-

h=Monthly hours of operation during which positive flow is present in the continuous front-end process vent, hr/month.

Ci=Concentration, ppmv, dry basis, of organic HAP j as measured by Method 18 or Method 25A of 40 CFR part 60, appendix A. M_i=Molecular weight of organic HAP j, gram per gram-mole.

n=Number of organic HAP in stream.

(A) The values of Q and C_i shall be determined during a performance test conducted under representative operating conditions. The values of Q and Ci shall be established in the Notification of Compliance Status and shall be up-

dated as provided in paragraph (g)(2)(ii)(B) of this section.

(B) If there is a change in capacity utilization other than a change in monthly operating hours, or if any other change is made to the process or product recovery equipment or operation such that the previously measured values of Q and C_j are no longer representative, a new performance test shall be conducted to determine new representative values of Q and Ci. These new values shall be used to calculate debits and credits from the time of the change forward, and the new values shall be reported in the next Periodic Report.

(iii) The following procedures and equations shall be used to calculate ECFEPViactual:

(A) If the continuous front-end process vent is not controlled by a control device or pollution prevention measure, $\text{ECFEPV}_{iACTUAL} = \text{ECFEPV}_{iu}$, where

§ 63.503

ECFEPV $_{iu}$ is calculated according to the procedures contained in paragraphs (g)(2)(i) and (g)(2)(ii) of this section.

(B) If the continuous front-end process vent is controlled using a control

device or a pollution prevention measure achieving less than 98-percent reduction, ECFEPV_{iACTUAL} is calculated using Equation 35.

$$ECFEPV_{iACTUAL} = ECFEPV_{iu} \times \left(1 - \frac{Percent\ reduction}{100\%}\right)$$
 [Eq. 35]

Where:

ECFEPV_{iACTUAL} = Emissions from each Group 1 continuous front-end process vent i that is uncontrolled or is controlled to a level less stringent than the reference control technology.

ECFEPV_{iu} = Uncontrolled continuous frontend process vent emission rate from continuous front-end process vent i, Mg/ month.

- (1) The percent reduction shall be measured according to the procedures in §63.116 if a combustion control device is used. For a flare meeting the criteria in §63.116(a), or a boiler or process heater meeting the criteria in §63.116(b), the percent reduction shall be 98 percent. If a noncombustion control device is used, percent reduction shall be demonstrated by a performance test at the inlet and outlet of the device, or, if testing is not feasible, by a control design evaluation and documented engineering calculations.
- (2) For determining debits from Group 1 continuous front-end process vents, product recovery devices shall not be considered control devices and

shall not be assigned a percent reduction in calculating ECFEPV_{iACTUAL}. The sampling site for measurement of uncontrolled emissions shall be after the final uncontrolled recovery device. However, as provided in §63.113(a)(3), a Group 1 continuous front-end process vent may add sufficient product recovery to raise the TRE index value above 1.0, thereby becoming a Group 2 continuous front-end process vent. Such a continuous front-end process vent is not a Group 1 continuous front-end process vent and should, therefore, not be included in determining debits under this paragraph.

- (3) Procedures for calculating the percent reduction of pollution prevention measures are specified in paragraph (j) of this section.
- (3) Emissions from storage vessels shall be calculated using the procedures specified in §63.150(g)(3).
- (4) Emissions from back-end process operations shall be calculated as follows:
- (i) Equation 36 shall be used to calculate $\mathrm{EBEP}_{\mathrm{ACTUAL}};$

$$EBEP_{ACTUAL} = (1,000) \sum_{i=1}^{n} (C_i)(P_i)$$
 [Eq. 36]

where:

$$\begin{split} EBEP_{ACTUAL} &= Actual \ emissions \ from \ backend \ process \ operations, \ Mg/month. \end{split}$$

- C_i = Residual organic HAP content of sample i, kg organic HAP per Mg latex or dry crumb rubber.
- P_i = Weight of latex or dry crumb rubber leaving the stripper represented by sample i, Mg.
- (ii) Equation 37 shall be used to calculate EBEPc:

40 CFR Ch. I (7-1-12 Edition)

$$EBEP_{C} = (1,000)(HAP_{limit})(P_{month}) \qquad [Eq. 37]$$

where:

 $EBEP_c$ = Emissions from back-end process operations if the residual organic HAP limits in §63.494(a) were met, Mg/month.

 HAP_{limit} = Residual organic HAP limits in §63.494 of this subpart, kg organic HAP per Mg latex or dry crumb rubber.

 P_{month} =Weight of latex or dry crumb rubber leaving the stripper in the month, Mg.

- (5) Emissions from wastewater shall be calculated using the procedures specified in $\S63.150(g)(5)$.
- (6) Emissions from batch front-end process vents shall be calculated as follows:
- (i) EBFEPV $_{iu}$ for each batch front-end process vent i shall be calculated using the procedures specified in §63.488(b).

- (ii) The following procedures and equations shall be used to determine EBFEPV_{iACTUAL}:
- (A) If the batch front-end process vent is not controlled by a control device or pollution prevention measure, $EBFEPV_{iACTUAL} = EBFEPV_{iu},$ EBFEPViu is calculated according to the procedures in §63.488(b).
- (B) If the batch front-end process vent is controlled using a control device or a pollution prevention measure achieving less than 90 percent reduction for the batch cycle, calculate $EBFEPV_{iACTUAL} \quad using \quad Equation \quad 38,$ where percent reduction is for the batch cycle.

$$EBFEPV_{iACTUAL} = EBFEPV_{iu} \times \left(1 - \frac{Percent\ reduction}{100\%}\right) \qquad [Eq.\ 38]$$

- (1) The percent reduction for the batch cycle shall be measured according to the procedures in $\S63.490(c)(2)$.
- (2) The percent reduction for control devices shall be calculated according to the procedures in $\S63.490$ (c)(2)(i) through (c)(2)(iii).
- (3) The percent reduction of pollution prevention measures shall be calculated using the procedures specified in paragraph (j) of this section.
- (7) Emissions from aggregate batch vents shall be calculated as follows:
- (i) For purposes of determining aggregate batch vent stream flow rate, organic HAP concentrations, and temperature, the sampling site shall be before any control device and before discharge to the atmosphere. Method 1 or 1A of 40 CFR part 60, appendix A, shall be used for selection of the sampling
- (ii) EABV_{iu} for each aggregate batch vent i shall be calculated using Equation 39.

EABV_{iu} =
$$(2.494 \times 10^{-9})$$
 Qh $\left(\sum_{j=1}^{n} C_{j} M_{j}\right)$ [Eq. 39]

where:

 $EABV_{iu} \\ = \\ Uncontrolled \ aggregate \ batch \ vent$ emission rate from aggregate batch vent i,

Q=Vent stream flow rate, dry standard cubic meters per minute, measured using Method 2, 2A, 2C, or 2D of 40 CFR part 60, appendix A, as appropriate.

h=Monthly hours of operation during which positive flow is present from the aggregate batch vent stream, hr/month.

Cj=Concentration, ppmv, dry basis, of organic HAP j as measured by Method 18 of 40 CFR part 60, appendix A.

 M_j =Molecular weight of organic HAP j, gram per gram-mole.

n=Number of organic HAP in the stream.

§ 63.503

(A) The values of Q and C_j shall be determined during a performance test conducted under representative operating conditions. The values of Q and C_j shall be established in the Notification of Compliance Status and shall be updated as provided in paragraph (g)(7)(ii)(B) of this section.

(B) If there is a change in capacity utilization other than a change in monthly operating hours, or if any other change is made to the process or product recovery equipment or operation such that the previously measured values of Q and C_j are no longer representative, a new performance test shall be conducted to determine new representative values of Q and C_j. These new values shall be used to calculate debits and credits from the time

of the change forward, and the new values shall be reported in the next Periodic Report.

(iii) The following procedures and equations shall be used to calculate ${\rm EABV_{iACTUAL}};$

(A) If the aggregate batch vent is not controlled by a control device or pollution prevention measure, EABV $_{ia}$ (actual = EABV $_{iu}$, where EABV $_{iu}$ is calculated according to the procedures in paragraphs (g)(7)(i) and (g)(7)(ii) of this section.

(B) If the aggregate batch vent stream is controlled using a control device or a pollution prevention measure achieving less than 90 percent reduction, calculate EABV $_{\rm iACTUAL}$ using Equation 40.

$$EABV_{iACTUAL} = EABV_{iu} \times \left(1 - \frac{Percent\ reduction}{100\%}\right)$$
 [Eq. 40]

(1) The percent reduction for control devices shall be determined according to the procedures in §63.490(e).

(2) The percent reduction of pollution prevention measures shall be calculated according to the procedures specified in paragraph (j) of this section.

(h) Credits are generated by the difference between emissions that are allowed for each Group 1 and Group 2 emission point and the actual emis-

sions from that Group 1 or Group 2 emission point that has been controlled after November 15, 1990 to a level more stringent than what is required by this subpart or any other State or Federal rule or statute. Credits shall be calculated as follows:

(1) Source-wide credits shall be calculated using Equation 41. Credits and all terms of the equation are in units of Mg/month, and the baseline date is November 15, 1990:

$$\begin{split} &\operatorname{Credits} = D\sum_{i=1}^{n} \left((0.02) \operatorname{ECFEPV1}_{iu} - \operatorname{ECFEPV1}_{iACTUAL} \right) + D\sum_{i=1}^{m} \left(\operatorname{ECFEPV2}_{iBASE} - \operatorname{ECFEPV2}_{iACTUAL} \right) \\ &+ D\sum_{i=1}^{n} \left((0.05) \operatorname{ES1}_{iu} - \operatorname{ES1}_{iACTUAL} \right) + D\sum_{i=1}^{m} \left(\operatorname{ES2}_{iBASE} - \operatorname{ES2}_{iACTUAL} \right) + D \left(\operatorname{EBEP}_{e} \right) - \left(\operatorname{EBEP}_{ACTUAL} \right) \\ &+ D\sum_{i=1}^{n} \left(\operatorname{EWW1}_{ie} - \operatorname{EWW1}_{iACTUAL} \right) + D\sum_{i=1}^{m} \left(\operatorname{EWW2}_{iBASE} - \operatorname{EWW2}_{iACTUAL} \right) \\ &+ D\sum_{i=1}^{n} \left((0.1) \operatorname{EBFEPV1}_{iu} - \operatorname{EBFEPV1}_{iACTUAL} \right) + D\sum_{i=1}^{n} \left((0.1) \operatorname{EABV1}_{iu} - \operatorname{EABV1}_{iACTUAL} \right) \\ &+ D\sum_{i=1}^{m} \left(\operatorname{EBFEPV2}_{iBASE} - \operatorname{EBFEPV2}_{iACTUAL} \right) + D\sum_{i=1}^{m} \left(\operatorname{EABV2}_{iBASE} - \operatorname{EABV2}_{iACTUAL} \right) \end{split}$$

Where:

D = Discount factor = 0.9 for all credit generating emission points, except those controlled by a pollution prevention measure; discount factor = 1.0 for each credit generating emission point controlled by a pollution prevention measure (i.e., no discount provided).

Document #1421812

- ECFEPV1_{iACTUAL} = Emissions for each Group 1 continuous front-end process vent i that is controlled to a level more stringent than reference control technology. ECFEPV1_{iACTUAL} is calculated according to paragraph (h)(2)(ii) of this section.
- (0.02)ECFEPV1_{iu} = Emissions from each Group 1 continuous front-end process vent i if the reference control technology had been applied to the uncontrolled emissions. ECFEPV1iu is calculated according to paragraph (h)(2)(i) of this section.
- $ECFEPV2_{iACTUAL} = Emissions$ Group 2 continuous front-end process vent i that is controlled. ECFEPV2iACTUAL is calculated according to paragraph (h)(2)(iii) of
- ECFEPV2_{iBASE} = Emissions from each Group 2 continuous front-end process vent i at the baseline date. $ECFEPV2_{iBASE}$ is calculated in paragraph (h)(2)(iv) of this sec-
- $ES1_{iACTUAL}$ = Emissions from each Group 1 storage vessel i that is controlled to a level more stringent than the reference control technology or standard. ES1;ACTUAL is calculated according to paragraph (h)(3) of this section.
- (0.05) ES1_{iu} = Emissions from each Group 1 storage vessel i if the reference control technology had been applied to the uncontrolled emissions. ES1iu is calculated according to paragraph (h)(3) of this section.
- $ES2_{iACTUAL}$ = Emissions from each Group 2 storage vessel i that is controlled. $\mathrm{ES2}_{i\mathrm{ACTUAL}}$ is calculated according to paragraph (h)(3) of this section.
- $ES2_{iBASE}$ = Emissions from each Group 2 storage vessel i at the baseline date. $\mathrm{ES2}_{iBASE}$ is calculated in paragraph (h)(3) of this sec-
- $EBEP_{ACTUAL}$ = Actual emissions from back- $\begin{array}{lll} \text{end} & \text{process} & \text{operations,} & \text{Mg/month.} \\ \text{EBEP}_{\text{ACTUAL}} & \text{is} & \text{calculated} & \text{in} & \text{paragraph} \\ \end{array}$ (h)(4)(i) of this section.
- EBEP_c = Emissions from back-end process operations if the residual organic HAP limits in §63.494(a) were met, Mg/month. EBEPc is calculated in paragraph (h)(4)(ii) of this section.
- $EWW1_{iACTUAL} = Emissions \ from \ each \ Group \ 1$ wastewater stream i that is controlled to a level more stringent than the reference control technology. EWW1_{iACTUAL} is calculated according to paragraph (h)(5) of this section.
- $EWW1_{ic}$ = Emissions from each Group 1 wastewater stream i if the reference control technology had been applied to the un-

40 CFR Ch. I (7-1-12 Edition)

- controlled emissions. EWW1:, is calculated according to paragraph (h)(5) of this section.
- $EWW2_{iACTUAL}$ = Emissions from each Group 2 wastewater stream i that is controlled. $EWW2_{iACTUAL}$ is calculated according to paragraph (h)(5) of this section.
- EWW2_{iBASE} = Emissions from each Group 2 wastewater stream i at the baseline date. EWW2_{iBASE} is calculated according to paragraph (h)(5) of this section.
- (0.1) EBFEPV1_{in} = Emissions from each Group 1 batch front-end process vent i if the applicable standard had been applied to the uncontrolled emissions. EBFEPV1_{iu} is calculated according to paragraph (h)(6)(i) of this section.
- $EBFEPV1_{iACTUAL}$ = Emissions from each Group 1 batch front-end process vent i that is controlled to a level more stringent than the applicable standard. EBFEPV1_{iACTUAL} is calculated according to paragraph (h)(6)(ii) of this section.
- $(0.1)EABV1_{iu}$ = Emissions from each Group 1 aggregate batch vent stream i if the applicable standard had been applied to the uncontrolled emissions. EABV1_{iu} is calculated according to paragraph (h)(7)(i) of this section.
- $EABV1_{iACTUAL}$ = Emissions from each Group 1 aggregate batch vent stream i that is controlled to a level more stringent than the applicable standard. EABV1iACTUAL is calculated according to paragraph (h)(7)(ii) of this section.
- EBFEPV2_{iBASE} = Emissions from each Group 2 batch front-end process vent i at the baseline date. EBFEPV2_{iBASE} is calculated according to paragraph (h)(6)(iv) of this section.
- EBFEPV2_{iACTUAL} = Emissions from each Group 2 batch front-end process vent i that is controlled. $EBFEPV2_{iACTUAL}$ is calculated according to paragraph (h)(6)(iii) of this section.
- EABV2_{iBASE} = Emissions from each Group 2 aggregate batch vent stream i at the baseline date. EABV2;BASE is calculated according to paragraph (h)(7)(iv) of this section.
- EABV2_{iACTUAL} = Emissions from each Group 2 aggregate batch vent stream i that is controlled. EABV2_{iACTUAL} is calculated according to paragraph (h)(7)(iii) of this section.
- n = Number of Group 1 emission points included in the emissions average. The value of n is not necessarily the same for continuous front-end process vents, batch frontend process vents, aggregate batch vent streams, storage vessels, wastewater streams, or the collection of process sections within the affected source.
- m = Number of Group 2 emission points included in the emissions average. The value of m is not necessarily the same for continuous front-end process vents, batch frontend process vents, aggregate batch vent streams, storage vessels, wastewater

§ 63.503

streams, or the collection of process sections within the affected source.

- (i) Except as specified in paragraph (h)(1)(iv) of this section, for an emission point controlled using a reference control technology, the percent reduction for calculating credits shall be no greater than the nominal efficiency associated with the reference control technology, unless a higher nominal efficiency is assigned as specified in paragraph (h)(1)(ii) of this section.
- (ii) For an emission point controlled to a level more stringent than the reference control technology, the nominal efficiency for calculating credits shall be assigned as described in paragraph (i) of this section. A reference control technology may be approved for use in a different manner and assigned a higher nominal efficiency according to the procedures in paragraph (i) of this section. A reference control technology may be approved for use in a different manner and assigned a higher nominal efficiency according to the procedure in paragraph (i) of this section.
- (iii) For an emission point controlled using a pollution prevention measure, except for back-end process operation emissions, the nominal efficiency for calculating credits shall be as determined as described in paragraph (j) of

this section. Emissions for back-end process operations shall be determined as described in paragraph (h)(4) of this section.

- (iv) For Group 1 and Group 2 batch front-end process vents and Group 1 and Group 2 aggregate batch vent streams, the percent reduction for calculating credits shall be the percent reduction determined according to the procedures in paragraphs (h)(6)(ii) and (h)(6)(iii) of this section for batch front-end process vents and paragraphs (h)(7)(ii) and (h)(7)(iii) of this section for aggregate batch vent streams.
- (2) Emissions from continuous frontend process vents shall be determined as follows:
- (i) Uncontrolled emissions from Group 1 continuous front-end process vents, ECFEPV1_{iu}, shall be calculated according to the procedures and equation for ECFEPViu in paragraphs (g)(2)(i) and (g)(2)(ii) of this section.
- (ii) Actual emissions from Group 1 continuous front-end process vents controlled using a technology with an approved nominal efficiency greater than 98 percent or a pollution prevention measure achieving greater than 98 reduction. percent emission ECFEPV1_{iACTUAL}, shall be calculated using Equation 42.

$$ECFEPV1_{iACTUAL} = ECFEPV1_{iu} \left(1 - \frac{Nominal efficiency \%}{100\%} \right)$$
 [Eq. 42]

Where:

 $ECFEPV1_{iACTUAL}$ = Emissions for each Group 1 continuous front-end process vent i that is controlled to a level more stringent than the reference control technology.

ECFEPV1_{iu} = Emissions from each Group 1 continuous front-end process vent i if the reference control technology had been applied to the uncontrolled emissions.

(iii) The following procedures shall be used to calculate actual emissions

from Group 2 continuous front-end process vents, ECFEPV2_{iACTUAL}:

(A) For a Group 2 continuous frontend process vent controlled by a control device, a recovery device applied as a pollution prevention project, or a pollution prevention measure, where the control achieves a percent reduction less than or equal to 98 percent reduction, Equation 43 shall be used.

$$ECFEPV2_{iACTUAL} = ECFEPV2_{iu} \left(1 - \frac{Percent\ reduction}{100\%} \right) \qquad [Eq.\ 43]$$

Where:

 $\begin{array}{c} ECFEPV2_{iACTUAL} {=} Emissions \ from \ each \ Group \\ 2 \ continuous \end{array}$

front-end process vent i that is controlled. ECFEPV2_{iu}=Emissions from each Group 2 continuous front-end process vent i if the reference control technology had been applied to the uncontrolled emissions.

- (1) ECFEPV2 $_{iu}$ shall be calculated according to the equations and procedures for ECFEPV $_{iu}$ in paragraphs (g)(2)(i) and (g)(2)(ii) of this section, except as provided in paragraph (h)(2)(iii)(A)(3) of this section.
- (2) The percent reduction shall be calculated according to the procedures in paragraphs (g)(2)(iii)(B)(1) through (g)(2)(iii)(B)(3) of this section, except as provided in paragraph (h)(2)(iii)(A)(4) of this section.
- (3) If a recovery device was added as part of a pollution prevention project,

40 CFR Ch. I (7-1-12 Edition)

ECFEPV2 $_{iu}$ shall be calculated prior to that recovery device. The equation for ECFEPV $_{iu}$ in paragraph (g)(2)(ii) of this section shall be used to calculate ECFEPV2 $_{iu}$; however, the sampling site for measurement of vent stream flow rate and organic HAP concentration shall be at the inlet of the recovery device.

- (4) If a recovery device was added as part of a pollution prevention project, the percent reduction shall be demonstrated by conducting a performance test at the inlet and outlet of that recovery device.
- (B) For a Group 2 continuous frontend process vent controlled using a technology with an approved nominal efficiency greater than 98 percent or a pollution prevention measure achieving greater than 98 percent reduction, Equation 44 shall be used.

$$ECFEPV2_{iACTUAL} = ECFEPV2_{iu} \left(1 - \frac{Nominal efficiency \%}{100\%} \right)$$
 [Eq. 44]

Where:

 $\begin{array}{l} {\rm ECFEPV2_{i_{\rm ACTUAL}}}{\rm =}{\rm Emissions} \ {\rm from} \ {\rm each} \ {\rm Group} \\ {\rm 2 \ continuous} \end{array}$

front-end process vent i that is controlled. $ECFEPV2_{iu}=Emissions$ from each Group 2 continuous front-end process vent i if the reference control technology had been applied to the uncontrolled emissions.

(iv) Emissions from Group 2 continuous front-end process vents at baseline, ECFEPV2 $_{\rm iBASE}$, shall be calculated as follows:

(A) If the continuous front-end process vent was uncontrolled on November 15, 1990, ECFEPV2 $_{iBASE}$ =ECFEPV2 $_{iu}$ and shall be calculated according to the procedures and equation for ECFEPV $_{iu}$ in paragraphs (g)(2)(i) and (g)(2)(ii) of this section.

(B) If the continuous front-end process vent was controlled on November 15, 1990, Equation 45 shall be used.

$$ECFEPV2_{iBASE} = ECFEPV2_{iu} \left(1 - \frac{Percent\ reduction}{100\%} \right)$$
 [Eq. 45]

(1) ECFEPV2 $_{iu}$ is calculated according to the procedures and equation for ECFEPV $_{iu}$ in paragraphs (g)(2)(i) and (g)(2)(ii) of this section.

(2) The percent reduction shall be calculated according to the procedures specified in paragraphs (g)(2)(iii)(B)(1) through (g)(2)(iii)(B)(3) of this section.

(C) If a recovery device was added as part of a pollution prevention project initiated after November 15, 1990, ECFEPV2 $_{iBASE}$ =ECFEPV2 $_{iu}$, where ECFEPV2 $_{iu}$ is calculated according to paragraph (h)(2)(iii)(A)(3) of this section.

§ 63.503

- (3) Emissions from storage vessels shall be calculated using the procedures specified in §63.150(h)(3).
- (4) Emissions from back-end process operations shall be calculated as follows:
- (i) EBEP_{ACTUAL} shall be calculated according to the equation for $EBEP_{ACTUAL}$ contained in paragraph (g)(4)(i) of this section.
- (ii) EBEPc shall be calculated according to the equation for EBEPc contained in paragraph (g)(4)(ii) of this section.
- (5)Emissions from wastewater streams shall be calculated using the procedures specified in §63.150(h)(5).

- (6) Emissions from batch front-end process vents shall be determined as follows:
- (i) Uncontrolled emissions from Group 1 batch front-end process vents (EBFEPV1iu) shall be calculated according using the procedures specified in §63.488(b).
- (ii) Actual emissions from Group 1 batch front-end process vents controlled to a level more stringent than the standard (EBFEPV1_{iACTUAL}) shall be calculated using Equation 46, where percent reduction is for the batch

$$EBFEPV1_{iACTUAL} = EBFEPV1_{iu} \left(1 - \frac{Percent\ reduction}{100\%} \right)$$
 [Eq. 46]

- (A) The percent reduction for the batch cycle shall be calculated according to the procedures in 63.490(c)(2).
- (B) The percent reduction for control devices shall be determined according to the procedures in $\S63.490(c)(2)(i)$ through (c)(2)(iii).
- (C) The percent reduction of pollution prevention measures shall be cal-

culated using the procedures specified in paragraph (j) of this section.

(iii) Actual emissions from Group 2 batch front-end process vents $(EBFEPV2_{iACTUAL})\ shall\ be\ calculated$ using Equation 47 and the procedures in paragraphs (h)(6)(ii)(A) through (h)(6)(ii)(C) of this section. EBFEPV2_{iu} shall be calculated using the procedures specified in §63.488(b).

$$EBFEPV2_{iACTUAL} = EBFEPV2_{iu} \times \left(1 - \frac{Percent\ reduction}{100\%}\right) \qquad [Eq.\ 47]$$

- (iv) Emissions from Group 2 batch front-end process vents at baseline shall be calculated as follows:
- (A) If the batch front-end process vent was uncontrolled on November 15, 1990, EBFEPV2_{iBASE}=EBFEPV2_{iu} and shall be calculated according to the procedures using the procedures specified in §63.488(b).
- (B) If the batch front-end process vent was controlled on November 15, 1990, use Equation 48 and the procedures in paragraphs (h)(6)(ii)(A) through (h)(6)(ii)(C) of this section. EBFEPV2iu shall be calculated using the procedures specified in §63.488(b).

$$EBFEPV2_{iBASE} = EBFEPV2_{iu} \left(1 - \frac{Percent \ reduction}{100\%} \right) \qquad [Eq. 48]$$

40 CFR Ch. I (7-1-12 Edition)

- (7) Emissions from aggregate batch vent streams shall be determined as follows:
- (i) Uncontrolled emissions from Group 1 aggregate batch vent streams (EABV1_{iu}) shall be calculated according to the procedures and equation for

 $EABV_{iu}$ in paragraphs (g)(7)(i) and (g)(7)(ii) of this section.

(ii) Actual emissions from Group 1 aggregate batch vent streams controlled to a level more stringent than the standard (EABV1iACTUAL) shall be calculated using Equation 49:

$$EABV1_{iACTUAL} = EABV1_{iu} \left(1 - \frac{Percent\ reduction}{100\%} \right)$$
 [Eq. 49]

- (A) The percent reduction for control devices shall be determined according to the procedures in §63.490(e).
- (B) The percent reduction of pollution prevention measures shall be calculated using the procedures specified in paragraph (j) of this section.
- (iii) Actual emissions from Group 2 aggregate batch vents streams

 $(EABV2_{iACTUAL})$ shall be calculated using Equation 50 and the procedures in paragraphs (h)(7)(ii)(A) through (h)(7)(ii)(B) of this section. EABV2_{iu} shall be calculated according to the equations and procedures for EABViu in paragraphs (g)(7)(i) and (g)(7)(ii) of this section.

$$EABV2_{iACTUAL} = EABV2_{iu} \left(1 - \frac{Percent\ reduction}{100\%} \right)$$
 [Eq. 50]

- (iv) Emissions from Group 2 aggregate batch vent streams at baseline shall be calculated as follows:
- (A) If the aggregate batch vent stream was uncontrolled on November 15, 1990, EABV2 $_{iBASE}$ =EABV2 $_{iu}$ and shall be calculated according to the procedures and equation for EABViu in paragraph (g)(7)(i) and (g)(7)(ii) of this sec-

(B) If the aggregate batch vent stream was controlled on November 15, 1990, use Equation 51 and the procedures in paragraphs (h)(7)(ii)(A) through (h)(7)(ii)(B) of this section. $\mathrm{EABV2}_{iu}$ shall be calculated according to the equations and procedures for $EABV_{iu}$ in paragraphs (g)(7)(i) and (g)(7)(ii) of this section.

$$EABV2_{iBASE} = EABV2_{iu} \left(1 - \frac{Percent\ reduction}{100\%} \right)$$
 [Eq. 51]

(i) The following procedures shall be followed to establish nominal efficiencies for emission controls for storage vessels, continuous front-end process vents, and process wastewater streams. The procedures in paragraphs (i)(1) through (i)(6) of this section shall be followed for control technologies that are different in use or design from the reference control technologies and

achieve greater percent reductions than the percent efficiencies assigned to the reference control technologies in § 63.111.

(1) In those cases where the owner or operator is seeking permission to take credit for use of a control technology that is different in use or design from the reference control technology, and the different control technology will be

§ 63.503

used in more than three applications at a single plant-site, the owner or operator shall submit the information specified in paragraphs (i)(1)(i) through (i)(1)(iv) of this section, as specified in §63.506(e)(7)(ii), to the Director of the EPA Office of Air Quality Planning and Standards, in writing.

- (i) Emission stream characteristics of each emission point to which the control technology is or will be applied, including the kind of emission point, flow, organic HAP concentration, and all other stream characteristics necessary to design the control technology or determine its performance.
- (ii) Description of the control technology, including design specifications.
- (iii) Documentation demonstrating to the Administrator's satisfaction the control efficiency of the control technology. This may include performance test data collected using an appropriate EPA Method or any other method validated according to Method 301, 40 CFR part 63, appendix A. If it is infeasible to obtain test data, documentation may include a design evaluation and calculations. The engineering basis of the calculation procedures and all inputs and assumptions made in the calculations shall be documented.
- (iv) A description of the parameter or parameters to be monitored to ensure that the control technology will be operated in conformance with its design and an explanation of the criteria used for selection of that parameter (or parameters).
- (2) The Administrator shall determine within 120 days whether an application presents sufficient information to determine nominal efficiency. The Administrator reserves the right to request specific data in addition to the items listed in paragraph (i)(1) of this section.
- (3) The Administrator shall determine within 120 days of the submittal of sufficient data whether a control technology shall have a nominal efficiency and the level of that nominal efficiency. If, in the Administrator's judgment, the control technology achieves a level of emission reduction greater than the reference control technology for a particular kind of emission point, the Administrator will

publish a FEDERAL REGISTER notice establishing a nominal efficiency for the control technology.

- (4) The Administrator may grant permission to take emission credits for use of the control technology. The Administrator may also impose requirements that may be necessary to ensure operation and maintenance to achieve the specified nominal efficiency.
- (5) In those cases where the owner or operator is seeking permission to take credit for use of a control technology that is different in use or design from the reference control technology and the different control technology will be used in no more than three applications at a single plant site, the owner or operator shall submit the information listed in paragraph (i)(1)(i) as specified in §63.506(e)(7)(ii) to the Administrator
- (i) In these instances, use and conditions for use of the control technology may be approved by the permitting authority as part of an operating permit application or modification. The permitting authority shall follow the procedures specified in paragraphs (i)(2) through (i)(4) of this section except that, in these instances, a FEDERAL REGISTER notice is not required to establish the nominal efficiency for the different technology.
- (ii) If, in reviewing the application, the permitting authority believes the control technology has broad applicability for use by other sources, the permitting authority shall submit the information provided in the application to the Director of the EPA Office of Air Quality Planning and Standards. The Administrator shall review the technology for broad applicability and may publish a FEDERAL REGISTER notice; however, this review shall not affect the permitting authority's approval of the nominal efficiency of the control technology for the specific application.
- (6) If, in reviewing an application for a control technology for an emission point, the Administrator or permitting authority determines that the control technology is not different in use or design from the reference control technology, the Administrator or permitting authority shall deny the application.

40 CFR Ch. I (7-1-12 Edition)

- (j) The following procedures shall be used for calculating the efficiency (percent reduction) of pollution prevention measures for storage vessels, continuous front-end process vents, batch front-end process vents, aggregate batch vent streams, and wastewater streams:
- (1) A pollution prevention measure is any practice which meets the criteria of paragraphs (j)(1)(i) and (j)(1)(ii) of this section.
- (i) A pollution prevention measure is any practice that results in a lesser quantity of organic HAP emissions per unit of product released to the atmosphere prior to out-of-process recycling, treatment, or control of emissions, while the same product is produced.
- (ii) Pollution prevention measures may include substitution of feedstocks that reduce organic HAP emissions; alterations to the production process to reduce the volume of materials released to the environment; equipment modifications; housekeeping measures; and in-process recycling that returns waste materials directly to production

- as raw materials. Production cutbacks do not qualify as pollution prevention.
- (2) The emission reduction efficiency of pollution prevention measures implemented after November 15, 1990, may be used in calculating the actual emissions from an emission point in the debit and credit equations in paragraphs (g) and (h) of this section.
- (i) For pollution prevention measures, the percent reduction is used in the equations in paragraphs (g)(2)through (g)(7) of this section and paragraphs (h)(2) through (h)(7) of this section is the percent difference between the monthly organic HAP emissions for each emission point after the pollution prevention measure for the most recent month versus monthly emissions from the same emission point before the pollution prevention measure, adjusted by the volume of product produced during the two monthly periods.
- (ii) Equation 52 shall be used to calculate the percent reduction of a pollution prevention measure for each emission point.

$$Percent reduction = E_B - \frac{\left(E_{pp} \times P_B\right)}{E_B} \times 100 \qquad [Eq. 52]$$

where:

Percent reduction=Efficiency of pollution prevention measure (percent organic HAP reduction).

E_B=Monthly emissions before the pollution prevention measure, Mg/month, determined as specified paragraphs in (j)(2)(ii)(A), (j)(2)(ii)(B), and (j)(2)(ii)(C) of this section.

 E_{pp} =Monthly emissions after the pollution prevention measure, Mg/month, as determined for the most recent month, determined as specified in either paragraphs (i)(2)(ii)(D) or (i)(2)(ii)(E) of this section.

P_B=Monthly production before the pollution prevention measure, Mg/month, during the same period over which E_{B} is calculated.

Ppp=Monthly production after the pollution prevention measure, Mg/month, as determined for the most recent month.

(A) The monthly emissions before the pollution prevention measure, E_B, shall be determined in a manner consistent with the equations and procedures in paragraph (g)(2) of this section for continuous front-end process vents, paragraph (g)(3) of this section for storage vessels, paragraph (g)(6) of this section for batch front-end process vents, and paragraph (g)(7) of this section for aggregate batch vent streams.

(B) For wastewater, E_{B} shall be calculated according to §63.150(j)(2)(ii)(B).

(C) If the pollution prevention measure was implemented prior to September 5, 1996, records may be used to determine E_B .

(D) The monthly emissions after the pollution prevention measure, Epp, may be determined during a performance test or by a design evaluation and documented engineering calculations. Once an emissions-to-production ratio

§ 63.503

has been established, the ratio may be used to estimate monthly emissions from monthly production records.

- (E) For wastewater, E_{pp} shall be calculated according to 63.150(j)(2)(ii)(E).
- (iii) All equations, calculations, test procedures, test results, and other information used to determine the percent reduction achieved by a pollution prevention measure for each emission point shall be fully documented.
- (iv) The same pollution prevention measure may reduce emissions from multiple emission points. In such cases, the percent reduction in emissions for each emission point shall be calculated.
- (v) For the purposes of the equations in paragraphs (h)(2) through (h)(7) of this section, used to calculate credits for emission points controlled more stringently than the reference control technology or standard, the nominal efficiency of a pollution prevention measure is equivalent to the percent reduction of the pollution prevention measure. When a pollution prevention measure is used, the owner or operator of an affected source is not required to apply to the Administrator for a nominal efficiency and is not subject to paragraph (i) of this section.
- (k) The owner or operator shall demonstrate that the emissions from the emission points proposed to be included in the emissions average will not result in greater hazard, or at the option of the Administrator, greater risk to human health or the environment than if the emission points were controlled according to the provisions in §§ 63.484, 63.485, 63.486, 63.493, and 63.501.
- (1) This demonstration of hazard or risk equivalency shall be made to the satisfaction of the Administrator.
- (i) The Administrator may require owners and operators to use specific methodologies and procedures for making a hazard or risk determination.
- (ii) The demonstration and approval of hazard or risk equivalency shall be made according to any guidance that the Administrator makes available for use.
- (2) Owners and operators shall provide documentation demonstrating the hazard or risk equivalency of their proposed emissions average in their operating permit application or in their

Emissions Averaging Plan if an operating permit application has not yet been submitted.

- (3) An Emissions Averaging Plan that does not demonstrate hazard or risk equivalency to the satisfaction of the Administrator shall not be approved. The Administrator may require such adjustments to the Emissions Averaging Plan as are necessary in order to ensure that the emissions average will not result in greater hazard or risk to human health or the environment than would result if the emission points were controlled according to §§63.484, 63.485, 63.486, 63.493, and 63.501.
- (4) A hazard or risk equivalency demonstration shall:
- (i) Be a quantitative, bona fide chemical hazard or risk assessment;
- (ii) Account for differences in chemical hazard or risk to human health or the environment; and
- (iii) Meet any requirements set by the Administrator for such demonstrations
- (1) For periods of monitoring excursions, an owner or operator may request that the provisions of paragraphs (1)(1) through (1)(4) of this section be followed instead of the procedures in paragraphs (f)(2)(i) and (f)(2)(ii) of this section.
- (1) The owner or operator shall notify the Administrator of monitoring excursions in the Periodic Reports as required in §63.506(e)(6).
- (2) The owner or operator shall demonstrate that other types of monitoring data or engineering calculations are appropriate to establish that the control device for the emission point was operating in such a fashion to warrant assigning full or partial credits and debits. This demonstration shall be made to the Administrator's satisfaction, and the Administrator may establish procedures for demonstrating compliance that are acceptable.
- (3) The owner or operator shall provide documentation of the excursion and the other types of monitoring data or engineering calculations to be used to demonstrate that the control device for the emission point was operating in such a fashion to warrant assigning full or partial credits and debits.

- (4) The Administrator may assign full or partial credit and debits upon review of the information provided.
- (m) For each emission point included in an emissions average, the owner or operator shall perform testing, monitoring, recordkeeping, and reporting equivalent to that required for Group 1 emission points complying with §§ 63.484, 63.485, 63.486, 63.493, and 63.501, as applicable. If back-end process operations are included in an emissions average, the owner or operator shall pertesting, monitoring, recordkeeping, and reporting equivalent to that required for back-end process operations complying with §63.493. The specific requirements for continuous front-end process vents, batch frontend process vents, aggregate batch vent streams, storage vessels, back-end process operations, and wastewater are identified in paragraphs (m)(1) through (m)(6) of this section.
- (1) For each continuous front-end process vent equipped with a flare, incinerator, boiler, or process heater, as appropriate to the control technique:
- (i) Determine whether the continuous front-end process vent is Group 1 or Group 2 according to the procedures specified in §63.115 and as required by § 63.485;
- (ii) Conduct initial performance tests to determine percent reduction as specified in §63.116 and as required by §63.485; and
- (iii) Monitor the operating parameters, keep records, and submit reports as specified in §§63.114, 63.117(a), and 63.118(a) and (f), as required, for the specific control device as required by 8 63, 485.
- (2) For each continuous front-end process vent equipped with a carbon adsorber, absorber, or condenser but not equipped with a control device, as appropriate to the control technique:
- (i) Determine the flow rate, organic HAP concentration, and TRE index value according to the procedures specified in §63.115; and
- (ii) Monitor the operating parameters, keep records, and submit reports according to the procedures specified in §§ 63.114, 63.117(a), and 63.118(b) and (f), as required, for the specific recovery device, and as required by §63.485.

40 CFR Ch. I (7-1-12 Edition)

- (3) For each storage vessel controlled with an internal floating roof, external roof, or a closed vent system with a control device, as appropriate to the control technique:
- (i) Perform the monitoring or inspection procedures according to the procedures specified in §63.120, and as required by §63.484;
- (ii) Perform the reporting and recordkeeping procedures according to the procedures specified in §§ 63.122 and 63.123, and as required by § 63.484; and
- (iii) For closed vent systems with control devices, conduct an initial design evaluation and submit an operating plan according to the procedures specified in §§63.120(d) and 63.122(b), and as required by §63.484.
- (4) For back-end process operations included in an emissions average:
- (i) If stripping technology, and no control or recovery device, is used to reduce back-end process operation emissions, the owner or operator shall implement the following portions of this subpart:
- (A) Paragraphs (b)(1), (b)(2), and (b)(3) of §63.495, paragraph (b) of §63.498, and the applicable provisions of §63.499, or
- (B) Paragraphs (c) (1), (2), and (3) of §63.495, paragraph (c) of §63.498, and the applicable provisions of §63.499;
- (ii) If a control or recovery device is used to reduce back-end process operation emissions, the owner or operator of the affected source shall comply with §§ 63.496, 63.497, 63.498(d), and the applicable provisions of 63.499, and shall implement the provisions of these
- (5) For wastewater emission points. as appropriate to the control techniques:
- (i) For wastewater treatment processes, conduct tests according to the procedures specified in §63.138(i) and (j), and as required by §63.501;
- (ii) Conduct inspections and monitoring according to the procedures specified in §63.143, and as required by § 63.501;
- (iii) Implement a recordkeeping program according to the procedures specified in §63.147, and as required by §63.501; and
- (iv) Implement a reporting program according to the procedures specified in §63.146, and as required by §63.501.

§ 63.504

- (6) For each batch front-end process vent and aggregate batch vent stream equipped with a control device, as appropriate to the control technique:
- (i) Determine whether the batch front-end process vent or aggregate batch vent stream is Group 1 or Group 2 according to the procedures specified in §63.488:
- (ii) Conduct performance tests according to the procedures specified in § 63.490;
- (iii) Conduct monitoring according to the procedures specified in §63.489; and
- (iv) Perform the recordkeeping and reporting procedures according to the procedures specified in §§ 63.491 and
- (7) If an emission point in an emissions average is controlled using a pollution prevention measure or a device or technique for which no monitoring parameters or inspection procedures are required by §§63.484, 63.485, 63.486, 63.493, or §63.501, the owner or operator shall submit the information specified in §63.506(f) for alternate monitoring parameters or inspection procedures in the Emissions Averaging Plan or operating permit application.
- (n) Records of all information required to calculate emission debits and credits shall be retained for 5 years.
- (o) Precompliance Reports, Emission Averaging Plans, Notifications of Compliance Status, Periodic Reports, and other reports shall be submitted as required by §63.506.

[62 FR 46925, Sept. 5, 1996, as amended at 64 FR 11543, Mar. 9, 1999; 65 FR 38071, June 19, 2000: 76 FR 22592, Apr. 21, 20111

§63.504 Additional requirements for performance testing.

- (a) Performance testing shall be conducted in accordance with §63.7(a)(1), (a)(3), (d), (e)(1), (e)(2), (e)(4), (g), and(h), with the exceptions specified in paragraphs (a)(1) through (a)(5) of this section and the additions specified in paragraph (b) of this section. Sections 63.484 through 63.501 also contain specific testing requirements.
- (1) Performance tests shall be conducted at maximum representative operating conditions achievable during one of the time periods described in paragraph (a)(1)(i) of this section, without causing any of the situations de-

scribed in paragraph (a)(1)(ii) of this section to occur. Upon request, the owner or operator shall make available to the Administrator such records as may be necessary to determine the conditions of performance tests.

- (i) The 6-month period that ends 2 months before the Notification of Compliance Status is due, according to $\S63.506(e)(5)$; or the 6-month period that begins 3 months before the performance test and ends 3 months after the performance test.
- (ii) Causing damage to equipment; necessitating that the owner or operator make product that does not meet an existing specification for sale to a customer; or necessitating that the owner or operator make product in excess of demand.
- (2) References in §63.7(g) to the Notification of Compliance Status requirements in \$63.9(h) shall refer to the requirements in $\S63.506(e)(5)$.
- (3) Because the site-specific test plans in §63.7(c)(3) are not required, $\S63.7(h)(4)(ii)$ is not applicable.
- (4) The owner or operator shall notify the Administrator of the intent to conduct a performance test at least 30 days before the performance test is scheduled, to allow the Administrator the opportunity to have an observer present during the test. If after 30 days notice for an initially scheduled performance test, there is a delay (due to operational problems, etc.) in conducting the scheduled performance test, the owner or operator of an affected facility shall notify the Administrator as soon as possible of any delay in the original test date, either by providing at least 7 days prior notice of the rescheduled date of the performance test, or by arranging a rescheduled date with the Administrator by mutual agreement.
- (5) Performance tests shall be performed no later than 150 days after the compliance dates specified in this subpart (i.e., in time for the results to be included in the Notification of Compliance Status), rather than according to the time periods in $\S63.7(a)(2)$.
- (b) Data shall be reduced in accordance with the EPA approved methods specified in the applicable subpart or, if other test methods are used, the data



40 CFR Ch. I (7-1-12 Edition)

§63.650 Gasoline loading rack provisions.

- (a) Except as provided in paragraphs (b) through (c) of this section, each owner or operator of a Group 1 gasoline loading rack classified under Standard Industrial Classification code 2911 located within a contiguous area and under common control with a petroleum refinery shall comply with subpart R, §§63.421, 63.422(a) through (c) and (e), 63.425(a) through (c) and (i), 63.425(e) through (h), 63.427(a) and (b), and 63.428(b), (c), (g)(1), (h)(1) through (3), and (k).
- (b) As used in this section, all terms not defined in §63.641 shall have the meaning given them in subpart A or in 40 CFR part 63, subpart R. The §63.641 definition of "affected source" applies under this section.
- (c) Gasoline loading racks regulated under this subpart are subject to the compliance dates specified in §63.640(h).

[60 FR 43260, Aug. 18, 1995, as amended at 61 FR 29880, June 12, 1996; 74 FR 55685, Oct. 28, 2009]

§ 63.651 Marine tank vessel loading operation provisions.

- (a) Except as provided in paragraphs (b) through (d) of this section, each owner or operator of a marine tank vessel loading operation located at a petroleum refinery shall comply with the requirements of §§63.560 through 63.568
- (b) As used in this section, all terms not defined in §63.641 shall have the meaning given them in subpart A or in 40 CFR part 63, subpart Y. The §63.641 definition of "affected source" applies under this section.
- (c) The notification reports under $\S63.567(b)$ are not required.
- (d) The compliance time of 4 years after promulgation of 40 CFR part 63, subpart Y does not apply. The compliance time is specified in §63.640(h)(3).

[60 FR 43260, Aug. 18, 1995, as amended at 61 FR 29880, June 12, 1996; 74 FR 55685, Oct. 28, 2009]

§ 63.652 Emissions averaging provisions.

(a) This section applies to owners or operators of existing sources who seek

- to comply with the emission standard in §63.642(g) by using emissions averaging according to §63.642(1) rather than following the provisions of §§63.643 through 63.647, and §§63.650 and 63.651. Existing marine tank vessel loading operations located at the Valdez Marine Terminal source may not comply with the standard by using emissions averaging.
- (b) The owner or operator shall develop and submit for approval an Implementation Plan containing all of the information required in §63.653(d) for all points to be included in an emissions average. The Implementation Plan shall identify all emission points to be included in the emissions average. This must include any Group 1 emission points to which the reference control technology (defined in §63.641) is not applied and all other emission points being controlled as part of the average
- (c) The following emission points can be used to generate emissions averaging credits if control was applied after November 15, 1990 and if sufficient information is available to determine the appropriate value of credits for the emission point:
 - (1) Group 2 emission points;
- (2) Group 1 storage vessels, Group 1 wastewater streams, Group 1 gasoline loading racks, Group 1 marine tank vessels, and Group 1 miscellaneous process vents that are controlled by a technology that the Administrator or permitting authority agrees has a higher nominal efficiency than the reference control technology. Information on the nominal efficiencies for such technologies must be submitted and approved as provided in paragraph (i) of this section; and
- (3) Emission points from which emissions are reduced by pollution prevention measures. Percentages of reduction for pollution prevention measures shall be determined as specified in paragraph (j) of this section.
- (i) For a Group 1 emission point, the pollution prevention measure must reduce emissions more than the reference control technology would have had the reference control technology been applied to the emission point instead of

§ 63.652

the pollution prevention measure except as provided in paragraph (c)(3)(ii) of this section.

- (ii) If a pollution prevention measure is used in conjunction with other controls for a Group 1 emission point, the pollution prevention measure alone does not have to reduce emissions more than the reference control technology, but the combination of the pollution prevention measure and other controls must reduce emissions more than the reference control technology would have had it been applied instead.
- (d) The following emission points cannot be used to generate emissions averaging credits:
- (1) Emission points already controlled on or before November 15, 1990 unless the level of control is increased after November 15, 1990, in which case credit will be allowed only for the increase in control after November 15, 1990:
- (2) Group 1 emission points that are controlled by a reference control technology unless the reference control technology has been approved for use in a different manner and a higher nominal efficiency has been assigned according to the procedures in paragraph (i) of this section. For example, it is not allowable to claim that an internal floating roof meeting only the specifications stated in the reference control technology definition in §63.641 (i.e., that meets the specifications of §63.119(b) of subpart G but does not have controlled fittings per §63.119 (b)(5) and (b)(6) of subpart G) applied to a storage vessel is achieving greater than 95 percent control;
- (3) Emission points on shutdown process units. Process units that are shut down cannot be used to generate credits or debits;
- (4) Wastewater that is not process wastewater or wastewater streams treated in biological treatment units. These two types of wastewater cannot be used to generate credits or debits. Group 1 wastewater streams cannot be left undercontrolled or uncontrolled to generate debits. For the purposes of this section, the terms "wastewater" and "wastewater stream" are used to mean process wastewater; and
- (5) Emission points controlled to comply with a State or Federal rule

other than this subpart, unless the level of control has been increased after November 15, 1990 above what is required by the other State or Federal rule. Only the control above what is required by the other State or Federal rule will be credited. However, if an emission point has been used to generate emissions averaging credit in an approved emissions average, and the point is subsequently made subject to a State or Federal rule other than this subpart, the point can continue to generate emissions averaging credit for the purpose of complying with the previously approved average.

- (e) For all points included in an emissions average, the owner or operator shall:
- (1) Calculate and record monthly debits for all Group 1 emission points that are controlled to a level less stringent than the reference control technology for those emission points. Equations in paragraph (g) of this section shall be used to calculate debits.
- (2) Calculate and record monthly credits for all Group 1 or Group 2 emission points that are overcontrolled to compensate for the debits. Equations in paragraph (h) of this section shall be used to calculate credits. Emission points and controls that meet the criteria of paragraph (c) of this section may be included in the credit calculation, whereas those described in paragraph (d) of this section shall not be included.
- (3) Demonstrate that annual credits calculated according to paragraph (h) of this section are greater than or equal to debits calculated for the same annual compliance period according to paragraph (g) of this section.
- (i) The initial demonstration in the Implementation Plan that credit-generating emission points will be capable of generating sufficient credits to offset the debits from the debit-generating emission points must be made under representative operating conditions.
- (ii) After the compliance date, actual operating data will be used for all debit and credit calculations.
- (4) Demonstrate that debits calculated for a quarterly (3-month) period according to paragraph (g) of this section are not more than 1.30 times

the credits for the same period calculated according to paragraph (h) of this section. Compliance for the quarter shall be determined based on the ratio of credits and debits from that quarter, with 30 percent more debits than credits allowed on a quarterly basis.

- (5) Record and report quarterly and annual credits and debits in the Perias specified odic Reports §63.655(g)(8). Every fourth Periodic Report shall include a certification of compliance with the emissions averaging provisions as required §63.655(g)(8)(iii).
- (f) Debits and credits shall be calculated in accordance with the methods and procedures specified in paragraphs (g) and (h) of this section, respectively, and shall not include emissions from the following:
- (1) More than 20 individual emission points. Where pollution prevention measures (as specified in paragraph (j)(1) of this section) are used to control emission points to be included in an emissions average, no more than 25 emission points may be included in the average. For example, if two emission points to be included in an emissions average are controlled by pollution prevention measures, the average may include up to 22 emission points.

40 CFR Ch. I (7-1-12 Edition)

- (2) Periods of startup, shutdown, and malfunction as described in the source's startup, shutdown, and malfunction plan required by §63.6(e)(3) of subpart A of this part.
- (3) For emission points for which continuous monitors are used, periods of excess emissions as defined in $\S63.655(g)(6)(i)$. For these periods, the calculation of monthly credits and debits shall be adjusted as specified in paragraphs (f)(3)(i) through (f)(3)(iii) of this section.
- (i) No credits would be assigned to the credit-generating emission point.
- (ii) Maximum debits would be assigned to the debit-generating emission
- (iii) The owner or operator may use the procedures in paragraph (1) of this section to demonstrate to the Administrator that full or partial credits or debits should be assigned.
- (g) Debits are generated by the difference between the actual emissions from a Group 1 emission point that is uncontrolled or is controlled to a level less stringent than the reference control technology, and the emissions allowed for Group 1 emission point. Debits shall be calculated as follows:
- (1) The overall equation for calculating sourcewide debits is:

$$\begin{aligned} \text{Debits} = \sum_{i=1}^{n} & \left(\text{EPV}_{\text{iACTUAL}} - \left(0.02 \right) \text{EPV}_{\text{iu}} \right) + \sum_{i=1}^{n} \left(\text{ES}_{\text{iACTUAL}} - \left(0.05 \right) \text{ES}_{\text{iu}} \right) + \sum_{i=1}^{n} \left(\text{EGLR}_{\text{iACTUAL}} - \text{EGLR}_{\text{iC}} \right) \\ & + \sum_{i=1}^{n} \left(\text{EMV}_{\text{iACTUAL}} - \left(0.03 \right) \text{EMViu} \right) \end{aligned}$$

where:

Debits and all terms of the equation are in units of megagrams per month, and

EPV; ACTUAL = Emissions from each Group 1 miscellaneous process vent i that is uncontrolled or is controlled to a level less stringent than the reference control technology. This is calculated according to paragraph (g)(2) of this section.

(0.02) EPV_{iu}=Emissions from each Group 1 miscellaneous process vent i if the reference control technology had been applied to the uncontrolled emissions, calculated according to paragraph (g)(2) of this sec-

ES_{iACTUAL}=Emissions from each Group 1 storage vessel i that is uncontrolled or is controlled to a level less stringent than the reference control technology. This is calculated according to paragraph (g)(3) of this section.

(0.05) ES_{iu}=Emissions from each Group 1 storage vessel i if the reference control technology had been applied to the uncontrolled emissions, calculated according to paragraph (g)(3) of this section.

EGLR_{iACTUAL}=Emissions from each Group 1 gasoline loading rack i that is uncontrolled or is controlled to a level less stringent than the reference control technology. This is calculated according to paragraph (g)(4) of this section.

EGLR_{ic}=Emissions from each Group 1 gasoline loading rack i if the reference control

§ 63.652

technology had been applied to the uncontrolled emissions. This is calculated according to paragraph (g)(4) of this section.

EMV_{ACTUAL}=Emissions from each Group 1 marine tank vessel i that is uncontrolled or is controlled to a level less stringent than the reference control technology. This is calculated according to paragraph (g)(5) of this section.

(0.03) EMV_{in}=Emissions from each Group 1 marine tank vessel i if the reference control technology had been applied to the uncontrolled emissions calculated according to paragraph (g)(5) of this section.

n=The number of Group 1 emission points being included in the emissions average. The value of n is not necessarily the same for each kind of emission point.

Emissions from miscellaneous process vents shall be calculated as follows:

(i) For purposes of determining miscellaneous process vent stream flow rate, organic HAP concentrations, and temperature, the sampling site shall be after the final product recovery device, if any recovery devices are present; before any control device (for miscellaneous process vents, recovery devices shall not be considered control devices); and before discharge to the atmosphere. Method 1 or 1A of part 60, appendix A shall be used for selection of the sampling site.

(ii) The following equation shall be used for each miscellaneous process vent i to calculate EPV_{iu}:

$$EPV_{iu} = (2.494 \times 10^{-9})Qh \left(\sum_{j=1}^{n} C_{j}M_{j}\right)$$

 $EPV_{iu} \text{=} Uncontrolled process vent emission}$ rate from miscellaneous process vent i, megagrams per month.

Q=Vent stream flow rate, dry standard cubic meters per minute, measured using Methods 2, 2A, 2C, or 2D of part 60 appendix A, as appropriate.

h=Monthly hours of operation during which positive flow is present in the vent, hours per month.

C_j=Concentration, parts per million by volume, dry basis, of organic HAP j as measured by Method 18 of part 60 appendix A.

M=Molecular weight of organic HAP j, gram per gram-mole.

n=Number of organic HAP's in the miscellaneous process vent stream.

(A) The values of Q, C_i, and M_i shall be determined during a performance test conducted under representative operating conditions. The values of Q, C_i, and M_i shall be established in the Notification of Compliance Status report and must be updated as provided in paragraph (g)(2)(ii)(B) of this section.

(B) If there is a change in capacity utilization other than a change in monthly operating hours, or if any other change is made to the process or product recovery equipment or operation such that the previously measured values of Q, C_j , and M_j are no longer representative, a new performance test shall be conducted to determine new representative values of Q, C_i, and M_i. These new values shall be used to calculate debits and credits from the time of the change forward, and the new values shall be reported in the next Periodic Report.

(iii) The following procedures and equations shall be used to calculate EPV_{iACTUAL}:

(A) If the vent is not controlled by a control device or pollution prevention measure, $EPV_{iACTUAL} = EPV_{iu}$, where EPViu is calculated according to the procedures in paragraphs (g)(2)(i) and (g)(2)(ii) of this section.

(B) If the vent is controlled using a control device or a pollution prevention measure achieving less than 98percent reduction,

$$\mathrm{EPV}_{\mathrm{iACTUAL}} = \mathrm{EPV}_{\mathrm{iu}} \times \left(1 - \frac{\mathrm{Percent \, reduction}}{100\%}\right)$$

measured according to the procedures meeting the criteria in §63.116(a) of

(1) The percent reduction shall be control device is used. For a flare in §63.116 of subpart G if a combustion subpart G, or a boiler or process heater

meeting the criteria in §63.645(d) of this subpart or §63.116(b) of subpart G, the percentage of reduction shall be 98 percent. If a noncombustion control device is used, percentage of reduction shall be demonstrated by a performance test at the inlet and outlet of the device, or, if testing is not feasible, by a control design evaluation and documented engineering calculations.

- (2) For determining debits from miscellaneous process vents, product recovery devices shall not be considered control devices and cannot be assigned a percentage of reduction in calculating EPV_{IACTUAL}. The sampling site for measurement of uncontrolled emissions is after the final product recovery device.
- (3) Procedures for calculating the percentage of reduction of pollution prevention measures are specified in paragraph (j) of this section.
- (3) Emissions from storage vessels shall be calculated as specified in §63.150(g)(3) of subpart G.
- (4) Emissions from gasoline loading racks shall be calculated as follows:
- (i) The following equation shall be used for each gasoline loading rack i to calculate $EGLR_{iu}$:

$$EGLR_{iu} = \left(1.20 \times 10^{-7}\right) \frac{SPMG}{T}$$

where:

 $\begin{array}{ll} EGLR_{iu} = Uncontrolled \ transfer \ HAP \ emission \\ rate \ from \ gasoline \ loading \ rack \ i, \\ megagrams \ per \ month \end{array}$

S=Saturation factor, dimensionless (see table 33 of subpart G).

P=Weighted average rack partial pressure of organic HAP's transferred at the rack during the month, kilopascals.

M=Weighted average molecular weight of organic HAP's transferred at the gasoline loading rack during the month, gram per gram-mole.

G=Monthly volume of gasoline transferred from gasoline loading rack, liters per month.

T=Weighted rack bulk liquid loading temperature during the month, degrees kelvin (degrees Celsius °C + 273).

(ii) The following equation shall be used for each gasoline loading rack i to calculate the weighted average rack partial pressure:

40 CFR Ch. I (7-1-12 Edition)

$$P = \frac{\sum_{j=n}^{j=n} (P_j) (G_j)}{G}$$

where:

 P_j =Maximum true vapor pressure of individual organic HAP transferred at the rack, kilopascals.

G=Monthly volume of organic HAP transferred, liters per month, and

$$G = \sum_{1}^{j=n} G_j$$

 G_j =Monthly volume of individual organic HAP transferred at the gasoline loading rack, liters per month.

n=Number of organic HAP's transferred at the gasoline loading rack.

(iii) The following equation shall be used for each gasoline loading rack i to calculate the weighted average rack molecular weight:

$$M = \frac{\sum_{j=n}^{j=n} (M_j)(G_j)}{G}$$

where:

 M_j =Molecular weight of individual organic HAP transferred at the rack, gram per gram-mole.

 $G, \ G_j, \ and \ n \ are as defined in paragraph <math display="inline">(g)(4)(ii)$ of this section.

(iv) The following equation shall be used for each gasoline loading rack i to calculate the monthly weighted rack bulk liquid loading temperature:

$$T = \frac{\displaystyle\sum_{j=n}^{j=n} \left(T_{j}\right) \!\! \left(G_{j}\right)}{G}$$

T_j=Average annual bulk temperature of individual organic HAP loaded at the gasoline loading rack, kelvin (degrees Celsius °C+273).

G, G_j , and n are as defined in paragraph (g)(4)(ii) of this section.

(v) The following equation shall be used to calculate $\mathrm{EGLR}_{\mathrm{ic}}$:

$$EGLR_{ic} = 1 \times 10^{-8} G$$

G is as defined in paragraph (g)(4)(ii) of this section.

§ 63.652

- (vi) The following procedures and equations shall be used to calculate EGLR_{iACTUAL}:
- (A) If the gasoline loading rack is not controlled, EGLR_{iACTUAL}=EGLR_{iu}, where EGLR_{iu} is calculated using the equations specified in paragraphs (g)(4)(i) through (g)(4)(iv) of this section.
- (B) If the gasoline loading rack is controlled using a control device or a pollution prevention measure not achieving the requirement of less than 10 milligrams of TOC per liter of gasoline loaded.

$$EGLR_{iACTUAL} = EGLR_{iu} \left(\frac{1 - Percent reduction}{100\%} \right)$$

- (1) The percent reduction for a control device shall be measured according to the procedures and test methods specified in §63.128(a) of subpart G. If testing is not feasible, the percentage of reduction shall be determined through a design evaluation according to the procedures specified in §63.128(h) of subpart G.
- (2) Procedures for calculating the percentage of reduction for pollution prevention measures are specified in paragraph (j) of this section.
- (5) Emissions from marine tank vessel loading shall be calculated as fol-
- (i) The following equation shall be used for each marine tank vessel i to calculate EMV_{in}:

$$EMV_{iu} = \sum_{i=1}^{m} (Q_i)(F_i)(P_i)$$

where:

- EMV: =Uncontrolled marine tank vessel HAP emission rate from marine tank vessel i, megagrams per month.
- Qi=Quantity of commodity loaded (per vessel type), liters.
- Fi=Emission factor, megagrams per liter.
- P=Percent HAP.
- m=Number of combinations of commodities and vessel types loaded.

Emission factors shall be based on test data or emission estimation procedures specified in §63.565(1) of subpart Υ.

- (ii) The following procedures and equations shall be used to calculate EMV_{iACTUAL}:
- (A) If the marine tank vessel is not controlled, $EMV_{iACTUAL}$ = EMV_{iu} , where EMV_{in} is calculated using the equations specified in paragraph (g)(5)(i) of this
- (B) If the marine tank vessel is controlled using a control device or a pollution prevention measure achieving less than 97-percent reduction,

$$EMV_{iACTUAL} = EMV_{iu} \left(\frac{1 - Percent reduction}{100\%} \right)$$

- (1) The percent reduction for a control device shall be measured according to the procedures and test methods specified in §63.565(d) of subpart Y. If testing is not feasible, the percentage of reduction shall be determined through a design evaluation according to the procedures specified in §63.128(h) of subpart G.
- (2) Procedures for calculating the percentage of reduction for pollution prevention measures are specified in paragraph (j) of this section.
- (h) Credits are generated by the difference between emissions that are allowed for each Group 1 and Group 2 emission point and the actual emissions from a Group 1 or Group 2 emission point that has been controlled

40 CFR Ch. I (7-1-12 Edition)

after November 15, 1990 to a level more stringent than what is required by this subpart or any other State or Federal

rule or statute. Credits shall be calculated as follows:

(1) The overall equation for calculating sourcewide credits is:

$$\begin{split} \text{Credits} &= D \sum_{i=1}^{n} \left(\left(0.02 \right) \text{ EPV1}_{iu} - \text{EPV1}_{iACTUAL} \right) + D \sum_{i=1}^{m} \left(\text{EPV2}_{iBASE} - \text{EPV2}_{iACTUAL} \right) + \\ & D \sum_{i=1}^{n} \left(\left(0.05 \right) \text{ ES1}_{iu} - \text{ES1}_{iACTUAL} \right) + D \sum_{i=1}^{m} \left(\text{ES2}_{iBASE} - \text{ES2}_{iACTUAL} \right) + \\ & D \sum_{i=1}^{n} \left(\text{ EGLR}_{ic} - \text{EGLRI}_{iACTUAL} \right) + D \sum_{i=1}^{m} \left(\text{EGLR2}_{iBASE} - \text{EGLR2}_{iACTUAL} \right) + \\ & D \sum_{i=1}^{n} \left(\left(0.03 \right) \text{ EMV1}_{iu} - \text{EMV1}_{iACTUAL} \right) + D \sum_{i=1}^{m} \left(\text{EMV2}_{iBASE} - \text{EMV2}_{iACTUAL} \right) + \\ & D \sum_{i=1}^{n} \left(\text{EWW1}_{ic} - \text{EWW1}_{iACTUAL} \right) + D \sum_{i=1}^{m} \left(\text{EWW2}_{iBASE} - \text{EWW2}_{iACTUAL} \right) \end{split}$$

where:

Credits and all terms of the equation are in units of megagrams per month, the baseline date is November 15, 1990, and

D=Discount factor=0.9 for all credit-generating emission points except those controlled by a pollution prevention measure, which will not be discounted.

EPV1_{iACTUAL}=Emissions for each Group 1 miscellaneous process vent i that is controlled to a level more stringent than the reference control technology, calculated according to paragraph (h)(2) of this section.

(0.02) EPV1_{iu}=Emissions from each Group 1 miscellaneous process vent i if the reference control technology had been applied to the uncontrolled emissions. EPV1_{iu} is calculated according to paragraph (h)(2) of

EPV2_{iBASE}=Emissions from each Group 2 miscellaneous process vent; at the baseline date, as calculated in paragraph (h)(2) of this section.

EPV2;ACTUAL=Emissions from each Group 2 miscellaneous process vent that is controlled, calculated according to paragraph (h)(2) of this section.

ES1_{iACTUAL}=Emissions from each Group 1 storage vessel i that is controlled to a level more stringent than the reference control technology, calculated according to paragraph (h)(3) of this section.

(0.05) ES1_{iu}=Emissions from each Group 1 storage vessel i if the reference control technology had been applied to the uncon-

trolled emissions. ES1iu is calculated according to paragraph (h)(3) of this section. ES2_{iACTUAL}=Emissions from each Group 2

storage vessel i that is controlled, calculated according to paragraph (h)(3) of this section.

ES2_{iBASE}=Emissions from each Group 2 storage vessel i at the baseline date, as calculated in paragraph (h)(3) of this section.

EGLR1_{iACTUAL}=Emissions from each Group 1 gasoline loading rack i that is controlled to a level more stringent than the reference control technology, calculated according to paragraph (h)(4) of this section.

EGLRic=Emissions from each Group 1 gasoline loading rack i if the reference control technology had been applied to the uncontrolled emissions. EGLRiu is calculated according to paragraph (h)(4) of this section.

EGRL2_{iACTUAL}=Emissions from each Group 2 gasoline loading rack i that is controlled, calculated according to paragraph (h)(4) of this section.

EGLR2_{iBASE}=Emissions from each Group 2 gasoline loading rack i at the baseline date, as calculated in paragraph (h)(4) of this section.

EMV1;ACTUAL=Emissions from each Group 1 marine tank vessel i that is controlled to a level more stringent than the reference control technology, calculated according to paragraph (h)(4) of this section.

§ 63.652

(0.03)EMV1_{in}=Emissions from each Group 1 marine tank vessel i if the reference control technology had been applied to the uncontrolled emissions. EMV1iu is calculated according to paragraph (h)(5) of this section.

 $EMV2_{iACTUAL}$ =Emissions from each Group 2 marine tank vessel i that is controlled, calculated according to paragraph (h)(5) of this section.

EMV2iBASE=Emissions from each Group 2 marine tank vessel i at the baseline date, as calculated in paragraph (h)(5) of this sec-

EWW1_{iACTUAL}=Emissions from each Group 1 wastewater stream i that is controlled to a level more stringent than the reference control technology, calculated according to paragraph (h)(6) of this section.

EWW1ic=Emissions from each Group 1 wastewater stream i if the reference control technology had been applied to the uncontrolled emissions, calculated according to paragraph (h)(6) of this section.

EWW2_{iACTUAL}=Emissions from each Group 2 wastewater stream i that is controlled, calculated according to paragraph (h)(6) of this section.

EWW2_{iBASE}=Emissions from each Group 2 wastewater stream i at the baseline date, calculated according to paragraph (h)(6) of this section.

n=Number of Group 1 emission points included in the emissions average. The value of n is not necessarily the same for each kind of emission point.

m=Number of Group 2 emission points included in the emissions average. The value of m is not necessarily the same for each kind of emission point.

(i) For an emission point controlled using a reference control technology, the percentage of reduction for calculating credits shall be no greater than the nominal efficiency associated with the reference control technology, unless a higher nominal efficiency is assigned as specified in paragraph (h)(1)(ii) of this section.

(ii) For an emission point controlled to a level more stringent than the reference control technology, the nominal efficiency for calculating credits shall be assigned as described in paragraph (i) of this section. A reference control technology may be approved for use in a different manner and assigned a higher nominal efficiency according to the procedures in paragraph (i) of this section.

(iii) For an emission point controlled using a pollution prevention measure, the nominal efficiency for calculating credits shall be determined as described in paragraph (j) of this section.

(2) Emissions from process vents shall be determined as follows:

(i) Uncontrolled emissions from miscellaneous process vents, EPV1iu, shall be calculated according to the procedures and equation for EPViu in paragraphs (g)(2)(i) and (g)(2)(ii) of this section.

(ii) Actual emissions from miscellaneous process vents controlled using a technology with an approved nominal efficiency greater than 98 percent or a pollution prevention measure achieving greater than 98 percent emission reduction, EPV1_{iACTUAL}, shall be calculated according to the following equation:

$$EPV1_{iACTUAL} = EPV1_{iu} \left(1 - \frac{Nominal efficiency\%}{100\%} \right)$$

(iii) The following procedures shall be used to calculate actual emissions from Group 2 process vents, EPV2_{iACTUAL}:

(A) For a Group 2 process vent controlled by a control device, a recovery device applied as a pollution prevention project, or a pollution prevention measure, if the control achieves a percentage of reduction less than or equal to a 98 percent reduction,

$$EPV2_{iACTUAL} = EPV2_{iu} \times \left(1 - \frac{Percent\ reduction}{100\%}\right)$$

40 CFR Ch. I (7-1-12 Edition)

- (1) EPV2iu shall be calculated according to the equations and procedures for EPV_{iu} in paragraphs (g)(2)(i) and (g)(2)(ii) of this section except as provided in paragraph (h)(2)(iii)(A)(3) of this section.
- (2) The percentage of reduction shall be calculated according to the procedures in paragraphs (g)(2)(iii)(B)(1) through (g)(2)(iii)(B)(3) of this section except as provided in paragraph (h)(2)(iii)(A)(4) of this section.
- (3) If a recovery device was added as part of a pollution prevention project, $\mathrm{EPV2}_{iu}$ shall be calculated prior to that recovery device. The equation for $EPV_{iu} \ in \ paragraph \ (g)(2)(ii)$ of this sec-

tion shall be used to calculate EPV2iu; however, the sampling site for measurement of vent stream flow rate and organic HAP concentration shall be at the inlet of the recovery device.

- (4) If a recovery device was added as part of a pollution prevention project, the percentage of reduction shall be demonstrated by conducting a performance test at the inlet and outlet of that recovery device.
- (B) For a Group 2 process vent controlled using a technology with an approved nominal efficiency greater than a 98 percent or a pollution prevention measure achieving greater than 98 percent reduction,

$$EPV2_{iACTUAL} = EPV2_{iu} \left(1 - \frac{Nominal efficiency\%}{100\%} \right)$$

- (iv) Emissions from Group 2 process vents at baseline, EPV2_{iBASE}, shall be calculated as follows:
- (A) If the process vent was uncontrolled on November 15, 1990, $EPV2_{iBASE}$ = $EPV2_{iu}$, and shall be cal-

culated according to the procedures and equation for EPV_{in} in paragraphs (g)(2)(i) and (g)(2)(ii) of this section.

(B) If the process vent was controlled on November 15, 1990,

$$EPV2_{BASE} = EPV2_{iu} \left(1 - \frac{Percent reduction\%}{100\%} \right)$$

- where EPV2iu is calculated according to the procedures and equation for EPViu in paragraphs (g)(2)(i) and (g)(2)(ii) of this section. The percentage of reduction shall be calculated according to the procedures specified in paragraphs (g)(2)(iii)(B)(1) through (g)(2)(iii)(B)(3) of this section.
- (C) If a recovery device was added to a process vent as part of a pollution prevention project initiated after November 15, 1990, $EPV2_{iBASE} = EPV2_{iu}$, where EPV2_{in} is calculated according to paragraph (h)(2)(iii)(A)(3) of this sec-
- (3) Emissions from storage vessels shall be determined as specified in §63.150(h)(3) of subpart G, except as fol-
- (i) All references to §63.119(b) in §63.150(h)(3) of subpart G shall be re-

placed with: §63.119 (b) or §63.119(b) except for $\S63.119(b)(5)$ and (b)(6).

- (ii) All references to §63.119(c) in §63.150(h)(3) of subpart G shall be replaced with: §63.119(c) or §63.119(c) except for §63.119(c)(2).
- (iii) All references to §63.119(d) in §63.150(h)(3) of subpart G shall be replaced with: §63.119(d) or §63.119(d) except for §63.119(d)(2).
- (4) Emissions from gasoline loading racks shall be determined as follows:
- (i) Uncontrolled emissions from Group 1 gasoline loading racks, EGLR1_{iu}, shall be calculated according to the procedures and equations for EGLR_{iu} as described in paragraphs (g)(4)(i) through (g)(4)(iv) of this section.

§ 63.652

(ii) Emissions from Group 1 gasoline loading racks if the reference control technology had been applied, EGLR_{ic}, shall be calculated according to the procedures and equations in paragraph (g)(4)(v) of this section.

(iii) Actual emissions from Group 1 gasoline loading racks controlled to less than 10 milligrams of TOC per liter of gasoline loaded; $EGLR_{iACTUAL}$, shall be calculated according to the following equation:

$$EGLR1_{iACTUAL} = EGLR1_{iu} \left(1 - \frac{Nominal efficiency}{100\%} \right)$$

(iv) The following procedures shall be used to calculate actual emissions from Group 2 gasoline loading racks, $EGLR2_{iACTUAL}$:

(A) For a Group 2 gasoline loading rack controlled by a control device or a pollution prevention measure achieving emissions reduction but where emissions are greater than the 10 milligrams of TOC per liter of gasoline loaded requirement,

$$EGLR2_{iACTUAL} = EGLR2_{iu} \left(1 - \frac{Percent\ reduction}{100\%} \right)$$

(1) EGLR2iu shall be calculated according to the equations and procedures for $EGLR_{iu}$ in paragraphs (g)(4)(i)through (g)(4)(iv) of this section.

(2) The percentage of reduction shall be calculated according to the procedures in paragraphs (g)(4)(vi)(B)(1) and (g)(4)(vi)(B)(2) of this section.

(B) For a Group 2 gasoline loading rack controlled by using a technology with an approved nominal efficiency greater than 98 percent or a pollution prevention measure achieving greater than a 98-percent reduction,

$$EGLR2_{iACTUAL} = EGLR2_{iu} \left(1 - \frac{Nominal\ efficiency}{100\%} \right)$$

(v) Emissions from Group 2 gasoline loading racks at baseline, EGLR2_{iBASE}, shall be calculated as follows:

(A) If the gasoline loading rack was uncontrolled on November 15, 1990, $EGLR2_{iBASE}$ = $EGLR2_{iu}$, and shall be calculated according to the procedures and equations for $EGLR_{iu}$ in paragraphs (g)(4)(i) through (g)(4)(iv) of this section.

(B) If the gasoline loading rack was controlled on November 15, 1990,

$$EGLR2_{iBASE} = EGLR2_{iu} \left(1 - \frac{Percent\ reduction}{100\%} \right)$$

40 CFR Ch. I (7-1-12 Edition)

where EGLR2iu is calculated according to the procedures and equations for EGLRiu in paragraphs (g)(4)(i) through (g)(4)(iv) of this section. Percentage of reduction shall be calculated according to the procedures in paragraphs (g)(4)(vi)(B)(1)(g)(4)(vi)(B)(2) of this section.

- (5) Emissions from marine tank vessels shall be determined as follows:
- (i) Uncontrolled emissions from Group 1 marine tank vessels, EMV1iu,

shall be calculated according to the procedures and equations for EMV_{in} as described in paragraph (g)(5)(i) of this

(ii) Actual emissions from Group 1 marine tank vessels controlled using a technology or pollution prevention measure with an approved nominal efficiency greater than 97 percent, EMV_{iACTUAL}, shall be calculated according to the following equation:

$$EMV1_{iACTUAL} = EMV1_{iu} \left(1 - \frac{Nominal efficiency}{100\%} \right)$$

(iii) The following procedures shall be used to calculate actual emissions from Group 2 marine tank vessels, $EMV2_{iACTUAL}$:

(A) For a Group 2 marine tank vessel controlled by a control device or a pollution prevention measure achieving a percentage of reduction less than or equal to 97 percent reduction,

$$EMV2_{iACTUAL} = EMV2_{iu} \left(1 - \frac{Percent\ reduction}{100\%} \right)$$

- (1) EMV2iu shall be calculated according to the equations and procedures for EMV_{iu} in paragraph (g)(5)(i) of this sec-
- (2) The percentage of reduction shall be calculated according to the proce-

dures in paragraphs (g)(5)(ii)(B)(1) and (g)(5)(ii)(B)(2) of this section.

(B) For a Group 2 marine tank vessel controlled using a technology or a pollution prevention measure with an approved nominal efficiency greater than 97 percent.

$$EMV2_{iACTUAL} = EMV2_{iu} \left(1 - \frac{Nominal efficiency}{100\%} \right)$$

- (iv) Emissions from Group 2 marine tank vessels at baseline, EMV2_{iBASE}, shall be calculated as follows:
- (A) If the marine terminal was uncontrolled on November 15, 1990, EMV2_{iBASE} equals EMV2_{iu}, and shall be

calculated according to the procedures and equations for EMV_{iu} in paragraph (g)(5)(i) of this section.

(B) If the marine tank vessel was controlled on November 15, 1990,

$$EMV2_{iBASE} = EMV2_{iu} \left(1 - \frac{Percent\ reduction}{100\%} \right)$$

§ 63.652

where $EMV2_{iu}$ is calculated according to the procedures and equations for EMV_{iu} in paragraph (g)(5)(i) of this section. Percentage of reduction shall be calculated according to the procedures in paragraphs (g)(5)(ii)(B)(1) and (g)(5)(ii)(B)(2) of this section.

- (6) Emissions from wastewater shall be determined as follows:
- (i) For purposes of paragraphs (h)(4)(ii) through (h)(4)(vi) of this section, the following terms will have the meaning given them in paragraphs (h)(6)(i)(A) through (h)(6)(i)(C) of this section.
- (A) Correctly suppressed means that a wastewater stream is being managed according to the requirements of §§ 61.343 through 61.347 or

§61.342(c)(1)(iii) of 40 CFR part 61, subpart FF, as applicable, and the emissions from the waste management units subject to those requirements are routed to a control device that reduces HAP emissions by 95 percent or greater.

- (B) Treatment process has the meaning given in §61.341 of 40 CFR part 61, subpart FF except that it does not include biological treatment units.
- (C) Vapor control device means the control device that receives emissions vented from a treatment process or treatment processes.
- (ii) The following equation shall be used for each wastewater stream i to calculate EWW_{ic} :

$$EWW_{ic} = \left(6.0*10^{-8}\right)Q_{i} H_{i} \sum_{m=1}^{s} \left(1 - Fr_{m}\right)Fe_{m} HAP_{im} + \left(0.05\right)\!\left(6.0*10^{-8}\right)Q_{i} H_{i} \sum_{m=1}^{s} \left(Fr_{m} HAP_{im}\right)$$

where:

 $\begin{array}{ll} EWW_{\rm ic} = Monthly \ was tewater \ stream \ emission \ rate \ if \ was tewater \ stream \ i \ were \ controlled \ by \ the \ reference \ control \ technology, megagrams per month. \end{array}$

 Q_i = Average flow rate for wastewater stream i, liters per minute.

H_i = Number of hours during the month that wastewater stream i was generated, hours per month.

 ${\rm Fr_m}{=}{\rm Fraction}$ removed of organic HAP m in wastewater, from table 7 of this subpart, dimensionless.

Fe_m=Fraction emitted of organic HAP m in wastewater from table 7 of this subpart, dimensionless.

s=Total number of organic HAP's in wastewater stream i.

HAP_{im}=Average concentration of organic HAP m in wastewater stream i, parts per million by weight.

(A) ${\rm HAP_{im}}$ shall be determined for the point of generation or at a location downstream of the point of generation. Wastewater samples shall be collected using the sampling procedures specified in Method 25D of 40 CFR part 60, appendix A. Where feasible, samples shall be taken from an enclosed pipe prior to the wastewater being exposed to the atmosphere. When sampling from an enclosed pipe is not feasible, a minimum of three representative samples shall be collected in a manner to minimize exposure of the sample to the atmosphere and loss of organic HAP's

prior to sampling. The samples collected may be analyzed by either of the following procedures:

- (1) A test method or results from a test method that measures organic HAP concentrations in the wastewater, and that has been validated pursuant to section 5.1 or 5.3 of Method 301 of appendix A of this part may be used; or
- (2) Method 305 of appendix A of this part may be used to determine $C_{\rm im}$, the average volatile organic HAP concentration of organic HAP m in wastewater stream i, and then HAP $_{\rm im}$ may be calculated using the following equation: HAP $_{\rm im}$ = $C_{\rm im}$ /Fm $_{\rm m}$, where Fm $_{\rm m}$ for organic HAP m is obtained from table 7 of this subpart.
- (B) Values for Q_i , HAP_{im} , and C_{im} shall be determined during a performance test conducted under representative conditions. The average value obtained from three test runs shall be used. The values of Q_i , HAP_{im} , and C_{im} shall be established in the Notification of Compliance Status report and must be updated as provided in paragraph (h)(6)(i)(C) of this section.
- (C) If there is a change to the process or operation such that the previously measured values of Q_i , HAP_{im} , and C_{im} are no longer representative, a new performance test shall be conducted to determine new representative values of

40 CFR Ch. I (7-1-12 Edition)

 Q_i , HAP_{im}, and C_{im} . These new values shall be used to calculate debits and credits from the time of the change forward, and the new values shall be reported in the next Periodic Report.

(iii) The following equations shall be used to calculate EWW1_{iACTUAL} for each Group 1 wastewater stream i that is correctly suppressed and is treated to a level more stringent than the reference control technology.

(A) If the Group 1 wastewater stream i is controlled using a treatment process or series of treatment processes with an approved nominal reduction efficiency for an individually speciated HAP that is greater than that specified in table 7 of this subpart, and the vapor control device achieves a percentage of reduction equal to 95 percent, the following equation shall be used:

$$EWW1_{iACTUAL} = \left(6.0*10^{-8}\right)Q_{i} H_{i} \sum_{m=1}^{s} \left[Fe_{m} HAP_{im} \left(1 - PR_{im}\right)\right] + 0.05 \left(6.0*10^{-8}\right)Q_{i} H_{i} \sum_{m=1}^{s} \left[HAP_{im} PR_{im}\right]$$

Where:

EWW_{iactual}=Monthly wastewater stream emission rate if wastewater stream i is treated to a level more stringent than the reference control technology, megagrams per month.

 PR_{im} =The efficiency of the treatment process, or series of treatment processes, that treat wastewater stream i in reducing the emission potential of organic HAP m in wastewater, dimensionless, as calculated by:

$$\mathrm{PR}_{\mathrm{im}} = \frac{\mathrm{HAP}_{\mathrm{im-in}} - \mathrm{HAP}_{\mathrm{im-out}}}{\mathrm{HAP}_{\mathrm{im-in}}}$$

Where:

 ${\rm HAP_{im\cdot in}}{=}{\rm Average}$ concentration of organic HAP m, parts per million by weight, as defined and determined according to paragraph ${\rm (h)(6)(ii)(A)}$ of this section, in the

wastewater entering the first treatment process in the series.

HAP_{im-out}=Average concentration of organic HAP m, parts per million by weight, as defined and determined according to paragraph (h)(6)(ii)(A) of this section, in the wastewater exiting the last treatment process in the series.

All other terms are as defined and determined in paragraph (h)(6)(ii) of this section.

(B) If the Group 1 wastewater stream i is not controlled using a treatment process or series of treatment processes with an approved nominal reduction efficiency for an individually speciated HAP that is greater than that specified in table 7 of this subpart, but the vapor control device has an approved nominal efficiency greater than 95 percent, the following equation shall be used:

$$EWW1_{iACTUAL} = \left(6.0*10^{-8}\right)Q_i H_i \sum_{m=1}^{s} \left[Fe_m HAP_{im} \left(1-A_m\right)\right] + \left(1-\frac{Nominal\ efficiency\ \%}{100}\right) \left(6.0*10^{-8}\right)Q_i H_i \sum_{m=1}^{s} \left[HAP_{im}A_m\right] + \left(1-\frac{Nominal\ efficiency\ \%}{100}\right) \left(6.0*10^{-8}\right)Q_i H_i + \left(1-\frac{Nominal\ efficiency\ \%}{100}\right) \left(6.0*10^{-8}\right)Q_i H_i + \left(1-\frac{Nominal\ efficiency\ \%}{100}\right)$$

Where:

Nominal efficiency=Approved reduction efficiency of the vapor control device, dimensionless, as determined according to the procedures in §63.652(i).

A_m=The efficiency of the treatment process, or series of treatment processes, that treat wastewater stream i in reducing the emission potential of organic HAP m in wastewater, dimensionless.

All other terms are as defined and determined in paragraphs (h)(6)(ii) and (h)(6)(iii)(A) of this section.

(1) If a steam stripper meeting the specifications in the definition of reference control technology for wastewater is used, A_m shall be equal to the value of Fr_m given in table 7 of this subpart.

(2) If an alternative control device is used, the percentage of reduction must be determined using the equation and methods specified in paragraph (h)(6)(iii)(A) of this section for determining PR_{im}. If the value of PR_{im} is greater than or equal to the value of

§ 63.652

Fr_m given in table 7 of this subpart, then A_m equals Fr_m unless a higher nominal efficiency has been approved. If a higher nominal efficiency has been approved for the treatment process, the owner or operator shall determine EWW1_{iACTUAL} according to paragraph (h)(6)(iii)(B) of this section rather than paragraph (h)(6)(iii)(A) of this section. If PR_{im} is less than the value of FR_m given in table 7 of this subpart, emissions averaging shall not be used for this emission point.

(C) If the Group 1 wastewater stream i is controlled using a treatment process or series of treatment processes with an approved nominal reduction efficiency for an individually speciated hazardous air pollutant that is greater than that specified in table 7 of this subpart, and the vapor control device has an approved nominal efficiency greater than 95 percent, the following equation shall be used:

$$EWW1_{iACTUAL} = \left(6.0*10^{-8}\right)Q_{i}H_{i}\sum_{m=1}^{s}\left[Fe_{m}HAP_{im}\left(1-PR_{im}\right)\right] + \left(1-\frac{Nominal\,efficiency\,\%}{100}\right)\!\left(6.0*10^{-8}\right)Q_{i}\,H_{i}\sum_{m=1}^{s}\left[HAP_{im}PR_{im}\right] + \left(1-\frac{Nominal\,efficiency\,\%}{100}\right)\!\left(6.0*10^{-8}\right)Q_{i}\,H_{i}\sum_{m=1}^{s}\left[HAP_{im}PR_{im}\right] + \left(1-\frac{Nominal\,efficiency\,\%}{100}\right)\!\left(6.0*10^{-8}\right)Q_{i}\,H_{i}\sum_{m=1}^{s}\left[HAP_{im}PR_{im}\right] + \left(1-\frac{Nominal\,efficiency\,\%}{100}\right)\left(6.0*10^{-8}\right)Q_{i}\,H_{i}\sum_{m=1}^{s}\left[HAP_{im}PR_{im}\right] + \left(1-\frac{Nominal\,efficiency\,\%}{100}\right)Q_{i}^{2}$$

where all terms are as defined and determined in paragraphs (h)(6)(ii) and (h)(6)(iii)(A) of this section.

(iv) The following equation shall be used to calculate EWW2_{iBASE} for each Group 2 wastewater stream i that on November 15, 1990 was not correctly suppressed or was correctly suppressed but not treated:

EWW2_{iBASE} =
$$(6.0 \times 10^{-8})$$
Q_i H_i $\sum_{m=1}^{s}$ Fe_mHAP_{im}

Where:

EWW2_{iBASE}=Monthly wastewater stream emission rate if wastewater stream i is not correctly suppressed, megagrams per month.

 Q_{i} , H_{i} , s, Fe_{m} , and HAP_{im} are as defined and determined according to paragraphs (h)(6)(ii) and (h)(6)(iii)(A) of this section.

(v) The following equation shall be used to calculate EWW2_{iBASE} for each

Group 2 wastewater stream i on November 15, 1990 was correctly suppressed. $EWW2_{iBASE}$ shall be calculated as if the control methods being used on November 15, 1990 are in place and any control methods applied after November 15, 1990 are ignored. However, values for the parameters in the equation shall be representative of present production levels and stream properties.

$$EWW2_{iBASE} = \left(6.0*10^{-8}\right)Q_{i} \; H_{i} \; \sum_{m=1}^{s} \left[Fe_{m} HAP_{im} \left(1-PR_{im}\right)\right] + \left(1-\frac{R_{i}}{100\%}\right) \left(6.0*10^{-8}\right)Q_{i} \; H_{i} \; \sum_{m=1}^{s} \left[HAP_{im} PR_{im}\right] + \left(1-\frac{R_{i}}{100\%}\right) \left(6.0*10^{-8}\right)Q_{i} \; H_{i} \; \sum_{m=1}^{s} \left[HAP_{im} PR_{im}\right] + \left(1-\frac{R_{i}}{100\%}\right) \left(6.0*10^{-8}\right)Q_{i} \; H_{i} \; \sum_{m=1}^{s} \left[HAP_{im} PR_{im}\right] + \left(1-\frac{R_{i}}{100\%}\right) \left(6.0*10^{-8}\right)Q_{i} \; H_{i} \; \sum_{m=1}^{s} \left[HAP_{im} PR_{im}\right] + \left(1-\frac{R_{i}}{100\%}\right) \left(6.0*10^{-8}\right)Q_{i} \; H_{i} \; \sum_{m=1}^{s} \left[HAP_{im} PR_{im}\right] + \left(1-\frac{R_{i}}{100\%}\right) \left(6.0*10^{-8}\right)Q_{i} \; H_{i} \; \sum_{m=1}^{s} \left[HAP_{im} PR_{im}\right] + \left(1-\frac{R_{i}}{100\%}\right) \left(6.0*10^{-8}\right)Q_{i} \; H_{i} \; \sum_{m=1}^{s} \left[HAP_{im} PR_{im}\right] + \left(1-\frac{R_{i}}{100\%}\right)Q_{i} \; H_{i} \; D_{i} \; H_{i} \; D_{i} \; D_{i} \; H_{i} \; D_{i} \; D_{i} \; H_{i} \; D_{i} \; D$$

where R_i is calculated according to paragraph (h)(6)(vii) of this section and all other terms are as defined and determined according to paragraphs (h)(6)(ii) and (h)(6)(iii)(A) of this section.

(vi) For Group 2 wastewater streams that are correctly suppressed. EWW2_{iACTUAL} shall be calculated ac-

cording to the equation for EWW2iBASE in paragraph (h)(6)(v) of this section. EWW2_{iACTUAL} shall be calculated with all control methods in place accounted

- (vii) The reduction efficiency, Ri, of the vapor control device shall be demonstrated according to the following procedures:
- (A) Sampling sites shall be selected using Method 1 or 1A of 40 CFR part 60, appendix A, as appropriate.
- (B) The mass flow rate of organic compounds entering and exiting the control device shall be determined as
- (1) The time period for the test shall not be less than 3 hours during which at least three runs are conducted.
- (2) A run shall consist of a 1-hour period during the test. For each run:
- (i) The volume exhausted shall be determined using Methods 2, 2A, 2C, or 2D of 40 CFR part 60 appendix A, as appro-
- (ii) The organic concentration in the vent stream entering and exiting the control device shall be determined using Method 18 of 40 CFR part 60, appendix A. Alternatively, any other test method validated according to the procedures in Method 301 of appendix A of this part may be used.
- (3) The mass flow rate of organic compounds entering and exiting the control device during each run shall be calculated as follows:

$$E_{a} = \frac{0.0416}{10^{6} \times m} \left[\sum_{p=1}^{m} V_{ap} \left(\sum_{p=1}^{n} C_{aip} MW_{i} \right) \right]$$

$$E_{b} = \frac{0.0416}{10^{6} \times m} \left[\sum_{p=1}^{m} V_{bp} \left(\sum_{p=1}^{n} C_{bip} MW_{i} \right) \right]$$

Ea=Mass flow rate of organic compounds exiting the control device, kilograms per

E_b=Mass flow rate of organic compounds entering the control device, kilograms per

 V_{ap} =Average volumetric flow rate of vent stream exiting the control device during run p at standards conditions, cubic meters per hour.

 V_{bp} = Average volumetric flow rate of vent stream entering the control device during run p at standards conditions, cubic meters per hour.

p = Run.

m = Number of runs.

Caip = Concentration of organic compound i measured in the vent stream exiting the

40 CFR Ch. I (7-1-12 Edition)

control device during run p as determined by Method 18 of 40 CFR part 60 appendix A, parts per million by volume on a dry basis.

C_{bip} = Concentration of organic compound i measured in the vent stream entering the control device during run p as determined by Method 18 of 40 CFR part 60, appendix A, parts per million by volume on a dry basis.

MWi = Molecular weight of organic compound i in the vent stream, kilograms per kilogram-mole.

n = Number of organic compounds in the vent stream.

0.0416 = Conversion factor for molar volume. kilograms-mole per cubic meter at 293 kelvin and 760 millimeters mercury absolute.

(C) The organic reduction efficiency for the control device shall be calculated as follows:

$$R = \frac{E_b - E_a}{E_b} \times 100$$

R = Total organic reduction efficiency for the control device, percentage.

E_b = Mass flow rate of organic compounds entering the control device, kilograms per hour.

Ea = Mass flow rate of organic compounds exiting the control device, kilograms per

- (i) The following procedures shall be followed to establish nominal efficiencies. The procedures in paragraphs (i)(1) through (i)(6) of this section shall be followed for control technologies that are different in use or design from the reference control technologies and achieve greater percentages of reduction than the percentages of efficiency assigned to the reference control technologies in §63.641.
- (1) In those cases where the owner or operator is seeking permission to take credit for use of a control technology that is different in use or design from the reference control technology, and the different control technology will be used in more than three applications at a single plant site, the owner or operator shall submit the information specified in paragraphs (i)(1)(i) through (i)(1)(iv) of this section to the Administrator in writing:
- (i) Emission stream characteristics of each emission point to which the control technology is or will be applied including the kind of emission point, flow, organic HAP concentration, and

§ 63.652

all other stream characteristics necessary to design the control technology or determine its performance;

(ii) Description of the control technology including design specifications;

- (iii) Documentation demonstrating to the Administrator's satisfaction the control efficiency of the control technology. This may include performance test data collected using an appropriate EPA method or any other method validated according to Method 301 of appendix A of this part. If it is infeasible to obtain test data, documentation may include a design evaluation and calculations. The engineering basis of the calculation procedures and all inputs and assumptions made in the calculations shall be documented; and
- (iv) A description of the parameter or parameters to be monitored to ensure that the control technology will be operated in conformance with its design and an explanation of the criteria used for selection of that parameter (or parameters).
- (2) The Administrator shall determine within 120 calendar days whether an application presents sufficient information to determine nominal efficiency. The Administrator reserves the right to request specific data in addition to the items listed in paragraph (i)(1) of this section.
- (3) The Administrator shall determine within 120 calendar days of the submittal of sufficient data whether a control technology shall have a nominal efficiency and the level of that nominal efficiency. If, in the Administrator's judgment, the control technology achieves a level of emission reduction greater than the reference control technology for a particular kind of emission point, the Administrator will publish a FEDERAL REGISTER notice establishing a nominal efficiency for the control technology.
- (4) The Administrator may grant conditional permission to take emission credits for use of the control technology on requirements that may be necessary to ensure operation and maintenance to achieve the specified nominal efficiency.
- (5) In those cases where the owner or operator is seeking permission to take credit for use of a control technology that is different in use or design from

the reference control technology and the different control technology will be used in no more than three applications at a single plant site, the information listed in paragraphs (i)(1)(i) through (i)(1)(iv) of this section can be submitted to the permitting authority for the source for approval instead of the Administrator.

- (i) In these instances, use and conditions for use of the control technology can be approved by the permitting authority. The permitting authority shall follow the procedures specified in paragraphs (i)(2) through (i)(4) of this section except that, in these instances, a FEDERAL REGISTER notice is not required to establish the nominal efficiency for the different technology.
- (ii) If, in reviewing the submittal, the permitting authority believes the control technology has broad applicability for use by other sources, the permitting authority shall submit the information provided in the application to the Director of the EPA Office of Air Quality Planning and Standards. The Administrator shall review the technology for broad applicability and may publish a FEDERAL REGISTER notice; however, this review shall not affect the permitting authority's approval of the nominal efficiency of the control technology for the specific application.
- (6) If, in reviewing an application for a control technology for an emission point, the Administrator or permitting authority determines the control technology is not different in use or design from the reference control technology, the Administrator or permitting authority shall deny the application.
- (j) The following procedures shall be used for calculating the efficiency (percentage of reduction) of pollution prevention measures:
- (1) A pollution prevention measure is any practice that meets the criteria of paragraphs (j)(1)(i) and (j)(1)(ii) of this section.
- (i) A pollution prevention measure is any practice that results in a lesser quantity of organic HAP emissions per unit of product released to the atmosphere prior to out-of-process recycling, treatment, or control of emissions while the same product is produced.

- 40 CFR Ch. I (7-1-12 Edition)
- (ii) Pollution prevention measures may include: Substitution of feedstocks that reduce HAP emissions, alterations to the production process to reduce the volume of materials released to the environment, equipment modifications; housekeeping measures, and in-process recycling that returns waste materials directly to production as raw materials. Production cutbacks do not qualify as pollution prevention.
- (2) The emission reduction efficiency of pollution prevention measures implemented after November 15, 1990 can be used in calculating the actual emissions from an emission point in the debit and credit equations in paragraphs (g) and (h) of this section.
- (i) For pollution prevention measures, the percentage of reduction used in the equations in paragraphs (g)(2)and (g)(3) of this section and paragraphs (h)(2) through (h)(4) of this section is the difference in percentage between the monthly organic HAP emissions for each emission point after the pollution prevention measure for the most recent month versus monthly emissions from the same emission point before the pollution prevention measure, adjusted by the volume of product produced during the two monthly periods.
- (ii) The following equation shall be used to calculate the percentage of reduction of a pollution prevention measure for each emission point.

$$Percent \ reduction = \frac{E_B \frac{\left(E_{pp} \times P_B\right)}{P_{pp}}}{E_B} \times 100\%$$

Where:

Percent reduction=Efficiency of pollution prevention measure (percentage of organic HAP reduction).

E_B=Monthly emissions before the pollution prevention measure, megagrams per month, determined as specified in paragraphs (j)(2)(ii)(A), (j)(2)(ii)(B),(j)(2)(ii)(C) of this section.

 E_{pp} =Monthly emissions after the pollution prevention measure, megagrams per month, as determined for the most recent month, determined as specified in paragraphs (j)(2)(ii)(D) or (j)(2)(ii)(E) of this section.

 P_B =Monthly production before the pollution prevention measure, megagrams

month, during the same period over which E_B is calculated.

 P_{pp} =Monthly production after the pollution prevention measure, megagrams per month, as determined for the most recent

(A) The monthly emissions before the pollution prevention measure, EB, shall be determined in a manner consistent with the equations and procedures in paragraphs (g)(2), (g)(3), (g)(4), and (g)(5) of this section for miscellaneous process vents, storage vessels, gasoline loading racks, and marine tank vessels.

(B) For wastewater, E_B shall be calculated as follows:

$$E_{B} = \sum_{i=1}^{n} \left[\left(6.0 * 10^{-8} \right) Q_{Bi} H_{Bi} \sum_{m=1}^{s} Fe_{m} HAP_{Bim} \right]$$

where:

n=Number of wastewater streams.

Q_{Bi}=Average flow rate for wastewater stream i before the pollution prevention measure, liters per minute.

HBi=Number of hours per month that wastewater stream i was discharged before the pollution prevention measure, hours per month.

s=Total number of organic HAP's in wastewater stream i.

§ 63.652

Fem=Fraction emitted of organic HAP m in wastewater from table 7 of this subpart, dimensionless.

HAP_{Bim}=Average concentration of organic HAP m in wastewater stream i, defined and determined according to paragraph (h)(6)(ii)(A)(2) of this section, before the pollution prevention measure, parts per million by weight, as measured before the implementation of the pollution measure.

(C) If the pollution prevention measure was implemented prior to July 14, 1994, records may be used to determine

(D) The monthly emissions after the pollution prevention measure, Epp, may be determined during a performance test or by a design evaluation and documented engineering calculations. Once an emissions-to-production ratio has been established, the ratio can be used to estimate monthly emissions from monthly production records.

(E) For wastewater, $E_{pp}\ shall$ be calculated using the following equation:

$$E_{pp} = \sum_{i=1}^{n} \left[\left(6.0 * 10^{-8} \right) Q_{ppi} H_{ppi} \sum_{m=1}^{s} Fe_{m} HAP_{ppim} \right]$$

where $n,\ Q,\ H,\ s,\ Fe_m,\ and\ HAP$ are defined and determined as described in paragraph (j)(2)(ii)(B) of this section except that Q_{ppi} , H_{ppi} , and HAP_{ppim} shall be determined after the pollution prevention measure has been implemented.

(iii) All equations, calculations, test procedures, test results, and other information used to determine the percentage of reduction achieved by a pollution prevention measure for each emission point shall be fully documented.

(iv) The same pollution prevention measure may reduce emissions from multiple emission points. In such cases, the percentage of reduction in emissions for each emission point must be calculated.

(v) For the purposes of the equations in paragraphs (h)(2) through (h)(6) of this section used to calculate credits for emission points controlled more stringently than the reference control technology, the nominal efficiency of a pollution prevention measure is equivalent to the percentage of reduction of the pollution prevention measure. When a pollution prevention measure is used, the owner or operator of a source is not required to apply to the Administrator for a nominal efficiency and is not subject to paragraph (i) of

(k) The owner or operator shall demonstrate that the emissions from the emission points proposed to be included in the average will not result in greater hazard or, at the option of the State or local permitting authority, greater risk to human health or the environment than if the emission points were controlled according to the provisions in §§ 63.643 through 63.647, and §§ 63.650 and 63.651.

(1) This demonstration of hazard or risk equivalency shall be made to the satisfaction of the State or local permitting authority.

(i) The State or local permitting authority may require owners and operators to use specific methodologies and procedures for making a hazard or risk determination.

(ii) The demonstration and approval of hazard or risk equivalency may be made according to any guidance that the EPA makes available for use.

(2) Owners and operators shall provide documentation demonstrating the hazard or risk equivalency of their proposed emissions average in their Implementation Plan.

(3) An emissions averaging plan that does not demonstrate an equivalent or lower hazard or risk to the satisfaction of the State or local permitting authority shall not be approved. The State or local permitting authority may require such adjustments to the emissions averaging plan as are necessary in order to ensure that the average will not result in greater hazard or risk to human health or the environment than would result if the emission points were controlled according to

40 CFR Ch. I (7-1-12 Edition)

§§ 63.643 through 63.647, and §§ 63.650 and 63.651.

- (4) A hazard or risk equivalency demonstration shall:
- (i) Be a quantitative, bona fide chemical hazard or risk assessment;
- (ii) Account for differences in chemical hazard or risk to human health or the environment; and
- (iii) Meet any requirements set by the State or local permitting authority for such demonstrations.
- (1) For periods of excess emissions, an owner or operator may request that the provisions of paragraphs (1)(1) through (1)(4) of this section be followed instead of the procedures in paragraphs (f)(3)(i) and (f)(3)(ii) of this section.
- (1) The owner or operator shall notify the Administrator of excess emissions in the Periodic Reports as required in §63.655(g)(6).
- (2) The owner or operator shall demonstrate that other types of monitoring data or engineering calculations are appropriate to establish that the control device for the emission point was operating in such a fashion to warrant assigning full or partial credits and debits. This demonstration shall be made to the Administrator's satisfaction, and the Administrator may establish procedures for demonstrating compliance that are acceptable.
- (3) The owner or operator shall provide documentation of the period of excess emissions and the other type of monitoring data or engineering calculations to be used to demonstrate that the control device for the emission point was operating in such a fashion to warrant assigning full or partial credits and debits.
- (4) The Administrator may assign full or partial credit and debits upon review of the information provided.

[60 FR 43260, Aug. 18, 1995; 60 FR 49976, Sept. 27, 1995; 61 FR 7051, Feb. 23, 1996, as amended at 61 FR 29881, June 12, 1996; 61 FR 33799, June 28, 1996; 74 FR 55686, Oct. 28, 2009]

§63.653 Monitoring, recordkeeping, and implementation plan for emissions averaging.

(a) For each emission point included in an emissions average, the owner or operator shall perform testing, monitoring, recordkeeping, and reporting equivalent to that required for Group 1 emission points complying with §§ 63.643 through 63.647, and §§ 63.650 and 63.651. The specific requirements for miscellaneous process vents, storage vessels, wastewater, gasoline loading racks, and marine tank vessels are identified in paragraphs (a)(1) through (a)(7) of this section.

- (1) The source shall implement the following testing, monitoring, record-keeping, and reporting procedures for each miscellaneous process vent equipped with a flare, incinerator, boiler, or process heater:
- (i) Conduct initial performance tests to determine the percentage of reduction as specified in §63.645 of this subpart and §63.116 of subpart G; and
- (ii) Monitor the operating parameters specified in §63.644, as appropriate for the specific control device.
- (2) The source shall implement the following procedures for each miscellaneous process vent, equipped with a carbon adsorber, absorber, or condenser but not equipped with a control device:
- (i) Determine the flow rate and organic HAP concentration using the methods specified in §63.115 (a)(1) and (a)(2), §63.115 (b)(1) and (b)(2), and §63.115(c)(3) of subpart G; and
- (ii) Monitor the operating parameters specified in §63.114 of subpart G, as appropriate for the specific recovery device.
- (3) The source shall implement the following procedures for each storage vessel controlled with an internal floating roof, external roof, or a closed vent system with a control device, as appropriate to the control technique:
- (i) Perform the monitoring or inspection procedures in §63.646 of this subpart and §63.120 of subpart G; and
- (ii) For closed vent systems with control devices, conduct an initial design evaluation as specified in §63.646 of this subpart and §63.120(d) of subpart G.
- (4) For each gasoline loading rack that is controlled, perform the testing and monitoring procedures specified in §§ 63.425 and 63.427 of subpart R of this part except § 63.425(d) or § 63.427(c).
- (5) For each marine tank vessel that is controlled, perform the compliance, monitoring, and performance testing, procedures specified in §§ 63.563, 63.564, and 63.565 of subpart Y of this part.

40 CFR Ch. I (7-1-12 Edition)

which the potline is meeting the TF emission limit.

Document #1421812

(4) The alternative opacity limit established in paragraph (i)(3) of this section shall not be greater than 20 percent opacity.

§63.846 Emission averaging.

- (a) General. The owner or operator of an existing potline or anode bake furnace in a State that does not choose to exclude emission averaging in the approved operating permit program may demonstrate compliance by emission averaging according to the procedures in this section.
- (b) Potlines. The owner or operator may average TF emissions from potlines and demonstrate compliance with the limits in Table 1 of this subpart using the procedures in paragraphs (b)(1) and (b)(2) of this section. The owner or operator also may average POM emissions from potlines and demonstrate compliance with the limits in Table 2 of this subpart using the procedures in paragraphs (b)(1) and (b)(3) of this section.
- (1) Monthly average emissions of TF and/or quarterly average emissions of POM shall not exceed the applicable emission limit in Table 1 of this subpart (for TF emissions) and/or Table 2 of this subpart (for POM emissions). The emission rate shall be calculated based on the total emissions from all potlines over the period divided by the quantity of aluminum produced during the period, from all potlines comprising the averaging group.
- (2) To determine compliance with the applicable emission limit in Table 1 of this subpart for TF emissions, the owner or operator shall determine the monthly average emissions (in lb/ton) from each potline from at least three runs per potline each month for TF secondary emissions using the procedures and methods in §§63.847 and 63.849. The owner or operator shall combine the results of secondary TF monthly average emissions with the TF results for the primary control system and divide total emissions by total aluminum production.
- (3) To determine compliance with the applicable emission limit in Table 2 of this subpart for POM emissions, the owner or operator shall determine the

quarterly average emissions (in lb/ton) from each potline from at least one run each month for POM emissions using the procedures and methods in §§ 63.847 and 63.849. The owner or operator shall combine the results of secondary POM quarterly average emissions with the POM results for the primary control system and divide total emissions by total aluminum production.

- (c) Anode bake furnaces. The owner or operator may average TF emissions from anode bake furnaces and demonstrate compliance with the limits in Table 3 of this subpart using the procedures in paragraphs (c)(1) and (c)(2) of this section. The owner or operator also may average POM emissions from anode bake furnaces and demonstrate compliance with the limits in Table 3 of this subpart using the procedures in paragraphs (c)(1) and (c)(2) of this section.
- (1) Annual emissions of TF and/or POM from a given number of anode bake furnaces making up each averaging group shall not exceed the applicable emission limit in Table 3 of this subpart in any one year; and
- (2) To determine compliance with the applicable emission limit in Table 3 of this subpart for anode bake furnaces, the owner or operator shall determine TF and/or POM emissions from the control device for each furnace at least once a year using the procedures and methods in §§ 63.847 and 63.849.
- (d) Implementation plan. The owner or operator shall develop and submit an implementation plan for emission averaging to the applicable regulatory authority for review and approval according to the following procedures and requirements:
- (1) Deadlines. The owner or operator must submit the implementation plan no later than 6 months before the date that the facility intends to comply with the emission averaging limits.
- (2) Contents. The owner or operator shall include the following information in the implementation plan or in the application for an operating permit for all emission sources to be included in an emissions average:
- (i) The identification of all emission sources (potlines or anode bake furnaces) in the average;

§ 63.847

- (ii) The assigned TF or POM emission limit for each averaging group of potlines or anode bake furnaces;
- (iii) The specific control technology or pollution prevention measure to be used for each emission source in the averaging group and the date of its installation or application. If the pollution prevention measure reduces or eliminates emissions from multiple sources, the owner or operator must identify each source;
- (iv) The test plan for the measurement of TF or POM emissions in accordance with the requirements in § 63.847(b):
- (v) The operating parameters to be monitored for each control system or device and a description of how the operating limits will be determined;
- (vi) If the owner or operator requests to monitor an alternative operating parameter pursuant to §63.848(1):
- (A) A description of the parameter(s) to be monitored and an explanation of the criteria used to select the parameter(s): and
- (B) A description of the methods and procedures that will be used to demonstrate that the parameter indicates proper operation of the control device; the frequency and content of monitoring, reporting, and recordkeeping requirements; and a demonstration, to the satisfaction of the applicable regulatory authority, that the proposed monitoring frequency is sufficient to represent control device operating conditions; and
- (vii) A demonstration that compliance with each of the applicable emission limit(s) will be achieved under representative operating conditions.
- (3) Approval criteria. Upon receipt, the regulatory authority shall review and approve or disapprove the plan or permit application according to the following criteria:
- (i) Whether the content of the plan includes all of the information specified in paragraph (d)(2) of this section;
- (ii) Whether the plan or permit application presents sufficient information to determine that compliance will be achieved and maintained.
- (4) Prohibitions. The applicable regulatory authority shall not approve an implementation plan or permit appli-

- cation containing any of the following provisions:
- (i) Any averaging between emissions of differing pollutants or between differing sources. Emission averaging shall not be allowed between TF and POM, and emission averaging shall not be allowed between potlines and bake furnaces:
- (ii) The inclusion of any emission source other than an existing potline or existing anode bake furnace or the inclusion of any potline or anode bake plant not subject to the same operating permit;
- (iii) The inclusion of any potline or anode bake furnace while it is shut down; or
- (iv) The inclusion of any periods of startup, shutdown, or malfunction, as described in the startup, shutdown, and malfunction plan required by §63.850(c), in the emission calculations.
- (5) Term. Following review, the applicable regulatory authority shall approve the plan or permit application, request changes, or request additional information. Once the applicable regulatory authority receives any additional information requested, the applicable regulatory authority shall approve or disapprove the plan or permit application within 120 days.
- (i) The applicable regulatory authority shall approve the plan for the term of the operating permit;
- (ii) To revise the plan prior to the end of the permit term, the owner or operator shall submit a request to the applicable regulatory authority; and
- (iii) The owner or operator may submit a request to the applicable regulatory authority to implement emission averaging after the applicable compliance date.
- (6) Operation. While operating under an approved implementation plan, the owner or operator shall monitor the operating parameters of each control system, keep records, and submit periodic reports as required for each source subject to this subpart.

§ 63.847 Compliance provisions.

(a) Compliance dates. The owner or operator of a primary aluminum plant must comply with the requirements of this subpart by:

2 to this subpart. You may not use an add-on control system or wet control device to meet the production-based compliance options.

(b) Compliance options for add-on control systems. You must use an emissions control system and demonstrate that the resulting emissions meet the compliance options and operating requirements in Tables 1B and 2 to this subpart. If you own or operate a reconstituted wood product press at a new or existing affected source or a reconstituted wood product board cooler at a new affected source, and you choose to comply with one of the concentrationbased compliance options for a control system outlet (presented as option numbers 2, 4, and 6 in Table 1B to this subpart), vou must have a capture device that either meets the definition of wood products enclosure in \63.2292 or achieves a capture efficiency of greater than or equal to 95 percent.

(c) Emissions averaging compliance option (for existing sources only). Using the procedures in paragraphs (c)(1) through (3) of this section, you must demonstrate that emissions included in the emissions average meet the compliance options and operating requirements. New sources may not use emissions averaging to comply with this subpart.

(1) Calculation of required and actual mass removal. Limit emissions of total HAP, as defined in §63.2292, to include acetaldehyde, acrolein, formaldehyde, methanol, phenol, and propional dehyde from your affected source to the standard specified by Equations 1, 2, and 3 of

$$RMR = 0.90 \times \left(\sum_{i=1}^{n} UCEP_i \times OH_i\right) (Eq. 1)$$

$$AMR = \left(\sum_{i=1}^{n} CD_{i} \times OCEP_{i} \times OH_{i}\right) (Eq. 2)$$

$$AMR \ge RMR$$
 (Eq. 3)

Where:

RMR = required mass removal of total HAP from all process units generating debits (i.e., all process units that are subject to the compliance options in Tables 1A and 1B to this subpart and that are either uncontrolled or under-controlled), pounds per semiannual period;

40 CFR Ch. I (7-1-12 Edition)

AMR = actual mass removal of total HAP from all process units generating credits (i.e., all process units that are controlled as part of the Emissions Averaging Plan including credits from debit-generating process units that are under-controlled), pounds per semiannual period;

UCEPi = mass of total HAP from an uncontrolled or under-controlled process unit (i) that generates debits, pounds per hour;

 OH_i = number of hours a process unit (i) is operated during the semiannual period, hours per 6-month period;

CDi = control system efficiency for the emission point (i) for total HAP, expressed as a fraction, and not to exceed 90 percent, unitless (Note: To calculate the control system efficiency of biological treatment units that do not meet the definition of biofilter in §63.2292, you must use 40 CFR part 63, appendix C, Determination of the Fraction Biodegraded (Fbio) in a Biological Treatment Unit.);

OCEPi = mass of total HAP from a process unit (i) that generates credits (including credits from debit-generating process units that are under-controlled), pounds per

0.90 = required control system efficiency of 90 percent multiplied, unitless.

(2) Requirements for debits and credits. You must calculate debits and credits as specified in paragraphs (c)(2)(i) through (vi) of this section.

(i) You must limit process units in the emissions average to those process units located at the existing affected source as defined in §63.2292.

(ii) You cannot use nonoperating process units to generate emissions averaging credits. You cannot use process units that are shut down to generate emissions averaging debits or

(iii) You may not include in your emissions average process units controlled to comply with a State, Tribal, or Federal rule other than this subpart.

(iv) You must use actual measurements of total HAP emissions from process units to calculate your required mass removal (RMR) and actual mass removal (AMR). The total HAP measurements must be obtained according to §63.2262(b) through (d), (g), and (h), using the methods specified in Table 4 to this subpart.

(v) Your initial demonstration that the credit-generating process units will be capable of generating enough credits to offset the debits from the debit-generating process units must be made

40 C.F.R. § 63.2240(c)

§ 63.2251

under representative operating conditions. After the compliance date, you must use actual operating data for all debit and credit calculations.

- (vi) Do not include emissions from the following time periods in your emissions averaging calculations:
- (A) Emissions during periods of startup, shutdown, and malfunction as described in the startup, shutdown, and malfunction plan (SSMP).
- (B) Emissions during periods of monitoring malfunctions, associated repairs, and required quality assurance or control activities or during periods of control device maintenance covered in your routine control device maintenance exemption. No credits may be assigned to credit-generating process units, and maximum debits must be assigned to debit-generating process units during these periods.
- (3) Operating requirements. You must meet the operating requirements in Table 2 to this subpart for each process unit or control device used in calculation of emissions averaging credits.

§63.2241 What are the work practice requirements and how must I meet

- (a) You must meet each work practice requirement in Table 3 to this subpart that applies to you.
- (b) As provided in §63.6(g), we, the EPA, may choose to grant you permission to use an alternative to the work practice requirements in this section.
- (c) If you have a dry rotary dryer, you may choose to designate your dry rotary dryer as a green rotary dryer and meet the more stringent compliance options and operating requirements in $\S63.2240$ for green rotary dryers instead of the work practices for dry rotary dryers. If you have a hardwood veneer dryer or veneer redryer, you may choose to designate your hardwood veneer dryer or veneer redryer as a softwood veneer dryer and meet the more stringent compliance options and operating requirements in §63.2240 for softwood veneer dryer heated zones instead of the work practices for hardwood veneer dryers or veneer redryers.

GENERAL COMPLIANCE REQUIREMENTS

§63.2250 What are the general require-

- (a) You must be in compliance with the compliance options, operating requirements, and the work practice requirements in this subpart at all times, except during periods of process unit or control device startup, shutdown, and malfunction; prior to process unit initial startup; and during the routine control device maintenance exemption specified in §63.2251. The compliance options, operating requirements, and work practice requirements do not apply during times when the process unit(s) subject to the compliance options, operating requirements, and work practice requirements are not operating, or during periods of startup, shutdown, and malfunction. Startup and shutdown periods must not exceed the minimum amount of time necessary for these events.
- (b) You must always operate and maintain your affected source, including air pollution control and monitoring equipment, according to the provisions in $\S63.6(e)(1)(i)$.
- (c) You must develop a written SSMP according to the provisions $\S63.6(e)(3)$
- (d) Shutoff of direct-fired burners resulting from partial and full production stoppages of direct-fired softwood veneer dryers or over-temperature events shall be deemed shutdowns and not malfunctions. Lighting or re-lighting any one or all gas burners in directfired softwood veneer dryers shall be deemed startups and not malfunctions.

[69 FR 46011, July 30, 2004, as amended at 71 FR 8372, Feb. 16, 2006; 71 FR 20463, Apr. 20,

§63.2251 What are the requirements for the routine control device maintenance exemption?

(a) You may request a routine control device maintenance exemption from the EPA Administrator for routine maintenance events such as control device bakeouts, washouts, media replacement, and replacement of corroded parts. Your request must justify the need for the routine maintenance on the control device and the time required to accomplish the maintenance

may use the precompliance report to request to monitor other parameters, and you must include a description of planned reporting and recordkeeping procedures and the basis for the selected monitoring frequencies and the methods that will be used.

Document #1421812

(d) If you transfer the wastewater offsite for enhanced biological treatment, you must obtain written certification from the offsite facility stating that the offsite facility will comply with the requirements of this subpart. The certifying entity may revoke the certification by providing 90 days notice. Upon expiration of the notice period, you may not transfer wastewater to that treatment facility.

§63.8025 What requirements apply to my transfer operations?

- (a) You must comply with each emission limit and work practice standard in Table 5 to this subpart that applies to your transfer operations, and you must meet all applicable requirements specified in §63.8000(b). For each control device used to comply with Table 5 to this subpart, you must comply with subpart SS of this part 63 as specified in §63.8000(c), except as specified in §63.8000(d) and paragraph (b) of this section.
- (b) If you have Group 1 transfer operations, as defined in §63.8105, then all transfer racks used for bulk loading coatings must meet the requirements for high throughput transfer racks in subpart SS of this part.

§63.8030 What requirements apply to my heat exchange systems?

- (a) You must comply with the requirements specified in Table 6 to this subpart that apply to your heat exchange systems, except as specified in paragraphs (b) through (e) of this section.
- (b) The phrase a chemical manufacturing process unit meeting the conditions of §63.100(b)(1) through (b)(3) of this section in §63.104(a) means the miscellaneous coating manufacturing operations defined in §63.7985(b) for the purposes of this subpart.
- (c) The reference to §63.100(c) in §63.104(a) does not apply for the purposes of this subpart.

40 CFR Ch. I (7-1-11 Edition)

- (d) The reference to §63.103(c)(1) in $\S63.104(f)(1)$ does not apply. For the purposes of this subpart, records must be retained as specified in §63.10(b)(1).
- (e) The reference to the periodic report required by §63.152(c) of subpart G of this part means the compliance report required by §63.8075(e) for the purposes of this subpart.

ALTERNATIVE MEANS OF COMPLIANCE

§63.8050 How do I comply with emissions averaging for stationary process vessels at existing sources

- (a) As an alternative to complying with the requirements in Table 1 to this subpart for each individual stationary process vessel, you may elect to comply with emissions averaging for stationary process vessels greater than or equal to 250 gallons (gal) at your existing affected source as specified in paragraphs (b) through (e) of this section.
- (b) General requirements. (1) A State may prohibit averaging of HAP emissions and require the owner or operator of an existing affected source to comply with the emission limits and work practice standards in Table 1 to this subpart.
- (2) All stationary process vessels in an emissions averaging group must be equipped with a tightly-fitting vented cover.
- (c) Initial compliance. To demonstrate initial compliance with the emissions averaging alternative, you must comply with the provisions in paragraphs (c)(1) through (4) of this section.
- (1) Estimate uncontrolled emissions from each affected stationary process vessel in pounds per batch using the procedures specified in §63.1257(d)(2), except as specified in paragraphs (c)(1)(i) and (ii) of this section. For the purposes of this section, uncontrolled emissions means the emissions from the vessel if it were equipped only with a tightly-fitting vented cover. must identify the range of typical operating parameters and perform the calculation using the values that result in the highest emissions, and you must document the operating parameters and resulting emissions calculations in the precompliance report.
- (i) When you are required to calculate uncontrolled emissions from

§ 63.8050

heating, you may not calculate emissions using Equation 13 of subpart GGG of this part 63.

(ii) The statement $\S63.1257(d)(2)(i)(B)$ that "the partial pressure of HAP shall be assumed to be 25 percent of the saturated value if the purge flow rate is greater than 100 scfm" does not apply. For the purposes of this subpart, multiply the HAP partial pressure in Equation 12 of 40 CFR part 63, subpart GGG by a HAP-specific saturation factor determined in accordance with Equations 1 through 3 of this section. Solve equation 1 of this section iteratively beginning with saturation factors (in the right-hand side of the equation) of 1.0 for each condensable compound. Stop iterating when the calculated saturation factors for all compounds are the same to two significant figures for subsequent iterations. Note that for multi-component emission streams, saturation factors must be calculated for all condensable compounds, not just the HAP.

$$S_{i} = \frac{K_{i}A}{K_{i}A + V + \sum_{i=1}^{n} S_{i}V_{i}^{sat}}$$
 Eq. 1

$$V_i^{\text{sat}} = \frac{VP_i}{\left(P_T - \sum_{i=1}^n P_i\right)}$$
 Eq. 2

$$K_i = K_o \left(\frac{M_o}{M_i}\right)^{1/3}$$
 Eq. 3

where:

 S_1 =saturation factor for individual condensable compounds in the emission stream

 $\begin{array}{l} P_i\text{=partial pressure of individual condensable}\\ \text{compounds in the emission stream calculated using Raoult's Law or other appropriate methods} \end{array}$

 P_T =pressure of the vessel vapor space A=surface area of liquid

V=purge flow rate as used in Equation 12 of 40 CFR part 63, subpart GGG

 V_{i}^{sat} =volumetric flowrate of condensable compounds in the emission stream

 $K_i = mass$ transfer coefficient of individual condensable compounds in the emission stream

 $K_o{=}mass\ transfer\ coefficient\ of\ a\ reference\ compound\ (e.g.,\ 0.83\ cm/s\ for\ water)$

 M_o =molecular weight of reference compound (e.g., 18.02 for water)

M_i=molecular weight of individual condensable compounds in the emission stream n=number of condensable compounds in the emission stream

- (2) Estimate controlled emissions in pounds per batch for each vessel as specified in paragraphs (c)(2)(i) through (iii) of this section.
- (i) Except as specified in paragraphs (c)(2)(ii) and (iii) of this section, estimate controlled emissions as if the vessel were controlled in compliance with entry 2.b.i. in Table 1 to this subpart.
- (ii) Estimate the controlled emissions using the control level achieved on November 15, 1990 if that value is greater than the applicable control level required by entry 2.b.i in Table 1 to this subpart.
- (iii) Estimate the controlled emissions using the control level required to comply with a State or Federal rule other than this subpart if that level is greater than the applicable control level required by entry 2.b.i in Table 1 to this subpart and the other rule was in effect before the date when you request approval to comply with emissions averaging.
- (3) Determine actual emissions in pounds per batch for each vessel in accordance with paragraph (c)(3)(i), (ii), or (iii) of this section, as applicable.
- (4) Provide rationale in the precompliance report for why the sum of the actual emissions will be less than the sum of emissions from the vessels if they had been controlled in accordance with Table 1 to this subpart. The approved actual emissions calculated according to paragraph (c)(3) of this section are emission limits that must be incorporated into your operating permit.
- (d) Continuous compliance. (1) Maintain a monthly log of the number of batches produced that can be correlated with the emissions estimates per batch developed in accordance with paragraph (c) of this section.
- (2) Sum the actual emissions for all of the process vessels in the emissions averaging group every three months, with the first 3-month period beginning on the compliance date, and compare

40 CFR Ch. I (7-1-11 Edition)

the resulting total with the total emissions for the vessels calculated in accordance with paragraph (c)(2) of this section. Compliance is demonstrated if the sum of the actual emissions is less than the emissions estimated in accordance with paragraph (c)(2) of this section.

Document #1421812

- (3) For control devices, establish operating limits and monitor as specified in §63.8000.
- (e) Recordkeeping and reporting. Comply with §§ 63.8070, 63.8075, and 63.8080.

[68 FR 69185, Dec. 11, 2003, as amended at 70 FR 25682, May 13, 2005]

§63.8055 How do I comply with a weight percent HAP limit in coating products?

- (a) As an alternative to complying with the requirements in Table 1 to this subpart for each individual stationary process vessel at an existing source, you may elect to comply with a 5 weight percent HAP limit for process vessels at your affected source that are used to manufacture coatings with a HAP content of less than 0.05 kg per kg product as specified in paragraph (b) of this section.
- (b) You may only comply with the alternative during the production of coatings that contain less than 5 weight percent HAP, as determined using any of the procedures specified in paragraphs (b)(1) through (4) of this section.
- (1) Method 311 (appendix A to 40 CFR part 63).
- (2) Method 24 (appendix A to 40 CFR part 60). You may use Method 24 to determine the mass fraction of volatile matter and use that value as a substitute for the mass fraction of HAP.
- (3) You may use an alternative test method for determining mass fraction of HAP if you obtain prior approval by the Administrator. You must follow the procedure in §63.7(f) to submit an alternative test method for approval.
- (4) You may rely on formulation data from raw material suppliers if it represents each organic HAP that is present at 0.1 percent by mass or more for OSHA-defined carcinogens, as specified in 29 CFR 1910.1200(d)(4), and at 1.0 percent by mass or more for other compounds. If the HAP weight percent estimated based on formulation data con-

flicts with the results of a test conducted according to paragraphs (b)(1) through (3) of this section, then there is a rebuttal presumption that the test results are accurate unless, after consultation, you demonstrate to the satisfaction of the permitting authority that the test results are not accurate and that the formulation data are more appropriate.

[68 FR 69185, Dec. 11, 2003, as amended at 70 FR 25682, May 13, 2005; 70 FR 75927, Dec. 21,

NOTIFICATION, REPORTS, AND RECORDS

§63.8070 What notifications must I submit and when?

- (a) You must submit all of the notifications in $\S 63.6(h)(4)$ and (5), 63.7(b)and (c), 63.8(e), (f)(4) and (6), 63.9(b) through (h) that apply to you by the dates specified.
- (b) Initial notification. (1) As specified in §63.9(b)(2), if you have an existing affected source on December 11, 2003, you must submit an initial notification not later than 120 calendar days after December 11, 2003.
- (2) As specified in $\S63.9(b)(3)$, if you start up your new affected source on or after December 11, 2003, you must submit an initial notification not later than 120 calendar days after you become subject to this subpart.
- (c) Notification of performance test. If you are required to conduct a performance test, you must submit a notification of intent to conduct a performance test at least 60 calendar days before the performance test is scheduled to begin as required in §63.7(b)(1). For any performance test required as part of the initial compliance procedures for process vessels in Table 1 to this subpart, you must also submit the test plan required by §63.7(c) and the emission profile with the notification of the performance test.

§63.8075 What reports must I submit and when?

- (a) You must submit each report in Table 9 to this subpart that applies to
- (b) Unless the Administrator has approved a different schedule for submission of reports under §63.10(a), you must submit each report as specified in

§63.9980

40 CFR Ch. I (7-1-12 Edition)

Citation	Subject	Applies to Subpart TTTTT	Explanation
63.6(a)-(g)	Compliance with Standards and Maintenance Requirements.	Yes.	
63.6(h)	Determining Compliance with Opacity and Visible Emission Standards.	No.	
63.6(i)–(j)	Extension of Compliance and Presidential Compliance Exemption.	Yes.	
63.7(a)(1)–(2)	Applicability and Performance Test Dates.	No	Subpart TTTTT specifies per- formance test applicability and dates.
63.7(a)(3), (b)–(h)	Performance Testing Requirements.	Yes.	
63.8 except for (a)(4),(c)(4), and (f)(6).	Monitoring Requirements	Yes.	
63.8(a)(4)	Additional Monitoring Require- ments for Control Devices in §63.11.	No	Subpart TTTTT does not require flares.
63.8(c)(4)	Continuous Monitoring System Requirements.	No	Subpart TTTTT specifies requirements for operation of CMS.
63.8(f)(6)	Relative Accuracy Test Alternative (RATA).	No	Subpart TTTTT does not require continuous emission monitoring systems.
63.9	Notification Requirements	Yes.	9 =,===
63.9(g)(5)	Data Reduction	No	Subpart TTTTT specifies data reduction requirements.
63.10 except for (b)(2)(xiii) and (c)(7)-(8).	Recordkeeping and Reporting Requirements.	Yes.	'
63.10(b)(2)(xiii)	Continuous Monitoring System (CMS) Records for RATA Alternative.	No	Subpart TTTTT does not require continuous emission monitoring systems.
63.10(c)(7)–(8)	Records of Excess Emissions and Parameter Monitoring Accedences for CMS.	No	Subpart TTTTT specifies record-keeping requirements.
63.11	Control Device Requirements	No	Subpart TTTTT does not require flares.
63.12	State Authority and Delegations	Yes.	
63.13–63.15	Addresses, Incorporation by Reference, Availability of Information.	Yes.	

Document #1421812

Subpart UUUUU—National Emission Standards for Hazardous Air Pollutants: Coal- and Oil-Fired Electric Utility Steam Generating Units

SOURCE: 77 FR 9464, Feb. 16, 2012, unless otherwise noted.

WHAT THIS SUBPART COVERS

§63.9980 What is the purpose of this subpart?

This subpart establishes national emission limitations and work practice standards for hazardous air pollutants (HAP) emitted from coal- and oil-fired electric utility steam generating units (EGUs) as defined in §63.10042 of this subpart. This subpart also establishes requirements to demonstrate initial

and continuous compliance with the emission limitations.

§ 63.9981 Am I subject to this subpart?

You are subject to this subpart if you own or operate a coal-fired EGU or an oil-fired EGU as defined in §63.10042 of this subpart.

$\S 63.9982$ What is the affected source of this subpart?

(a) This subpart applies to each individual or group of two or more new, reconstructed, and existing affected source(s) as described in paragraphs (a)(1) and (2) of this section within a contiguous area and under common control.

(1) The affected source of this subpart is the collection of all existing coal- or oil-fired EGUs, as defined in §63.10042, within a subcategory.

§ 63.9984

- (2) The affected source of this subpart is each new or reconstructed coal- or oil-fired EGU as defined in §63.10042.
- (b) An EGU is new if you commence construction of the coal- or oil-fired EGU after May 3, 2011, and you meet the applicability criteria at the time you commence construction.
- (c) An EGU is reconstructed if you meet the reconstruction criteria as defined in §63.2, you commence reconstruction after May 3, 2011, and you meet the applicability criteria at the time you commence reconstruction.
- (d) An EGU is existing if it is not new or reconstructed. An existing electric steam generating unit that meets the applicability requirements after the effective date of this final rule due to a change in process (e.g., fuel or utilization) is considered to be an existing source under this subpart.

[77 FR 9464, Feb. 16, 2012, as amended at 77 FR 23402, Apr. 19, 2012]

§63.9983 Are any EGUs not subject to this subpart?

The types of electric steam generating units listed in paragraphs (a) through (d) of this section are not subject to this subpart.

- (a) Any unit designated as a stationary combustion turbine, other than an integrated gasification combined cycle (IGCC) unit, covered by 40 CFR part 63, subpart YYYY.
- (b) Any electric utility steam generating unit that is not a coal- or oil-fired EGU and combusts natural gas for more than 10.0 percent of the average annual heat input during any 3 calendar years or for more than 15.0 percent of the annual heat input during any calendar year.
- (c) Any electric utility steam generating unit that has the capability of combusting more than 25 MW of coal or oil but did not fire coal or oil for more than 10.0 percent of the average annual heat input during any 3 calendar years or for more than 15.0 percent of the annual heat input during any calendar year. Heat input during any calendar year. Heat input means heat derived from combustion of fuel in an EGU and does not include the heat derived from preheated combustion air, recirculated flue gases or exhaust gases from other sources (such as stationary gas tur-

bines, internal combustion engines, and industrial boilers).

(d) Any electric steam generating unit combusting solid waste is a solid waste incineration unit subject to standards established under sections 129 and 111 of the Clean Air Act.

§63.9984 When do I have to comply with this subpart?

- (a) If you have a new or reconstructed EGU, you must comply with this subpart by April 16, 2012 or upon startup of your EGU, whichever is later, and as further provided for in §63.10005(g).
- (b) If you have an existing EGU, you must comply with this subpart no later than April 16, 2015.
- (c) You must meet the notification requirements in §63.10030 according to the schedule in §63.10030 and in subpart A of this part. Some of the notifications must be submitted before you are required to comply with the emission limits and work practice standards in this subpart.
- (d) An electric steam generating unit that does not meet the definition of an EGU subject to this subpart on April 16, 2012 for new sources or April 16, 2015 for existing sources must comply with the applicable existing source provisions of this subpart on the date such unit meets the definition of an EGU subject to this subpart.
- (e) If you own or operate an electric steam generating unit that is exempted from this subpart under §63.9983(d), if the manner of operating the unit changes such that the combustion of waste is discontinued and the unit becomes a coal-fired or oil-fired EGU (as defined in §63.10042), you must be in compliance with this subpart on April 16, 2015 or on the effective date of the switch from waste combustion to coal or oil combustion, whichever is later.
- (f) You must demonstrate that compliance has been achieved, by conducting the required performance tests and other activities, no later than 180 days after the applicable date in paragraph (a), (b), (c), (d), or (e) of this section.

§ 63.9985 What is a new EGU?

- (a) A new EGU is an EGU that meets any of the criteria specified in paragraph (a)(1) through (a)(2) of this section.
- (1) An EGU that commenced construction after May 3, 2011.
- (2) An EGU that commenced reconstruction after May 3, 2011.
 - (b) [Reserved]

[77 FR 9464, Feb. 16, 2012, as amended at 77 FR 23402, Apr. 19, 2012]

EMISSION LIMITATIONS AND WORK PRACTICE STANDARDS

§63.9990 What are the subcategories of EGUs?

- (a) Coal-fired EGUs are subcategorized as defined in paragraphs (a)(1) through (a)(2) of this section and as defined in §63.10042.
- (1) EGUs designed for coal with a heating value greater than or equal to 8,300 Btu/lb, and
- (2) EGUs designed for low rank virgin
- (b) Oil-fired EGUs are subcategorized as noted in paragraphs (b)(1) through (b)(4) of this section and as defined in 863.10042
 - (1) Continental liquid oil-fired EGUs
- (2) Non-continental liquid oil-fired EGUs.
- (3) Limited-use liquid oil-fired EGUs, and
- (4) EGUs designed to burn solid oilderived fuel.
- (c) IGCC units combusting either gasified coal or gasified solid oil-derived fuel. For purposes of compliance, monitoring, recordkeeping, and reporting requirements in this subpart, IGCC units are subject in the same manner as coal-fired units and solid oil-derived fuel-fired units, unless otherwise indicated.

§63.9991 What emission limitations, work practice standards, and operating limits must I meet?

- (a) You must meet the requirements in paragraphs (a)(1) and (2) of this section. You must meet these requirements at all times.
- (1) You must meet each emission limit and work practice standard in Table 1 through 3 to this subpart that applies to your EGU, for each EGU at

40 CFR Ch. I (7-1-12 Edition)

your source, except as provided under 863.10009.

- (2) You must meet each operating limit in Table 4 to this subpart that applies to your EGU.
- (b) As provided in §63.6(g), the Administrator may approve use of an alternative to the work practice standards in this section.
- (c) You may use the alternate SO₂ limit in Tables 1 and 2 to this subpart only if your EGU:
- (1) Has a system using wet or dry flue gas desulfurization technology and SO₂ continuous emissions monitoring system (CEMS) installed on the unit; and
- (2) At all times, you operate the wet or dry flue gas desulfurization technology installed on the unit consistent with §63.10000(b).

[77 FR 9464, Feb. 16, 2012, as amended at 77 FR 23402, Apr. 19, 2012]

GENERAL COMPLIANCE REQUIREMENTS

§63.10000 What are my general requirements for complying with this subpart?

- (a) You must be in compliance with the emission limits and operating limits in this subpart. These limits apply to you at all times except during periods of startup and shutdown; however, for coal-fired, liquid oil-fired, or solid oil-derived fuel-fired EGUs, you are required to meet the work practice requirements in Table 3 to this subpart during periods of startup or shutdown.
- (b) At all times you must operate and maintain any affected source, including associated air pollution control equipment and monitoring equipment, in a manner consistent with safety and good air pollution control practices for minimizing emissions. Determination of whether such operation and maintenance procedures are being used will be based on information available to the EPA Administrator which may include, but is not limited to, monitoring results, review of operation and maintenance procedures, review of operation and maintenance records, and inspection of the source.
- (c)(1) For coal-fired units, IGCC units, and solid oil-derived fuel-fired

40 C.F.R. § 63.10000(b)

40 CFR Ch. I (7-1-12 Edition)

(j) You must report the results of performance tests and performance tune-ups within 60 days after the completion of the performance tests and performance tune-ups. The reports for all subsequent performance tests must include all applicable information required in §63.10031.

[77 FR 9464, Feb. 16, 2012, as amended at 77 FR 23403, Apr. 19, 2012]

§ 63.10007 What methods and other procedures must I use for the performance tests?

- (a) Except as otherwise provided in this section, you must conduct all required performance tests according to §63.7(d), (e), (f), and (h). You must also develop a site-specific test plan according to the requirements in §63.7(c).
- (1) If you use CEMS (Hg, HCl, SO_2 , or other) to determine compliance with a 30-boiler operating day rolling average emission limit, you must collect data for all nonexempt unit operating conditions (see §63.10011(g) and Table 3 to this subpart).

40 C.F.R. § 63.100007(a)(2) (2) If you conduct performance testing with test methods in lieu of continuous monitoring, operate the unit at maximum normal operating load conditions during each periodic (e.g., quarterly) performance test. Maximum normal operating load will be generally between 90 and 110 percent of design capacity but should be representative of site specific normal operations during each test run.

(3) For establishing operating limits with particulate matter continuous parametric monitoring system (PM CPMS) to demonstrate compliance with a PM or non Hg metals emissions limit, operate the unit at maximum normal operating load conditions during the performance test period. Maximum normal operating load will be generally between 90 and 110 percent of design capacity but should be representative of site specific normal operations during each test run.

(b) You must conduct each performance test (including traditional 3-run stack tests, 30-boiler operating day tests based on CEMS data (or sorbent trap monitoring system data), and 30-boiler operating day Hg emission tests for LEE qualification) according to the

requirements in Table 5 to this subpart.

- (c) If you choose to comply with the filterable PM emission limit and demonstrate continuous performance using a PM CPMS for an applicable emission limit as provided for in §63.10000(c), you must also establish an operating limit according to §63.10011(b) and Tables 4 and 6 to this subpart. Should you desire to have operating limits that correspond to loads other than maximum normal operating load, you must conduct testing at those other loads to determine the additional operating limits.
- (d) Except for a 30-boiler operating day performance test based on CEMS (or sorbent trap monitoring system) data, where the concept of test runs does not apply, you must conduct a minimum of three separate test runs for each performance test, as specified in §63.7(e)(3). Each test run must comply with the minimum applicable sampling time or volume specified in Table 1 or 2 to this subpart. Sections 63.10005(d) and (h), respectively, provide special instructions for conducting performance tests based on CEMS or sorbent trap monitoring systems, and for conducting emission tests for LEE qualification.
- (e) To use the results of performance testing to determine compliance with the applicable emission limits in Table 1 or 2 to this subpart, proceed as follows:
- (1) Except for a 30-boiler operating day performance test based on CEMS (or sorbent trap monitoring system) data, if measurement results for any pollutant are reported as below the method detection level (e.g., laboratory analytical results for one or more sample components are below the method defined analytical detection level), you must use the method detection level as the measured emissions level for that pollutant in calculating compliance. The measured result for a multiple component analysis (e.g., analytical values for multiple Method 29 fractions both for individual HAP metals and for total HAP metals) may include a combination of method detection level data and analytical data reported above the method detection level.

Document #1421812

§ 63.10009

- (2) If the limits are expressed in lb/ MMBtu or lb/TBtu, you must use the Ffactor methodology and equations in sections 12.2 and 12.3 of EPA Method 19 in appendix A-7 to part 60 of this chapter. In cases where an appropriate Ffactor is not listed in Table 19-2 of Method 19, you may use F-factors from Table 1 in section 3.3.5 of appendix F to part 75 of this chapter, or F-factors derived using the procedures in section 3.3.6 of appendix to part 75 of this chapter. Use the following factors to convert the pollutant concentrations measured during the initial performance tests to units of lb/scf, for use in the applicable Method 19 equations:
 - (i) Multiply SO₂ ppm by 1.66×10^{-7} ;
 - (ii) Multiply HCl ppm by 9.43×10^{-8} ; (iii) Multiply HF ppm by 5.18×10^{-8} ;
- (iv) Multiply HAP metals concentrations (mg/dscm) by 6.24×10^{-8} ; and
- (v) Multiply Hg concentrations (μg / scm) by 6.24×10^{-11} .
- (3) To determine compliance with emission limits expressed in lb/MWh or lb/GWh, you must first calculate the pollutant mass emission rate during the performance test, in units of lb/h. For Hg, if a CEMS or sorbent trap monitoring system is used, use Equation A-2 or A-3 in appendix A to this subpart (as applicable). In all other cases, use an equation that has the general form of Equation A-2 or A-3, replacing the value of K with 1.66 \times 10⁻⁷ lb/scf-ppm for SO_2 , 9.43×10^{-8} lb/scf-ppm for HCl (if an HCl CEMS is used), 5.18×10^{-8} lb/scfppm for HF (if an HF CEMS is used), or 6.24×10^{-8} lb-scm/mg-scf for HAP metals and for HCl and HF (when performance stack testing is used), and defining Ch as the average SO2, HCl, or HF concentration in ppm, or the average HAP metals concentration in mg/dscm. This calculation requires stack gas volumetric flow rate (scfh) and (in some cases) moisture content data (see $\S 63.10005(h)(3)$ and 63.10010). Then, if the applicable emission limit is in units of lb/GWh, use Equation A-4 in appendix A to this subpart to calculate the pollutant emission rate in lb/GWh. In this calculation, define (M)_h as the calculated pollutant mass emission rate for the performance test (lb/h), and define (MW)h as the average electrical load during the performance test (megawatts). If the applicable emission

- limit is in lb/MWh rather than lb/GWh, omit the 10³ term from Equation A-4 to determine the pollutant emission rate
- (f) Upon request, you shall make available to the EPA Administrator such records as may be necessary to determine whether the performance tests have been done according to the requirements of this section.

[77 FR 9464, Feb. 16, 2012, as amended at 77 FR 23403, Apr. 19, 2012]

§63.10008 [Reserved]

§63.10009 May I use emissions averaging to comply with this subpart?

- (a) General eligibility. (1) You may use emissions averaging as described in paragraph (a)(2) of this section as an alternative to meeting the requirements of §63.9991 for filterable PM, SO₂, HF, HCl, non-Hg HAP metals, or Hg on an EGU-specific basis if:
- (i) You have more than one existing EGU in the same subcategory located at one or more contiguous properties, belonging to a single major industrial grouping, which are under common control of the same person (or persons under common control); and
- (ii) You use CEMS (or sorbent trap monitoring systems for determining Hg emissions) or quarterly emissions testing for demonstrating compliance.
- (2) You may demonstrate compliance by emissions averaging among the existing EGUs in the same subcategory, if your averaged Hg emissions for EGUs in the "unit designed for coal ≥ 8,300 Btu/lb'' subcategory are equal to or less than 1.0 lb/TBtu or 1.1E-2 lb/GWh or if your averaged emissions of individual, other pollutants from other subcategories of such EGUs are equal to or less than the applicable emissions limit in Table 2, according to the procedures in this section. Note that except for Hg emissions from EGUs in the "unit designed for coal ≥ 8,300 Btu/lb" subcategory, the averaging time for emissions averaging for pollutants is 30 days (rolling daily) using data from CEMS or a combination of data from CEMS and manual performance testing. The averaging time for emissions averaging for Hg from EGUs in the "unit designed for coal ≥ 8,300 Btu/lb" subcategory is 90 days (rolling daily)

using data from CEMS, sorbent trap monitoring, or a combination of monitoring data and data from manual performance testing. For the purposes of this paragraph, 30- (or 90-day) group boiler operating days is defined as a period during which at least one unit in the emissions averaging group has operated 30 (or 90) days. You must calculate the weighted average emissions rate for the group in accordance with the procedures in this paragraph using the data from all units in the group including any that operate fewer than 30 (or 90) days during the preceding 30 (or 90) group boiler days.

40 CFR Ch. I (7-1-12 Edition)

- (i) You may choose to have your EGU emissions averaging group meet either the heat input basis (MMBtu or TBtu, as appropriate for the pollutant) or gross electrical output basis (MWh or GWh, as appropriate for the pollutant).
- (ii) You may not mix bases within your EGU emissions averaging group.
- (iii) You may use emissions averaging for affected units in different subcategories if the units vent to the atmosphere through a common stack (see paragraph (m) of this section).
- (b) Equations. Use the following equations when performing calculations for your EGU emissions averaging group:
 - (1) Group eligibility equations.

$$WAERm = \frac{\left[\sum_{i=1}^{p} \left[\sum_{i=1}^{n} \left(Herm_{i} \times Rmm_{i}\right)\right]_{p}\right] + \sum_{i=1}^{m} \left(Ter_{i} \times Rmt_{i}\right)}{\left[\sum_{i=1}^{p} \left[\sum_{i=1}^{n} Rmm_{i}\right]_{p}\right] + \sum_{i=1}^{m} Rmt_{i}}$$
(Eq. 1a)

Where:

WAERm = Weighted average emissions rate maximum in terms of lb/heat input or lb/ gross electrical output,

Herm_i = Hourly emissions rate (e.g., 1b/ MMBtu, lb/MWh) from CEMS or sorbent trap monitoring for hour i,

Rmm; = Maximum rated heat input or gross electrical output of unit i in terms of heat input or gross electrical output.

p = number of EGUs in emissions averaging group that rely on CEMS,

n = number of hourly rates collected over 30group boiler operating days,

Ter_i = Emissions rate from most recent test of unit i in terms of lb/heat input or lb/ gross electrical output,

 $Rmt_i = Maximum rated heat input or gross$ electrical output of unit i in terms of 1b/ heat input or lb/gross electrical output,

m = number of EGUs in emissions averaging group that rely on emissions testing.

$$WAERm = \frac{\left[\sum_{i=1}^{p} \left[\sum_{i=1}^{n} \left(Herm_{i} \times Smm_{i} \times Cfm_{i}\right)\right]_{p}\right] + \sum_{i=1}^{m} \left(Ter_{i} \times Smt_{i} \times Cft_{i}\right)}{\left[\sum_{i=1}^{p} \left[\sum_{i=1}^{n} Smm_{i} \times Cfm_{i}\right]_{p}\right] + \sum_{i=1}^{m} Smt_{i} \times Cft_{i}}$$
(Eq. 1b)

Where:

variables with similar names share the descriptions for Equation 1a.

 $Smm_i = maximum steam generation in units$ of pounds from unit i that uses CEMS or sorbent trap monitoring,

Cfm; = conversion factor, calculated from the most recent emissions test results, in units of heat input per pound of steam generated or gross electrical output per pound of steam generated, from unit i that uses CEMS or sorbent trap monitoring,

Smt_i = maximum steam generation in units of pounds from unit i that uses emissions testing, and

Cft_i = conversion factor, calculated from the most recent emissions test results, in units of heat input per pound of steam generated or gross electrical output per pound of steam generated, from unit i that uses emissions testing.

(2) Weighted 30-day rolling average emissions rate equations for pollutants other than Hg. Use equation 2a or 2b to calculate the 30-day rolling average emissions daily.

§ 63.10009

$$WAER = \frac{\sum_{i=1}^{p} \left[\sum_{i=1}^{n} (Her_{i} \times Rm_{i}) \right]_{p} + \sum_{i=1}^{m} (Ter_{i} \times Rt_{i})}{\sum_{i=1}^{p} \left[\sum_{i=1}^{n} (Rm_{i}) \right]_{p} + \sum_{i=1}^{m} Rt_{i}}$$
 (Eq. 2a)

Where:

 Her_i = hourly emission rate (e.g., lb/MMBtu, 1b/MWh) from unit i's CEMS for the preceding 30-group boiler operating days,

Rm_i = hourly heat input or gross electrical output from unit i for the preceding 30group boiler operating days,

p = number of EGUs in emissions averaging group that rely on CEMS or sorbent trap monitoring.

n = number of hourly rates collected over 30group boiler operating days,

Ter_i = Emissions rate from most recent emissions test of unit i in terms of lb/heat input or lb/gross electrical output,

 Rt_i = Maximum rated heat input or gross electrical output of unit i in terms of 1b/ heat input or lb/gross electrical output, and

m = number of EGUs in emissions averaging group that rely on emissions testing.

$$WAER = \frac{\sum_{i=1}^{p} \left[\sum_{i=1}^{n} (Her_i \times Sm_i \times Cfm_i)\right]_p + \sum_{i=1}^{m} (Ter_i \times St_i \times Cft_i)}{\sum_{i=1}^{p} \left[\sum_{i=1}^{n} (Sm_i \times Cfm_i)\right]_p + \sum_{i=1}^{m} St_i \times Cft_i} (Eq. 2b)$$

Where:

variables with similar names share the descriptions for Equation 2a,

Sm_i = steam generation in units of pounds from unit i that uses CEMS for the preceding 30-group boiler operating days,

Cfm; = conversion factor, calculated from the most recent compliance test results, in units of heat input per pound of steam generated or gross electrical output per pound of steam generated, from unit i that uses CEMS from the preceding 30group boiler operating days,

= steam generation in units of pounds from unit i that uses emissions testing, and

 Cft_i = conversion factor, calculated from the most recent compliance test results, in units of heat input per pound of steam generated or gross electrical output per pound of steam generated, from unit i that uses emissions testing.

(3) Weighted 90-boiler operating day rolling average emissions rate equations for Hg emissions from EGUs in the "unit designed for coal ≥ 8,300 Btu/ 1b" subcategory. Use equation 3a or 3b to calculate the 90-day rolling average emissions daily.

$$WAER = \frac{\sum_{i=1}^{p} \left[\sum_{i=1}^{n} (Her_{i} \times Rm_{i}) \right]_{p} + \sum_{i=1}^{m} (Ter_{i} \times Rt_{i})}{\sum_{i=1}^{p} \left[\sum_{i=1}^{n} (Rm_{i}) \right]_{p} + \sum_{i=1}^{m} Rt_{i}}$$
 (Eq. 3a)

Where:

Heri = hourly emission rate from unit i's CEMS or Hg sorbent trap monitoring for the preceding 90-group boiler operating days,

 Rm_i = hourly heat input or gross electrical output from unit i for the preceding 90group boiler operating days,

p = number of EGUs in emissions averaging group that rely on CEMS,

n = number of hourly rates collected over the 90-group boiler operating days.

Ter; = Emissions rate from most recent emissions test of unit i in terms of lb/heat input or lb/gross electrical output.

Rt: = Maximum rated heat input or gross electrical output of unit i in terms of 1b/ heat input or lb/gross electrical output,

m = number of EGUs in emissions averaging group that rely on emissions testing.

40 CFR Ch. I (7-1-12 Edition)

$$WAER = \frac{\sum_{i=1}^{p} \left[\sum_{i=1}^{n} (Her_i \times Sm_i \times Cfm_i) \right]_p + \sum_{i=1}^{m} (Ter_i \times St_i \times Cft_i)}{\sum_{i=1}^{p} \left[\sum_{i=1}^{n} (Sm_i \times Cfm_i) \right]_p + \sum_{i=1}^{m} St_i \times Cft_i} \quad (Eq.3b)$$

Where:

variables with similar names share the descriptions for Equation 2a,

Sm_i = steam generation in units of pounds from unit i that uses CEMS or a Hg sorbent trap monitoring for the preceding 90-group boiler operating days,

Cfm_i = conversion factor, calculated from the most recent compliance test results, in units of heat input per pound of steam generated or gross electrical output per pound of steam generated, from unit i that uses CEMS or sorbent trap monitoring from the preceding 90-group boiler operating days,

St_i = steam generation in units of pounds from unit i that uses emissions testing.

Cft_i = conversion factor, calculated from the most recent emissions test results, in units of heat input per pound of steam generated or gross electrical output per pound of steam generated, from unit i that uses emissions testing.

(c) Separate stack requirements. For a group of two or more existing EGUs in the same subcategory that each vent to a separate stack, you may average filterable PM, SO₂, HF, HCl, non-Hg HAP metals, or Hg emissions to demonstrate compliance with the limits in Table 2 to this subpart if you satisfy the requirements in paragraphs (d) through (j) of this section.

(d) For each existing EGU in the averaging group:

(1) The emissions rate achieved during the initial performance test for the HAP being averaged must not exceed the emissions level that was being achieved 180 days after April 16, 2015, or the date on which emissions testing done to support your emissions averaging plan is complete (if the Administrator does not require submission and approval of your emissions averaging plan), or the date that you begin emissions averaging, whichever is earlier;

(2) The control technology employed during the initial performance test must not be less than the design efficiency of the emissions control technology employed 180 days after April 16, 2015 or the date that you begin

emissions averaging, whichever is ear-

(e) The weighted-average emissions rate from the existing EGUs participating in the emissions averaging option must be in compliance with the limits in Table 2 to this subpart at all times following the compliance date specified 180 days after April 16, 2015, or the date on which you complete the emissions measurements used to support your emissions averaging plan (if the Administrator does not require submission and approval of your emissions averaging plan), or the date that you begin emissions averaging, whichever is earlier.

(f) Emissions averaging group eligibility demonstration. You must demonstrate the ability for the EGUs included in the emissions averaging group to demonstrate initial compliance according to paragraph (f)(1) or (2)of this section using the maximum normal operating load of each EGU and the results of the initial performance tests. For this demonstration and prior to submitting your emissions averaging plan, if requested, you must conduct required emissions monitoring for 30 days of boiler operation and any required manual performance testing to calculate an initial weighted average emissions rate in accordance with this section. Should the Administrator require approval, you must submit your proposed emissions averaging plan and supporting data at least 120 days before April 16, 2015. If the Administrator requires approval of your plan, you may not begin using emissions averaging until the Administrator approves your

(1) You must use Equation 1a in paragraph (b) of this section to demonstrate that the maximum weighted average emissions rates of filterable PM, HF, SO₂, HCl, non-Hg HAP metals, or Hg emissions from the existing units participating in the emissions averaging option do not exceed the emissions limits in Table 2 to this subpart.

§ 63.10009

- (2) If you are not capable of monitoring heat input or gross electrical output, and the EGU generates steam for purposes other than generating electricity, you may use Equation 1b of this section as an alternative to using Equation 1a of this section to demonstrate that the maximum weighted average emissions rates of filterable PM, HF, SO₂, HCl, non-Hg HAP metals, or Hg emissions from the existing units participating in the emissions averaging group do not exceed the emission limits in Table 2 to this subpart.
- (g) You must determine the weighted average emissions rate in units of the applicable emissions limit on a 30 day rolling average (90 day rolling average for Hg) basis according to paragraphs (g)(1) through (2) of this section. The first averaging period begins on 30 (or 90 for Hg) days after February 16, 2015 or the date that you begin emissions averaging, whichever is earlier.
- (1) You must use Equation 2a or 3a of paragraph (b) of this section to calculate the weighted average emissions rate using the actual heat input or gross electrical output for each existing unit participating in the emissions averaging option.
- (2) If you are not capable of monitoring heat input or gross electrical output, you may use Equation 2b or 3b of paragraph (b) of this section as an alternative to using Equation 2a of paragraph (b) of this section to calculate the average weighted emission rate using the actual steam generation from the units participating in the emissions averaging option.
- (h) CEMS (or sorbent trap monitoring) use. If an EGU in your emissions averaging group uses CEMS (or a sorbent trap monitor for Hg emissions) to demonstrate compliance, you must use those data to determine the 30 (or 90) group boiler operating day rolling average emissions rate.
- (i) Emissions testing. If you use manual emissions testing to demonstrate compliance for one or more EGUs in your emissions averaging group, you must use the results from the most recent performance test to determine the 30 (or 90) day rolling average. You may use CEMS or sorbent trap data in combination with data from the most recent manual performance test in calcu-

- lating the 30 (or 90) group boiler operating day rolling average emissions
- (j) Emissions averaging plan. You must develop an implementation plan for emissions averaging according to the following procedures and requirements in paragraphs (j)(1) and (2) of this section.
- (1) You must include the information contained in paragraphs (j)(1)(i)through (v) of this section in your implementation plan for all the emissions units included in an emissions aver-
- (i) The identification of all existing EGUs in the emissions averaging group, including for each either the applicable HAP emission level or the control technology installed as of 180 days after February 16, 2015, or the date on which you complete the emissions measurements used to support your emissions averaging plan (if the Administrator does not require submission and approval of your emissions averaging plan), or the date that you begin emissions averaging, whichever is earlier; and the date on which you are requesting emissions averaging to commence;
- (ii) The process weighting parameter (heat input, gross electrical output, or steam generated) that will be monitored for each averaging group;
- (iii) The specific control technology or pollution prevention measure to be used for each emission EGU in the averaging group and the date of its installation or application. If the pollution prevention measure reduces or eliminates emissions from multiple EGUs, you must identify each EGU;
- (iv) The means of measurement (e.g., CEMS, sorbent trap monitoring, manual performance test) of filterable PM, SO₂, HF, HCl, individual or total non-Hg HAP metals, or Hg emissions in accordance with the requirements in §63.10007 and to be used in the emissions averaging calculations; and
- (v) A demonstration that emissions averaging can produce compliance with each of the applicable emission limit(s) in accordance with paragraph (b)(1) of this section.
- (2) If the Administrator requests you to submit the plan for review and approval, you must submit a complete

40 CFR Ch. I (7-1-12 Edition)

implementation plan at least 120 days before April 16, 2015. If the Administrator requests you to submit the plan for review and approval, you must receive approval before initiating emissions averaging.

Document #1421812

- (i) The Administrator shall use following criteria in reviewing and approving or disapproving the plan:
- (A) Whether the content of the plan includes all of the information specified in paragraph (j)(1) of this section;
- (B) Whether the plan presents information sufficient to determine that compliance will be achieved and maintained.
- (ii) The Administrator shall not approve an emissions averaging implementation plan containing any of the following provisions:
- (A) Any averaging between emissions of different pollutants or between units located at different facilities; or
- (B) The inclusion of any emissions unit other than an existing unit in the same subcategory.
- (k) Common stack requirements. For a group of two or more existing affected units, each of which vents through a single common stack, you may average emissions to demonstrate compliance with the limits in Table 2 to this subpart if you satisfy the requirements in paragraph (1) or (m) of this section.
- (1) For a group of two or more existing units in the same subcategory and which vent through a common emissions control system to a common stack that does not receive emissions from units in other subcategories or categories, you may treat such averaging group as a single existing unit for purposes of this subpart and comply with the requirements of this subpart as if the group were a single unit.
- (m) For all other groups of units subject to paragraph (k) of this section, you may elect to conduct manual performance tests according to procedures specified in §63.10007 in the common stack. If emissions from affected units included in the emissions averaging and from other units not included in the emissions averaging (e.g., in a different subcategory) or other nonaffected units all vent to the common stack, you must shut down the units not included in the emissions aver-

aging and the nonaffected units or vent their emissions to a different stack during the performance test. Alternatively, you may conduct a performance test of the combined emissions in the common stack with all units operating and show that the combined emissions meet the most stringent emissions limit. You may also use a CEMS or sorbent trap monitoring to apply this latter alternative to demonstrate that the combined emissions comply with the most stringent emissions limit on a continuous basis.

(n) Combination requirements. The common stack of a group of two or more existing EGUs in the same subcategory subject to paragraph (k) of this section may be treated as a single stack for purposes of paragraph (c) of this section and included in an emissions averaging group subject to paragraph (c) of this section.

[77 FR 9464, Feb. 16, 2012, as amended at 77 FR 23403, Apr. 19, 2012]

§63.10010 What are my monitoring, installation, operation, and maintenance requirements?

- (a) Flue gases from the affected units under this subpart exhaust to the atmosphere through a variety of different configurations, including but not limited to individual stacks, a common stack configuration or a main stack plus a bypass stack. For the CEMS, PM CPMS, and sorbent trap monitoring systems used to provide data under this subpart, the continuous monitoring system installation requirements for these exhaust configurations are as follows:
- (1) Single unit-single stack configurations. For an affected unit that exhausts to the atmosphere through a single, dedicated stack, you shall either install the required CEMS, PM CPMS, and sorbent trap monitoring systems in the stack or at a location in the ductwork downstream of all emissions control devices, where the pollutant and diluents concentrations are representative of the emissions that exit to the atmosphere.
- (2) Unit utilizing common stack with other affected unit(s). When an affected unit utilizes a common stack with one or more other affected units, but no non-affected units, you shall either:

40 CFR Ch. I (7-1-12 Edition)

part 60 of this chapter, using Method 5 at Appendix A–3 to part 60 of this chapter and ensuring that the front half filter temperature shall be $160^{\circ}\pm14$ °C $(320^{\circ}\pm25$ °F). The reportable measurement output from the PM CEMS must be expressed in units of the applicable emissions limit (e.g., lb/MMBtu, lb/MWh).

- (2) Operate and maintain your PM CEMS according to the procedures and requirements in Procedure 2—Quality Assurance Requirements for Particulate Matter Continuous Emission Monitoring Systems at Stationary Sources in Appendix F to part 60 of this chapter.
- (i) You must conduct the relative response audit (RRA) for your PM CEMS at least once annually.
- (ii) You must conduct the relative correlation audit (RCA) for your PM CEMS at least once every 3 years.
- (3) Collect PM CEMS hourly average output data for all boiler operating hours except as indicated in paragraph (i) of this section.
- (4) Calculate the arithmetic 30-boiler operating day rolling average of all of the hourly average PM CEMS output data collected during all nonexempt boiler operating hours.
- (5) You must collect data using the PM CEMS at all times the process unit is operating and at the intervals specified in paragraph (a) of this section, except for periods of monitoring system malfunctions, repairs associated with monitoring system malfunctions, and required monitoring system quality assurance or quality control activities.
- (i) You must use all the data collected during all boiler operating hours in assessing the compliance with your operating limit except:
- (A) Any data collected during monitoring system malfunctions, repairs associated with monitoring system malfunctions, or required monitoring system quality assurance or control activities conducted during monitoring system malfunctions in calculations and report any such periods in your annual deviation report:
- (B) Any data collected during periods when the monitoring system is out of control as specified in your site-specific monitoring plan, repairs associated with periods when the monitoring

system is out of control, or required monitoring system quality assurance or control activities conducted during out of control periods in calculations used to report emissions or operating levels and report any such periods in your annual deviation report;

- (C) Any data recorded during periods of startup or shutdown.
- (ii) You must record and make available upon request results of PM CEMS system performance audits, dates and duration of periods when the PM CEMS is out of control to completion of the corrective actions necessary to return the PM CEMS to operation consistent with your site-specific monitoring plan.
- (j) You may choose to comply with the metal HAP emissions limits using CEMS approved in accordance with §63.7(f) as an alternative to the performance test method specified in this rule. If approved to use a HAP metals CEMS, the compliance limit will be expressed as a 30-boiler operating day rolling average of the numerical emissions limit value applicable for your unit in tables 1 or 2. If approved, you may choose to install, certify, operate, and maintain a HAP metals CEMS and record the output of the HAP metals CEMS as specified in paragraphs (j)(1) through (5) of this section.
- (1)(i) Install and certify your HAP metals CEMS according to the procedures and requirements in you approved site specific test plan as required in §63.7(e). The reportable measurement output from the HAP metals CEMS must be expressed in units of the applicable emissions limit (e.g., 1b/MMBtu, 1b/MWh) and in the form of a 30-boiler operating day rolling average.
- (ii) Operate and maintain your HAP metals CEMS according to the procedures and criteria in your site specific performance evaluation and quality control program plan required in §63.8(d).
- (2) Collect HAP metals CEMS hourly average output data for all boiler operating hours except as indicated in section (j)(4) of this section.
- (3) Calculate the arithmetic 30-boiler operating day rolling average of all of the hourly average HAP metals CEMS output data collected during all non-exempt boiler operating hours data.

40 C.F.R. § 63.10010(j)

§63.10011

- (4) You must collect data using the HAP metals CEMS at all times the process unit is operating and at the intervals specified in paragraph (a) of this section, except for periods of monitoring system malfunctions, repairs associated with monitoring system malfunctions, and required monitoring system quality assurance or quality control activities.
- (i) You must use all the data collected during all boiler operating hours in assessing the compliance with your emission limit except:
- (A) Any data collected during monitoring system malfunctions, repairs associated with monitoring system malfunctions, or required monitoring system quality assurance or control activities conducted during monitoring system malfunctions in calculations and report any such periods in your annual deviation report;
- (B) Any data collected during periods when the monitoring system is out of control as specified in your site-specific monitoring plan, repairs associated with periods when the monitoring system is out of control, or required monitoring system quality assurance or control activities conducted during out of control periods in calculations used to report emissions or operating levels and report any such periods in your annual deviation report;
- (C) Any data recorded during periods of startup or shutdown.
- (ii) You must record and make available upon request results of HAP metals CEMS system performance audits, dates and duration of periods when the HAP metals CEMS is out of control to completion of the corrective actions necessary to return the HAP metals CEMS to operation consistent with your site-specific performance evaluation and quality control program plan.
- (k) If you demonstrate compliance with the HCl and HF emission limits for a liquid oil-fired EGU by conducting quarterly testing, you must also develop a site-specific monitoring plan as provided for in §63.10000(c)(2)(iii) and Table 7 to this subpart.

[77 FR 9464, Feb. 16, 2012, as amended at 77 FR 23404, Apr. 19, 2012]

- §63.10011 How do I demonstrate initial compliance with the emissions limits and work practice standards?
- (a) You must demonstrate initial compliance with each emissions limit that applies to you by conducting performance testing.
- (b) If you are subject to an operating limit in Table 4 to this subpart, you demonstrate initial compliance with HAP metals or filterable PM emission limit(s) through performance stack tests and you elect to use a PM CPMS to demonstrate continuous performance, or if, for a liquid oil-fired unit, and you use quarterly stack testing for HCl and HF plus site-specific parameter monitoring to demonstrate continuous performance, you must also establish a site-specific operating limit, in accordance with Table 4 to this subpart, §63.10007, and Table 6 to this subpart. You may use only the parametric data recorded during successful performance tests (i.e., tests that demonstrate compliance with the applicable emissions limits) to establish an operating limit.
- (c)(1) If you use CEMS or sorbent trap monitoring systems to measure a HAP (e.g., Hg or HCl) directly, the first 30-boiler operating day (or, if alternate emissions averaging is used for Hg, the 90-boiler operating day) rolling average emission rate obtained with certified CEMS after the applicable date in §63.9984 (or, if applicable, prior to that date, as described in §63.10005(b)(2)), expressed in units of the standard, is the initial performance test. Initial compliance is demonstrated if the results of the performance test meet the applicable emission limit in Table 1 or 2 to this subpart.
- (2) For a unit that uses a CEMS to measure SO_2 or PM emissions for initial compliance, the first 30 boiler operating day average emission rate obtained with certified CEMS after the applicable date in §63.9984 (or, if applicable, prior to that date, as described in §63.10005(b)(2)), expressed in units of the standard, is the initial performance test. Initial compliance is demonstrated if the results of the performance test meet the applicable SO_2 or filterable PM emission limit in Table 1 or 2 to this subpart.

(c) You must operate and maintain your process and control equipment such that the 30 operating day average PM CPMS output does not exceed the operating limit determined in paragraphs (a) and (b) of this section.

NOTIFICATION, REPORTS, AND RECORDS

§63.10030 What notifications must I submit and when?

- (a) You must submit all of the notifications in §§63.7(b) and (c), 63.8 (e), (f)(4) and (6), and 63.9 (b) through (h) that apply to you by the dates specified.
- (b) As specified in §63.9(b)(2), if you startup your affected source before April 16, 2012, you must submit an Initial Notification not later than 120 days after April 16, 2012.
- (c) As specified in §63.9(b)(4) and (b)(5), if you startup your new or reconstructed affected source on or after April 16, 2012, you must submit an Initial Notification not later than 15 days after the actual date of startup of the affected source.
- (d) When you are required to conduct a performance test, you must submit a Notification of Intent to conduct a performance test at least 30 days before the performance test is scheduled to
- (e) When you are required to conduct an initial compliance demonstration as specified in §63.10011(a), you must submit a Notification of Compliance Status according to §63.9(h)(2)(ii). The Notification of Compliance Status report must contain all the information specified in paragraphs (e)(1) through (7), as applicable.
- (1) A description of the affected source(s) including identification of which subcategory the source is in, the design capacity of the source, a description of the add-on controls used on the source, description of the fuel(s) burned, including whether the fuel(s) were determined by you or EPA through a petition process to be a nonwaste under 40 CFR 241.3, whether the fuel(s) were processed from discarded non-hazardous secondary materials within the meaning of 40 CFR 241.3, and justification for the selection of fuel(s) burned during the performance test.

40 CFR Ch. I (7-1-12 Edition)

- (2) Summary of the results of all performance tests and fuel analyses and calculations conducted to demonstrate initial compliance including all established operating limits.
- (3) Identification of whether you plan to demonstrate compliance with each applicable emission limit through performance testing; fuel moisture analyses; performance testing with operating limits (e.g., use of PM CPMS); CEMS: or a sorbent trap monitoring system.
- (4) Identification of whether you plan to demonstrate compliance by emissions averaging.
- (5) A signed certification that you have met all applicable emission limits and work practice standards.
- (6) If you had a deviation from any emission limit, work practice standard, or operating limit, you must also submit a brief description of the deviation, the duration of the deviation, emissions point identification, and the cause of the deviation in the Notification of Compliance Status report.
- (7) In addition to the information required in §63.9(h)(2), your notification of compliance status must include the following:
- (i) A summary of the results of the annual performance tests and documentation of any operating limits that were reestablished during this test, if applicable. If you are conducting stack tests once every 3 years consistent with §63.10006(b), the date of the last three stack tests, a comparison of the emission level you achieved in the last three stack tests to the 50 percent emission limit threshold required in §63.10006(i), and a statement as to whether there have been any operational changes since the last stack test that could increase emissions.
- (ii) Certifications of compliance, as applicable, and must be signed by a responsible official stating:
- (A) "This EGU complies with the requirements in §63.10021(a) to demonstrate continuous compliance." and
- (B) "No secondary materials that are solid waste were combusted in any affected unit.

[77 FR 9464, Feb. 16, 2012, as amended at 77 FR 23404, Apr. 19, 2012]

40 C.F.R. § 63.10030(e)(7)(i)

§ 63.10031

§63.10031 What reports must I submit and when?

- (a) You must submit each report in Table 8 to this subpart that applies to you. If you are required to (or elect to) continuously monitor Hg and/or HCl and/or HF emissions, you must also submit the electronic reports required under appendix A and/or appendix B to the subpart, at the specified frequency.
- (b) Unless the Administrator has approved a different schedule for submission of reports under §63.10(a), you must submit each report by the date in Table 8 to this subpart and according to the requirements in paragraphs (b)(1) through (5) of this section.
- (1) The first compliance report must cover the period beginning on the compliance date that is specified for your affected source in §63.9984 and ending on June 30 or December 31, whichever date is the first date that occurs at least 180 days after the compliance date that is specified for your source in §63.9984.
- (2) The first compliance report must be postmarked or submitted electronically no later than July 31 or January 31, whichever date is the first date following the end of the first calendar half after the compliance date that is specified for your source in §63.9984.
- (3) Each subsequent compliance report must cover the semiannual reporting period from January 1 through June 30 or the semiannual reporting period from July 1 through December 31.
- (4) Each subsequent compliance report must be postmarked or submitted electronically no later than July 31 or January 31, whichever date is the first date following the end of the semi-annual reporting period.
- (5) For each affected source that is subject to permitting regulations pursuant to part 70 or part 71 of this chapter, and if the permitting authority has established dates for submitting semiannual reports pursuant to 40 CFR 70.6(a)(3)(iii)(A) or 40 CFR 71.6(a)(3)(iii)(A), you may submit the first and subsequent compliance reports according to the dates the permitting authority has established instead of according to the dates in paragraphs (b)(1) through (4) of this section.

- (c) The compliance report must contain the information required in paragraphs (c)(1) through (4) of this section.
- (1) The information required by the summary report located in 63.10(e)(3)(vi).
- (2) The total fuel use by each affected source subject to an emission limit, for each calendar month within the semi-annual reporting period, including, but not limited to, a description of the fuel, whether the fuel has received a non-waste determination by EPA or your basis for concluding that the fuel is not a waste, and the total fuel usage amount with units of measure.
- (3) Indicate whether you burned new types of fuel during the reporting period. If you did burn new types of fuel you must include the date of the performance test where that fuel was in use.
- (4) Include the date of the most recent tune-up for each unit subject to the requirement to conduct a performance tune-up according to §63.10021(e). Include the date of the most recent burner inspection if it was not done every 36 (or 48) months and was delayed until the next scheduled unit shutdown.
- (d) For each excess emissions occurring at an affected source where you are using a CMS to comply with that emission limit or operating limit, you must include the information required in §63.10(e)(3)(v) in the compliance report specified in section (c).
- (e) Each affected source that has obtained a Title V operating permit pursuant to part 70 or part 71 of this chapter must report all deviations as defined in this subpart in the semiannual monitoring report required by 40 CFR 70.6(a)(3)(iii)(A)40 orCFR 71.6(a)(3)(iii)(A). If an affected source submits a compliance report pursuant to Table 8 to this subpart along with, or as part of, the semiannual monitoring report required by 40 CFR 70.6(a)(3)(iii)(A)or71.6(a)(3)(iii)(A), and the compliance report includes all required information concerning deviations from any emission limit, operating limit, or work practice requirement in this subpart, submission of the compliance report satisfies any obligation to report the same deviations in the semiannual

40 C.F.R. § 63.10031(c)(3)

§63.10031

40 CFR Ch. I (7-1-12 Edition)

monitoring report. Submission of a compliance report does not otherwise affect any obligation the affected source may have to report deviations from permit requirements to the permit authority.

(f) As of January 1, 2012, and within 60 days after the date of completing each performance test, you must submit the results of the performance tests required by this subpart to EPA's WebFIRE database by using the Compliance and Emissions Data Reporting Interface (CEDRI) that is accessed through EPA's Central Data Exchange (CDX) (www.epa.gov/cdx). Performance test data must be submitted in the file format generated through use of EPA's Electronic Reporting Tool (ERT) (see http://www.epa.gov/ttn/chief/ert/

index.html). Only data collected using those test methods on the ERT Web site are subject to this requirement for submitting reports electronically to WebFIRE. Owners or operators who claim that some of the information being submitted for performance tests is confidential business information (CBI) must submit a complete ERT file including information claimed to be CBI on a compact disk or other commonly used electronic storage media (including, but not limited to, flash drives) to EPA. The electronic media must be clearly marked as CBI and mailed to U.S. EPA/OAPQS/CORE CBI Office, Attention: WebFIRE Administrator, MD C404-02, 4930 Old Page Rd., Durham, NC 27703. The same ERT file with the CBI omitted must be submitted to EPA via CDX as described earlier in this paragraph. At the discretion of the delegated authority, you must also submit these reports, including the confidential business information, to the delegated authority in the format specified by the delegated au-

(1) Within 60 days after the date of completing each CEMS (SO₂, PM, HCl, HF, and Hg) performance evaluation test, as defined in §63.2 and required by this subpart, you must submit the relative accuracy test audit (RATA) data (or, for PM CEMS, RCA and RRA data) required by this subpart to EPA's WebFIRE database by using the Compliance and Emissions Data Reporting Interface (CEDRI) that is accessed

through EPA's Central Data Exchange (CDX) (www.epa.gov/cdx). The RATA data shall be submitted in the file format generated through use of EPA's Electronic Reporting Tool (ERT) (http://www.epa.gov/ttn/chief/ert/index.html). Only RATA data compounds listed on the ERT Web site are

index.html). Only RATA data compounds listed on the ERT Web site are subject to this requirement. Owners or operators who claim that some of the information being submitted for RATAs is confidential business information (CBI) shall submit a complete ERT file including information claimed to be CBI on a compact disk or other commonly used electronic storage media (including, but not limited to, flash drives) by registered letter to EPA and the same ERT file with the CBI omitted to EPA via CDX as described earlier in this paragraph. The compact disk or other commonly used electronic storage media shall be clearly marked as CBI and mailed to U.S. EPA/OAPQS/CORE CBI Office, Attention: WebFIRE Administrator, MD C404-02, 4930 Old Page Rd., Durham, NC 27703. At the discretion of the delegated authority, owners or operators shall also submit these RATAs to the delegated authority in the format specified by the delegated authority. Owners or operators shall submit calibration error testing, drift checks, and other information required in the performance evaluation as described in §63.2 and as required in this chapter.

(2) For a PM CEMS, PM CPMS, or approved alternative monitoring using a HAP metals CEMS, within 60 days after the reporting periods ending on March 31st, June 30th, September 30th, and December 31st, you must submit quarterly reports to EPA's WebFIRE database by using the Compliance and Emissions Data Reporting Interface (CEDRI) that is accessed through EPA's Central Data Exchange (CDX) (www.epa.gov/cdx). You must use the appropriate electronic reporting form in CEDRI or provide an alternate electronic file consistent with EPA's reporting form output format. For each reporting period, the quarterly reports must include all of the calculated 30boiler operating day rolling average values derived from the CEMS and PM CPMS.

Environmental Protection Agency

(3) Reports for an SO₂ CEMS, a Hg CEMS or sorbent trap monitoring system, an HCl or HF CEMS, and any supporting monitors for such systems (such as a diluent or moisture monitor) shall be submitted using the ECMPS Client Tool, as provided for in Appendices A and B to this subpart and §63.10021(f).

(4) Submit the compliance reports required under paragraphs (c) and (d) of this section and the notification of compliance status required under §63.10030(e) to EPA's WebFIRE database by using the Compliance and Emissions Data Reporting Interface (CEDRI) that is accessed through EPA's Central Data Exchange (CDX) (www.epa.gov/cdx). You must use the appropriate electronic reporting form in CEDRI or provide an alternate electronic file consistent with EPA's reporting form output format.

(5) All reports required by this subpart not subject to the requirements in paragraphs (f)(1) through (4) of this section must be sent to the Administrator at the appropriate address listed in §63.13. If acceptable to both the Administrator and the owner or operator of a source, these reports may be submitted on electronic media. The Administrator retains the right to require submittal of reports subject to paragraphs (f)(1), (2), and (3) of this section in paper format.

(g) If you had a malfunction during the reporting period, the compliance report must include the number, duration, and a brief description for each type of malfunction which occurred during the reporting period and which caused or may have caused any applicable emission limitation to be exceed-

[77 FR 9464, Feb. 16, 2012, as amended at 77 FR 23404, Apr. 19, 20121

§63.10032 What records must I keep?

- (a) You must keep records according to paragraphs (a)(1) and (2) of this section. If you are required to (or elect to) continuously monitor Hg and/or HCl and/or HF emissions, you must also keep the records required under appendix A and/or appendix B to this sub-
- (1) A copy of each notification and report that you submitted to comply

with this subpart, including all documentation supporting any Initial Notification or Notification of Compliance Status or semiannual compliance report that you submitted, according to the requirements in $\S63.10(b)(2)(xiv)$.

- (2) Records of performance stack tests, fuel analyses, or other compliance demonstrations and performance evaluations. as required §63.10(b)(2)(viii).
- (b) For each CEMS and CPMS, you must keep records according to paragraphs (b)(1) through (4) of this section.
- Records (1) described $\S63.10(b)(2)(vi)$ through (xi).
- (2) Previous (i.e., superseded) versions of the performance evaluation plan as required in $\S63.8(d)(3)$.
- (3) Request for alternatives to relative accuracy test for CEMS as required in $\S63.8(f)(6)(i)$.
- (4) Records of the date and time that each deviation started and stopped, and whether the deviation occurred during a period of startup, shutdown, or malfunction or during another period.
- (c) You must keep the records required in Table 7 to this subpart including records of all monitoring data and calculated averages for applicable PM CPMS operating limits to show continuous compliance with each emission limit and operating limit that applies to you.
- (d) For each EGU subject to an emission limit, you must also keep the records in paragraphs (d)(1) through (3) of this section.
- (1) You must keep records of monthly fuel use by each EGU, including the type(s) of fuel and amount(s) used.
- (2) If you combust non-hazardous secondary materials that have been determined not to be solid waste pursuant to 40 CFR 241.3(b)(1), you must keep a record which documents how the secondary material meets each of the legitimacy criteria. If you combust a fuel that has been processed from a discarded non-hazardous secondary material pursuant to 40 CFR 241.3(b)(2), you must keep records as to how the operations that produced the fuel satisfies the definition of processing in 40 CFR 241.2. If the fuel received a non-waste determination pursuant to the petition process submitted under 40 CFR 241.3(c), you must keep a record which

40 C.F.R. § 63.10031(g)

40 CFR Ch. I (7-1-12 Edition)

40 C.F.R. § 63.10032(d)(3)

40 C.F.R. § 63.10032(g)

40 C.F.R. § 63.10032(h)

documents how the fuel satisfies the requirements of the petition process.

- (3) For an EGU that qualifies as an LEE under §63.10005(h), you must keep annual records that document that your emissions in the previous stack test(s) continue to qualify the unit for LEE status for an applicable pollutant, and document that there was no change in source operations including fuel composition and operation of air pollution control equipment would cause emissions of the pollutant to increase within the past year.
- (e) If you elect to average emissions consistent with §63.10009, you must additionally keep a copy of the emissions averaging implementation plan required in §63.10009(g), all calculations $required \quad under \quad \S \, 63.10009, \quad including$ daily records of heat input or steam generation, as applicable, and monitoring records consistent with § 63.10022.
- (f) You must keep records of the occurrence and duration of each startup and/or shutdown.
- (g) You must keep records of the occurrence and duration of each malfunction of an operation (i.e., process equipment) or the air pollution control and monitoring equipment.
- (h) You must keep records of actions taken during periods of malfunction to minimize emissions in accordance with §63.10000(b), including corrective actions to restore malfunctioning process and air pollution control and monitoring equipment to its normal or usual manner of operation.
- (i) You must keep records of the type(s) and amount(s) of fuel used during each startup or shutdown.
- (j) If you elect to establish that an EGU qualifies as a limited-use liquid oil-fired EGU, you must keep records of the type(s) and amount(s) of fuel use in each calendar quarter to document that the capacity factor limitation for that subcategory is met.

§63.10033 In what form and how long must I keep my records?

- (a) Your records must be in a form suitable and readily available for expeditious review. according §63.10(b)(1).
- (b) As specified in §63.10(b)(1), you must keep each record for 5 years fol-

lowing the date of each occurrence, measurement, maintenance, corrective action, report, or record.

(c) You must keep each record on site for at least 2 years after the date of each occurrence, measurement, maintenance, corrective action, report, or record, according to §63.10(b)(1). You can keep the records off site for the remaining 3 years.

OTHER REQUIREMENTS AND INFORMATION

§63.10040 What parts of the General Provisions apply to me?

Table 9 to this subpart shows which parts of the General Provisions in §§ 63.1 through 63.15 apply to you.

§63.10041 Who implements and enforces this subpart?

- (a) This subpart can be implemented and enforced by U.S. EPA, or a delegated authority such as your state, local, or tribal agency. If the EPA Administrator has delegated authority to your state, local, or tribal agency, then that agency (as well as the U.S. EPA) has the authority to implement and enforce this subpart. You should contact your EPA Regional Office to find out if this subpart is delegated to your state, local, or tribal agency.
- (b) In delegating implementation and enforcement authority of this subpart to a state, local, or tribal agency under 40 CFR part 63, subpart E, the authorities listed in paragraphs (b)(1) through (4) of this section are retained by the EPA Administrator and are not transferred to the state, local, or tribal agency; moreover, the U.S. EPA retains oversight of this subpart and can take enforcement actions, as appropriate, with respect to any failure by any person to comply with any provision of this subpart.
- (1) Approval of alternatives to the non-opacity emission limits and work practice standards in §63.9991(a) and (b) under $\S63.6(g)$.
- (2) Approval of major change to test methods in Table 5 to this subpart under §63.7(e)(2)(ii) and (f) and as defined in §63.90, approval of minor and intermediate changes to monitoring performance specifications/procedures in Table 5 where the monitoring serves



SUBCHAPTER C—AIR PROGRAMS (CONTINUED)

PART 64—COMPLIANCE ASSURANCE MONITORING

Sec.

- 64.1 Definitions.
- 64.2 Applicability.
- 64.3 Monitoring design criteria.
- 64.4 Submittal requirements.
- 64.5 Deadlines for submittals.
- 64.6 Approval of monitoring.
- 64.7 Operation of approved monitoring.
- 64.8 Quality improvement plan (QIP) requirements.
- 64.9 Reporting and recordkeeping requirements.
- 64.10 Savings provisions.

AUTHORITY: 42 U.S.C. 7414 and 7661-7661f.

SOURCE: 62 FR 54940, Oct. 22, 1997, unless otherwise noted.

§ 64.1 Definitions.

The following definitions apply to this part. Except as specifically provided in this section, terms used in this part retain the meaning accorded them under the applicable provisions of the Act.

Act means the Clean Air Act, as amended by Pub. L. 101–549, 42 U.S.C. 7401 et sea

Applicable requirement shall have the same meaning as provided under part 70 of this chapter.

Capture system means the equipment (including but not limited to hoods, ducts, fans, and booths) used to contain, capture and transport a pollutant to a control device.

Continuous compliance determination method means a method, specified by the applicable standard or an applicable permit condition, which:

- (1) Is used to determine compliance with an emission limitation or standard on a continuous basis, consistent with the averaging period established for the emission limitation or standard; and
- (2) Provides data either in units of the standard or correlated directly with the compliance limit.

Control device means equipment, other than inherent process equipment, that is used to destroy or remove air pollutant(s) prior to discharge to the atmosphere. The types of equipment

that may commonly be used as control devices include, but are not limited to, fabric filters, mechanical collectors, electrostatic precipitators, inertial separators, afterburners, thermal or catalytic incinerators, adsorption devices (such as carbon beds), condensers, scrubbers (such as wet collection and gas absorption devices), selective catalytic or non-catalytic reduction systems, flue gas recirculation systems, spray dryers, spray towers, mist eliminators, acid plants, sulfur recovery plants, injection systems (such as water, steam, ammonia, sorbent or limestone injection), and combustion devices independent of the particular process being conducted at an emissions unit (e.g., the destruction of emissions achieved by venting process emission streams to flares, boilers or process heaters). For purposes of this part, a control device does not include passive control measures that act to prevent pollutants from forming, such as the use of seals, lids, or roofs to prevent the release of pollutants, use of low-polluting fuel or feedstocks, or the use of combustion or other process design features or characteristics. If an applicable requirement establishes that particular equipment which otherwise meets this definition of a control device does not constitute a control device as applied to a particular pollutant-specific emissions unit, then that definition shall be binding for purposes of this part.

Data means the results of any type of monitoring or method, including the results of instrumental or non-instrumental monitoring, emission calculations, manual sampling procedures, recordkeeping procedures, or any other form of information collection procedure used in connection with any type of monitoring or method.

Emission limitation or standard means any applicable requirement that constitutes an emission limitation, emission standard, standard of performance or means of emission limitation as defined under the Act. An emission limitation or standard may be expressed in terms of the pollutant, expressed either

§ 64.1

USCA Case #12-1100

40 CFR Ch. I (7-1-12 Edition)

as a specific quantity, rate or concentration of emissions (e.g., pounds of SO₂ per hour, pounds of SO₂ per million British thermal units of fuel input, kilograms of VOC per liter of applied coating solids, or parts per million by volume of SO₂) or as the relationship of uncontrolled to controlled emissions (e.g., percentage capture and destruction efficiency of VOC or percentage reduction of SO₂). An emission limitation or standard may also be expressed either as a work practice, process or control device parameter, or other form of specific design, equipment, operational, or operation and maintenance requirement. For purposes of this part, an emission limitation or standard shall not include general operation requirements that an owner or operator may be required to meet, such as requirements to obtain a permit, to operate and maintain sources in accordance with good air pollution control practices, to develop and maintain a malfunction abatement plan, to keep records, submit reports, or conduct monitoring.

Emissions unit shall have the same meaning as provided under part 70 of this chapter.

Exceedance shall mean a condition that is detected by monitoring that provides data in terms of an emission limitation or standard and that indicates that emissions (or opacity) are greater than the applicable emission limitation or standard (or less than the applicable standard in the case of a percent reduction requirement) consistent with any averaging period specified for averaging the results of the monitoring.

Excursion shall mean a departure from an indicator range established for monitoring under this part, consistent with any averaging period specified for averaging the results of the monitoring.

process equipment means Inherent equipment that is necessary for the proper or safe functioning of the process, or material recovery equipment that the owner or operator documents is installed and operated primarily for purposes other than compliance with air pollution regulations. Equipment that must be operated at an efficiency higher than that achieved during normal process operations in order to comply with the applicable emission limitation or standard is not inherent process equipment. For the purposes of this part, inherent process equipment is not considered a control device.

Filed: 02/21/2013

Major source shall have the same meaning as provided under part 70 or 71 of this chapter.

Monitoring means any form of collecting data on a routine basis to determine or otherwise assess compliance with emission limitations or standards. Recordkeeping may be considered monitoring where such records are used to determine or assess compliance with an emission limitation or standard (such as records of raw material content and usage, or records documenting compliance with work practice requirements). The conduct of compliance method tests, such as the procedures in appendix A to part 60 of this chapter, on a routine periodic basis may be considered monitoring (or as a supplement to other monitoring), provided that requirements to conduct such tests on a one-time basis or at such times as a regulatory authority may require on a non-regular basis are not considered monitoring requirements for purposes of this paragraph. Monitoring may include one or more than one of the following data collection techniques. where appropriate for a particular circumstance:

- (1) Continuous emission or opacity monitoring systems.
- (2) Continuous process, capture system, control device or other relevant parameter monitoring systems or procedures, including a predictive emission monitoring system.
- (3) Emission estimation and calculation procedures (e.g., mass balance or stoichiometric calculations).
- (4) Maintenance and analysis of records of fuel or raw materials usage.
- (5) Recording results of a program or protocol to conduct specific operation and maintenance procedures.
- (6) Verification of emissions, process parameters, capture system parameters, or control device parameters using portable or in situ measurement devices.
- (7) Visible emission observations.
- (8) Any other form of measuring, recording, or verifying on a routine basis

§ 64.2

emissions, process parameters, capture system parameters, control device parameters or other factors relevant to assessing compliance with emission limitations or standards.

Owner or operator means any person who owns, leases, operates, controls or supervises a stationary source subject to this part.

Part 70 or 71 permit shall have the same meaning as provided under part 70 or 71 of this chapter, provided that it shall also refer to a permit issued, renewed, amended, revised, or modified under any federal permit program promulgated under title V of the Act.

Part 70 or 71 permit application shall mean an application (including any supplement to a previously submitted application) that is submitted by the owner or operator in order to obtain a part 70 or 71 permit.

Permitting authority shall have the same meaning as provided under part 70 or 71 of this chapter.

Pollutant-specific emissions unit means an emissions unit considered separately with respect to each regulated air pollutant.

Potential to emit shall have the same meaning as provided under part 70 or 71 of this chapter, provided that it shall be applied with respect to an "emissions unit" as defined under this part in addition to a "stationary source" as provided under part 70 or 71 of this chapter.

Predictive emission monitoring system (PEMS) means a system that uses process and other parameters as inputs to a computer program or other data reduction system to produce values in terms of the applicable emission limitation or standard.

Regulated air pollutant shall have the same meaning as provided under part 70 or 71 of this chapter.

§ 64.2 Applicability.

(a) General applicability. Except for backup utility units that are exempt under paragraph (b)(2) of this section, the requirements of this part shall apply to a pollutant-specific emissions unit at a major source that is required to obtain a part 70 or 71 permit if the unit satisfies all of the following criteria:

- (1) The unit is subject to an emission limitation or standard for the applicable regulated air pollutant (or a surrogate thereof), other than an emission limitation or standard that is exempt under paragraph (b)(1) of this section;
- (2) The unit uses a control device to achieve compliance with any such emission limitation or standard; and
- (3) The unit has potential pre-control device emissions of the applicable regulated air pollutant that are equal to or greater than 100 percent of the amount, in tons per year, required for a source to be classified as a major source. For purposes of this paragraph, "potential pre-control device emissions" shall have the same meaning as "potential to emit," as defined in §64.1, except that emission reductions achieved by the applicable control device shall not be taken into account.
- (b) Exemptions—(1) Exempt emission limitations or standards. The requirements of this part shall not apply to any of the following emission limitations or standards:
- (i) Emission limitations or standards proposed by the Administrator after November 15, 1990 pursuant to section 111 or 112 of the Act.
- (ii) Stratospheric ozone protection requirements under title VI of the Act.
- (iii) Acid Rain Program requirements pursuant to sections 404, 405, 406, 407(a), 407(b), or 410 of the Act.
- (iv) Emission limitations or standards or other applicable requirements that apply solely under an emissions trading program approved or promulgated by the Administrator under the Act that allows for trading emissions within a source or between sources.
- (v) An emissions cap that meets the requirements specified in §70.4(b)(12) or §71.6(a)(13)(iii) of this chapter.
- (vi) Emission limitations or standards for which a part 70 or 71 permit specifies a continuous compliance determination method, as defined in §64.1. The exemption provided in this paragraph (b)(1)(vi) shall not apply if the applicable compliance method includes an assumed control device emission reduction factor that could be affected by the actual operation and maintenance of the control device

§ 64.3

40 CFR Ch. I (7-1-12 Edition)

(such as a surface coating line controlled by an incinerator for which continuous compliance is determined by calculating emissions on the basis of coating records and an assumed control device efficiency factor based on an initial performance test; in this example, this part would apply to the control device and capture system, but not to the remaining elements of the coating line, such as raw material usage).

- (2) Exemption for backup utility power emissions units. The requirements of this part shall not apply to a utility unit, as defined in §72.2 of this chapter, that is municipally-owned if the owner or operator provides documentation in a part 70 or 71 permit application that:
- (i) The utility unit is exempt from all monitoring requirements in part 75 (including the appendices thereto) of this chapter;
- (ii) The utility unit is operated for the sole purpose of providing electricity during periods of peak electrical demand or emergency situations and will be operated consistent with that purpose throughout the part 70 or 71 permit term. The owner or operator shall provide historical operating data and relevant contractual obligations to document that this criterion is satisfied; and
- (iii) The actual emissions from the utility unit, based on the average annual emissions over the last three calendar years of operation (or such shorter time period that is available for units with fewer than three years of operation) are less than 50 percent of the amount in tons per year required for a source to be classified as a major source and are expected to remain so.

§ 64.3 Monitoring design criteria.

- (a) General criteria. To provide a reasonable assurance of compliance with emission limitations or standards for the anticipated range of operations at a pollutant-specific emissions unit, monitoring under this part shall meet the following general criteria:
- (1) The owner or operator shall design the monitoring to obtain data for one or more indicators of emission control performance for the control device, any associated capture system and, if necessary to satisfy paragraph (a)(2) of this section, processes at a pol-

lutant-specific emissions unit. Indicators of performance may include, but are not limited to, direct or predicted emissions (including visible emissions or opacity), process and control device parameters that affect control device (and capture system) efficiency or emission rates, or recorded findings of inspection and maintenance activities conducted by the owner or operator.

- (2) The owner or operator shall establish an appropriate range(s) or designated condition(s) for the selected indicator(s) such that operation within the ranges provides a reasonable assurance of ongoing compliance with emission limitations or standards for the anticipated range of operating conditions. Such range(s) or condition(s) shall reflect the proper operation and maintenance of the control device (and associated capture system), in accordance with applicable design properties, for minimizing emissions over the anticipated range of operating conditions at least to the level required to achieve compliance with the applicable requirements. The reasonable assurance of compliance will be assessed by maintaining performance within the indicator range(s) or designated condition(s). The ranges shall be established in accordance with the design and performance requirements in this section and documented in accordance with the requirements in §64.4. If necessary to assure that the control device and associated capture system can satisfy this criterion, the owner or operator shall monitor appropriate process operational parameters (such as total throughput where necessary to stay within the rated capacity for a control device). In addition, unless specifically stated otherwise by an applicable requirement, the owner or operator shall monitor indicators to detect any bypass of the control device (or capture system) to the atmosphere, if such bypass can occur based on the design of the pollutant-specific emissions unit.
- (3) The design of indicator ranges or designated conditions may be:
- (i) Based on a single maximum or minimum value if appropriate (e.g., maintaining condenser temperatures a certain number of degrees below the condensation temperature of the applicable compound(s) being processed) or

at multiple levels that are relevant to distinctly different operating conditions (e.g., high versus low load levels).

- (ii) Expressed as a function of process variables (e.g., an indicator range expressed as minimum to maximum pressure drop across a venturi throat in a particulate control scrubber).
- (iii) Expressed as maintaining the applicable parameter in a particular operational status or designated condition (e.g., position of a damper controlling gas flow to the atmosphere through a by-pass duct).
- (iv) Established as interdependent between more than one indicator.
- (b) *Performance criteria*. The owner or operator shall design the monitoring to meet the following performance criteria:
- (1) Specifications that provide for obtaining data that are representative of the emissions or parameters being monitored (such as detector location and installation specifications, if applicable).
- (2) For new or modified monitoring equipment, verification procedures to confirm the operational status of the monitoring prior to the date by which the owner or operator must conduct monitoring under this part as specified in §64.7(a). The owner or operator shall consider the monitoring equipment manufacturer's requirements or recommendations for installation, calibration, and start-up operation.
- (3) Quality assurance and control practices that are adequate to ensure the continuing validity of the data. The owner or operator shall consider manufacturer recommendations or requirements applicable to the monitoring in developing appropriate quality assurance and control practices.
- (4) Specifications for the frequency of conducting the monitoring, the data collection procedures that will be used (e.g., computerized data acquisition and handling, alarm sensor, or manual log entries based on gauge readings), and, if applicable, the period over which discrete data points will be averaged for the purpose of determining whether an excursion or exceedance has occurred.
- (i) At a minimum, the owner or operator shall design the period over which data are obtained and, if applicable,

- averaged consistent with the characteristics and typical variability of the pollutant-specific emissions unit (including the control device and associated capture system). Such intervals shall be commensurate with the time period over which a change in control device performance that would require actions by owner or operator to return operations within normal ranges or designated conditions is likely to be observed.
- (ii) For all pollutant-specific emissions units with the potential to emit, calculated including the effect of control devices, the applicable regulated air pollutant in an amount equal to or greater than 100 percent of the amount, in tons per year, required for a source to be classified as a major source, for each parameter monitored, the owner or operator shall collect four or more data values equally spaced over each hour and average the values, as applicable, over the applicable averaging period as determined in accordance with paragraph (b)(4)(i) of this section. The permitting authority may approve a reduced data collection frequency, if appropriate, based on information presented by the owner or operator concerning the data collection mechanisms available for a particular parameter for the particular pollutant-specific emissions unit (e.g., integrated raw material or fuel analysis data, noninstrumental measurement waste feed rate or visible emissions, use of a portable analyzer or an alarm
- (iii) For other pollutant-specific emissions units, the frequency of data collection may be less than the frequency specified in paragraph (b)(4)(ii) of this section but the monitoring shall include some data collection at least once per 24-hour period (e.g., a daily inspection of a carbon adsorber operation in conjunction with a weekly or monthly check of emissions with a portable analyzer).
- (c) Evaluation factors. In designing monitoring to meet the requirements in paragraphs (a) and (b) of this section, the owner or operator shall take into account site-specific factors including the applicability of existing monitoring equipment and procedures,

§ 64.4

the ability of the monitoring to account for process and control device operational variability, the reliability and latitude built into the control technology, and the level of actual emissions relative to the compliance limitation.

- (d) Special criteria for the use of continuous emission, opacity or predictive monitoring systems. (1) If a continuous emission monitoring system (CEMS), continuous opacity monitoring system (COMS) or predictive emission monitoring system (PEMS) is required pursuant to other authority under the Act or state or local law, the owner or operator shall use such system to satisfy the requirements of this part.
- (2) The use of a CEMS, COMS, or PEMS that satisfies any of the following monitoring requirements shall be deemed to satisfy the general design criteria in paragraphs (a) and (b) of this section, provided that a COMS may be subject to the criteria for establishing indicator ranges under paragraph (a) of this section:
- (i) Section 51.214 and appendix P of part 51 of this chapter;
- (ii) Section 60.13 and appendix B of part 60 of this chapter;
- (iii) Section 63.8 and any applicable performance specifications required pursuant to the applicable subpart of part 63 of this chapter;
 - (iv) Part 75 of this chapter;
- (v) Subpart H and appendix IX of part 266 of this chapter; or
- (vi) If an applicable requirement does not otherwise require compliance with the requirements listed in the preceding paragraphs (d)(2)(i) through (v) of this section, comparable requirements and specifications established by the permitting authority.
- (3) The owner or operator shall design the monitoring system subject to this paragraph (d) to:
- (i) Allow for reporting of exceedances (or excursions if applicable to a COMS used to assure compliance with a particulate matter standard), consistent with any period for reporting of exceedances in an underlying requirement. If an underlying requirement does not contain a provision for establishing an averaging period for the reporting of exceedances or excursions, the criteria used to develop an aver-

aging period in (b)(4) of this section shall apply; and

(ii) Provide an indicator range consistent with paragraph (a) of this section for a COMS used to assure compliance with a particulate matter standard. If an opacity standard applies to the pollutant-specific emissions unit, such limit may be used as the appropriate indicator range unless the opacity limit fails to meet the criteria in paragraph (a) of this section after considering the type of control device and other site-specific factors applicable to the pollutant-specific emissions unit.

§ 64.4 Submittal requirements.

- (a) The owner or operator shall submit to the permitting authority monitoring that satisfies the design requirements in §64.3. The submission shall include the following information:
- (1) The indicators to be monitored to satisfy $\S 64.3(a)(1)-(2)$;
- (2) The ranges or designated conditions for such indicators, or the process by which such indicator ranges or designated conditions shall be established;
- (3) The performance criteria for the monitoring to satisfy §64.3(b); and
- (4) If applicable, the indicator ranges and performance criteria for a CEMS, COMS or PEMS pursuant to §64.3(d).
- (b) As part of the information submitted, the owner or operator shall submit a justification for the proposed elements of the monitoring. If the performance specifications proposed to satisfy §64.3(b)(2) or (3) include differences from manufacturer recommendations, the owner or operator shall explain the reasons for the differences between the requirements proposed by the owner or operator and the manufacturer's recommendations or requirements. The owner or operator also shall submit any data supporting the justification, and may refer to generally available sources of information used to support the justification (such as generally available air pollution engineering manuals, or EPA or permitting authority publications on appropriate monitoring for various types of control devices or capture systems). To justify the appropriateness of the monitoring elements proposed, the owner

§ 64.4

or operator may rely in part on existing applicable requirements that establish the monitoring for the applicable pollutant-specific emissions unit or a similar unit. If an owner or operator relies on presumptively acceptable monitoring, no further justification for the appropriateness of that monitoring should be necessary other than an explanation of the applicability of such monitoring to the unit in question, unless data or information is brought forward to rebut the assumption. Presumptively acceptable monitoring includes:

- (1) Presumptively acceptable or required monitoring approaches, established by the permitting authority in a rule that constitutes part of the applicable implementation plan required pursuant to title I of the Act, that are designed to achieve compliance with this part for particular pollutant-specific emissions units:
- (2) Continuous emission, opacity or predictive emission monitoring systems that satisfy applicable monitoring requirements and performance specifications as specified in §64.3(d);
- (3) Excepted or alternative monitoring methods allowed or approved pursuant to part 75 of this chapter;
- (4) Monitoring included for standards exempt from this part pursuant to §64.2(b)(1)(i) or (vi) to the extent such monitoring is applicable to the performance of the control device (and associated capture system) for the pollutant-specific emissions unit; and
- (5) Presumptively acceptable monitoring identified in guidance by EPA. Such guidance will address the requirements under §§ 64.4(a), (b), and (c) to the extent practicable.
- (c)(1) Except as provided in paragraph (d) of this section, the owner or operator shall submit control device (and process and capture system, if applicable) operating parameter data obtained during the conduct of the applicable compliance or performance test conducted under conditions specified by the applicable rule. If the applicable rule does not specify testing conditions or only partially specifies test conditions, the performance test generally shall be conducted under conditions representative of maximum emissions potential under anticipated operating

conditions at the pollutant-specific emissions unit. Such data may be supplemented, if desired, by engineering assessments and manufacturer's recommendations to justify the indicator ranges (or, if applicable, the procedures for establishing such indicator ranges). Emission testing is not required to be conducted over the entire indicator range or range of potential emissions.

- (2) The owner or operator must document that no changes to the pollutant-specific emissions unit, including the control device and capture system, have taken place that could result in a significant change in the control system performance or the selected ranges or designated conditions for the indicators to be monitored since the performance or compliance tests were conducted.
- (d) If existing data from unit-specific compliance or performance testing specified in paragraph (c) of this section are not available, the owner or operator:
- (1) Shall submit a test plan and schedule for obtaining such data in accordance with paragraph (e) of this section; or
- (2) May submit indicator ranges (or procedures for establishing indicator ranges) that rely on engineering assessments and other data, provided that the owner or operator demonstrates that factors specific to the type of monitoring, control device, or pollutant-specific emissions unit make compliance or performance testing unnecessary to establish indicator ranges at levels that satisfy the criteria in §64.3(a).
- (e) If the monitoring submitted by the owner or operator requires installation, testing, or other necessary activities prior to use of the monitoring for purposes of this part, the owner or operator shall include an implementation plan and schedule for installing, testing and performing any other appropriate activities prior to use of the monitoring. The implementation plan and schedule shall provide for use of the monitoring as expeditiously as practicable after approval of the monitoring in the part 70 or 71 permit pursuant to §64.6, but in no case shall the schedule for completing installation

§ 64.5

and beginning operation of the monitoring exceed 180 days after approval of the permit.

- (f) If a control device is common to more than one pollutant-specific emissions unit, the owner or operator may submit monitoring for the control device and identify the pollutant-specific emissions units affected and any process or associated capture device conditions that must be maintained or monitored in accordance with §64.3(a) rather than submit separate monitoring for each pollutant-specific emissions unit.
- (g) If a single pollutant-specific emissions unit is controlled by more than one control device similar in design and operation, the owner or operator may submit monitoring that applies to all the control devices and identify the control devices affected and any process or associated capture device conditions that must be maintained or monitored in accordance with §64.3(a) rather than submit a separate description of monitoring for each control device.

§ 64.5 Deadlines for submittals.

- (a) Large pollutant-specific emissions units. For all pollutant-specific emissions units with the potential to emit (taking into account control devices to the extent appropriate under the definition of this term in §64.1) the applicable regulated air pollutant in an amount equal to or greater than 100 percent of the amount, in tons per year, required for a source to be classified as a major source, the owner or operator shall submit the information required under §64.4 at the following times:
- (1) On or after April 20, 1998, the owner or operator shall submit information as part of an application for an initial part 70 or 71 permit if, by that date, the application either:
 - (i) Has not been filed; or
- (ii) Has not yet been determined to be complete by the permitting authority.
- (2) On or after April 20, 1998, the owner or operator shall submit information as part of an application for a significant permit revision under part 70 or 71 of this chapter, but only with respect to those pollutant-specific emissions units for which the proposed permit revision is applicable.

- (3) The owner or operator shall submit any information not submitted under the deadlines set forth in paragraphs (a)(1) and (2) of this section as part of the application for the renewal of a part 70 or 71 permit.
- (b) Other pollutant-specific emissions units. For all other pollutant-specific emissions units subject to this part and not subject to §64.5(a), the owner or operator shall submit the information required under §64.4 as part of an application for a renewal of a part 70 or 71 permit.
- (c) The effective date for the requirement to submit information under §64.4 shall be as specified pursuant to paragraphs (a)-(b) of this section and a permit reopening to require the submittal of information under this section shall not be required pursuant to §70.7(f)(1)(i) of this chapter, provided, however, that, if a part 70 or 71 permit is reopened for cause by EPA or the permitting authority pursuant to \$70.7(f)(1)(iii) or (iv), or \$71.7(f) or (g), the applicable agency may require the submittal of information under this section for those pollutant-specific emissions units that are subject to this part and that are affected by the permit reopening.
- (d) Prior to approval of monitoring that satisfies this part, the owner or operator is subject to the requirements of §70.6(a)(3)(i)(B).

§64.6 Approval of monitoring.

- (a) Based on an application that includes the information submitted in accordance with §64.5, the permitting authority shall act to approve the monitoring submitted by the owner or operator by confirming that the monitoring satisfies the requirements in §64.3.
- (b) In approving monitoring under this section, the permitting authority may condition the approval on the owner or operator collecting additional data on the indicators to be monitored for a pollutant-specific emissions unit, including required compliance or performance testing, to confirm the ability of the monitoring to provide data that are sufficient to satisfy the requirements of this part and to confirm the appropriateness of an indicator

§ 64.7

range(s) or designated condition(s) proposed to satisfy §64.3(a)(2) and (3) and consistent with the schedule in §64.4(e).

- (c) If the permitting authority approves the proposed monitoring, the permitting authority shall establish one or more permit terms or conditions that specify the required monitoring in accordance with $\S70.6(a)(3)(i)$ of this chapter. At a minimum, the permit shall specify:
- (1) The approved monitoring approach that includes all of the fol-
- (i) The indicator(s) to be monitored (such as temperature, pressure drop, emissions, or similar parameter);
- (ii) The means or device to be used to measure the indicator(s) (such as temperature measurement device, visual observation, or CEMS); and
- (iii) The performance requirements established to satisfy §64.3(b) or (d), as applicable.
- (2) The means by which the owner or operator will define an exceedance or excursion for purposes of responding to and reporting exceedances or excursions under §§64.7 and 64.8 of this part. The permit shall specify the level at which an excursion or exceedance will be deemed to occur, including the appropriate averaging period associated with such exceedance or excursion. For defining an excursion from an indicator range or designated condition, the permit may either include the specific value(s) or condition(s) at which an excursion shall occur, or the specific procedures that will be used to establish that value or condition. If the latter, the permit shall specify appropriate notice procedures for the owner or operator to notify the permitting authority upon any establishment or reestablishment of the value.
- (3) The obligation to conduct the monitoring and fulfill the other obligations specified in §§ 64.7 through 64.9 of
- (4) If appropriate, a minimum data availability requirement for valid data collection for each averaging period, and, if appropriate, a minimum data availability requirement for the averaging periods in a reporting period.
- (d) If the monitoring proposed by the owner or operator requires installation, testing or final verification of

operational status, the part 70 or 71 permit shall include an enforceable schedule with appropriate milestones for completing such installation, testing, or final verification consistent with the requirements in §64.4(e).

Filed: 02/21/2013

- (e) If the permitting authority disapproves the proposed monitoring, the following applies:
- (1) The draft or final permit shall include, at a minimum, monitoring that satisfies requirements the $\S70.6(a)(3)(i)(B);$
- (2) The permitting authority shall include in the draft or final permit a compliance schedule for the source owner to submit monitoring that satisfies §§ 64.3 and 64.4, but in no case shall the owner or operator submit revised monitoring more than 180 days from the date of issuance of the draft or final permit; and
- (3) If the source owner or operator does not submit the monitoring in accordance with the compliance schedule as required in paragraph (e)(2) of this section or if the permitting authority disapproves the monitoring submitted, the source owner or operator shall be deemed not in compliance with part 64, unless the source owner or operator successfully challenges the approval.

§64.7 Operation of approved moni-

- (a) Commencement of operation. The owner or operator shall conduct the monitoring required under this part upon issuance of a part 70 or 71 permit that includes such monitoring, or by such later date specified in the permit pursuant to §64.6(d).
- (b) Proper maintenance. At all times, the owner or operator shall maintain the monitoring, including but not limited to, maintaining necessary parts for routine repairs of the monitoring equipment.
- (c) Continued operation. Except for, as applicable, monitoring malfunctions, associated repairs, and required quality assurance or control activities (including, as applicable, calibration checks and required zero and span adjustments), the owner or operator shall conduct all monitoring in continuous operation (or shall collect data at all required intervals) at all times that

§ 64.8

the pollutant-specific emissions unit is operating. Data recorded during monitoring malfunctions, associated repairs, and required quality assurance or control activities shall not be used for purposes of this part, including data averages and calculations, or fulfilling a minimum data availability requirement, if applicable. The owner or operator shall use all the data collected during all other periods in assessing the operation of the control device and associated control system. A monitoring malfunction is any sudden, infrequent, not reasonably preventable failure of the monitoring to provide valid data. Monitoring failures that are caused in part by poor maintenance or careless operation are not malfunc-

(d) Response to excursions exceedances. (1) Upon detecting an excursion or exceedance, the owner or operator shall restore operation of the pollutant-specific emissions unit (including the control device and associated capture system) to its normal or usual manner of operation as expeditiously as practicable in accordance with good air pollution control practices for minimizing emissions. The response shall include minimizing the period of any startup, shutdown or malfunction and taking any necessary corrective actions to restore normal operation and prevent the likely recurrence of the cause of an excursion or exceedance (other than those caused by excused startup or shutdown conditions). Such actions may include initial inspection and evaluation, recording that operations returned to normal without operator action (such as through response by a computerized distribution control system), or any necessary follow-up actions to return operation to within the indicator range, designated condition, or below the applicable emission limitation or standard, as applicable.

(2) Determination of whether the owner or operator has used acceptable procedures in response to an excursion or exceedance will be based on information available, which may include but is not limited to, monitoring results, review of operation and maintenance procedures and records, and inspection

of the control device, associated capture system, and the process.

(e) Documentation of need for improved monitoring. After approval of monitoring under this part, if the owner or operator identifies a failure to achieve compliance with an emission limitation or standard for which the approved monitoring did not provide an indication of an excursion or exceedance while providing valid data, or the results of compliance or performance testing document a need to modify the existing indicator ranges or designated conditions, the owner or operator shall promptly notify the permitting authority and, if necessary, submit a proposed modification to the part 70 or 71 permit to address the necessary monitoring changes. Such a modification may include, but is not limited to, reestablishing indicator ranges or designated conditions, modifying the frequency of conducting monitoring and collecting data, or the monitoring of additional parameters.

§64.8 Quality improvement plan (QIP) requirements.

- (a) Based on the results of a determination made under §64.7(d)(2), the Administrator or the permitting authority may require the owner or operator to develop and implement a QIP. Consistent with §64.6(c)(3), the part 70 or 71 permit may specify an appropriate threshold, such as an accumulation of exceedances or excursions exceeding 5 percent duration of a pollutant-specific emissions unit's operating time for a reporting period, for requiring the implementation of a QIP. The threshold may be set at a higher or lower percent or may rely on other criteria for purposes of indicating whether a pollutant-specific emissions unit is being maintained and operated in a manner consistent with good air pollution control practices.
- (b) Elements of a QIP:
- (1) The owner or operator shall maintain a written QIP, if required, and have it available for inspection.
- (2) The plan initially shall include procedures for evaluating the control performance problems and, based on the results of the evaluation procedures, the owner or operator shall modify the plan to include procedures for

§ 64.10

conducting one or more of the following actions, as appropriate:

- (i) Improved preventive maintenance practices.
 - (ii) Process operation changes.
- (iii) Appropriate improvements to control methods.
- (iv) Other steps appropriate to correct control performance.
- (v) More frequent or improved monitoring (only in conjunction with one or more steps under paragraphs (b)(2)(i) through (iv) of this section).
- (c) If a QIP is required, the owner or operator shall develop and implement a QIP as expeditiously as practicable and shall notify the permitting authority if the period for completing the improvements contained in the QIP exceeds 180 days from the date on which the need to implement the QIP was determined.
- (d) Following implementation of a QIP, upon any subsequent determination pursuant to §64.7(d)(2) the Administrator or the permitting authority may require that an owner or operator make reasonable changes to the QIP if the QIP is found to have:
- (1) Failed to address the cause of the control device performance problems;
- (2) Failed to provide adequate procedures for correcting control device performance problems as expeditiously as practicable in accordance with good air pollution control practices for minimizing emissions.
- (e) Implementation of a QIP shall not excuse the owner or operator of a source from compliance with any existing emission limitation or standard, or any existing monitoring, testing, reporting or recordkeeping requirement that may apply under federal, state, or local law, or any other applicable requirements under the Act.

§64.9 Reporting and recordkeeping requirements.

(a) General reporting requirements. (1) On and after the date specified in §64.7(a) by which the owner or operator must use monitoring that meets the requirements of this part, the owner or operator shall submit monitoring reports to the permitting authority in accordance with §70.6(a)(3)(iii) of this chapter.

(2) A report for monitoring under this part shall include, at a minimum, the information required §70.6(a)(3)(iii) of this chapter and the following information, as applicable:

Filed: 02/21/2013

- (i) Summary information on the number, duration and cause (including unknown cause, if applicable) of excursions or exceedances, as applicable, and the corrective actions taken:
- (ii) Summary information on the number, duration and cause (including unknown cause, if applicable) for monitor downtime incidents (other than downtime associated with zero and span or other daily calibration checks, if applicable); and
- (iii) A description of the actions taken to implement a QIP during the reporting period as specified in §64.8. Upon completion of a QIP, the owner or operator shall include in the next summary report documentation that the implementation of the plan has been completed and reduced the likelihood of similar levels of excursions or exceedances occurring.
- (b) General recordkeeping requirements. (1) The owner or operator shall comply with the recordkeeping requirements specified in §70.6(a)(3)(ii) of this chapter. The owner or operator shall maintain records of monitoring data, monitor performance data, corrective actions taken, any written quality improvement plan required pursuant to §64.8 and any activities undertaken to implement a quality improvement plan, and other supporting information required to be maintained under this part (such as data used to document the adequacy of monitoring, or records of monitoring maintenance or corrective actions).
- (2) Instead of paper records, the owner or operator may maintain records on alternative media, such as microfilm, computer files, magnetic tape disks, or microfiche, provided that the use of such alternative media allows for expeditious inspection and review, and does not conflict with other applicable recordkeeping requirements.

§ 64.10 Savings provisions.

- (a) Nothing in this part shall:
- (1) Excuse the owner or operator of a source from compliance with any existing emission limitation or standard, or

Pt. 65

any existing monitoring, testing, reporting or recordkeeping requirement that may apply under federal, state, or local law, or any other applicable requirements under the Act. The requirements of this part shall not be used to justify the approval of monitoring less stringent than the monitoring which is required under separate legal authority and are not intended to establish minimum requirements for the purpose of determining the monitoring to be imposed under separate authority under the Act, including monitoring in permits issued pursuant to title I of the Act. The purpose of this part is to require, as part of the issuance of a permit under title V of the Act, improved or new monitoring at those emissions units where monitoring requirements do not exist or are inadequate to meet the requirements of this part.

- (2) Restrict or abrogate the authority of the Administrator or the permitting authority to impose additional or more stringent monitoring, recordkeeping, testing, or reporting requirements on any owner or operator of a source under any provision of the Act, including but not limited to sections 114(a)(1) and 504(b), or state law, as applicable.
- (3) Restrict or abrogate the authority of the Administrator or permitting authority to take any enforcement action under the Act for any violation of an applicable requirement or of any person to take action under section 304 of the Act.

PART 65—CONSOLIDATED FEDERAL **AIR RULE**

Subpart A—General Provisions

Sec.

- 65.1 Applicability.
- Definitions.
- 65.3 Compliance with standards and operation and maintenance requirements.
- 65.4 Recordkeeping.
- 65.5 Reporting requirements.
- 65.6 Startup, shutdown, and malfunction plan and procedures. 65.7 Monitoring, recordkeeping, and report-
- ing waivers and alternatives, and alternative work practice for equipment leaks.
- 65.8 Procedures for approval of alternative means of emission limitation.
- 65.9 Availability of information and confidentiality.

40 CFR Ch. I (7-1-12 Edition)

- 65.10 State authority.
- 65.11 Circumvention and prohibited activities.
- 65.12 Delegation of authority.
- 65.13 Incorporation by reference.
- 65.14 Addresses.
- 65.15-65.19 [Reserved]
- Table 1 to Subpart A of Part 65—Applica-BLE 40 CFR PARTS 60, 61, AND 63 GENERAL **PROVISIONS**
- Table 2 to Subpart A of Part 65—Applica-BLE REFERENCING SUBPART PROVISIONS
- Table 3 to Subpart A of Part 65—Detec-TION SENSITIVITY LEVELS (GRAMS PER HOUR.)

Subpart B [Reserved]

Subpart C—Storage Vessels

- 65.40Applicability.
- 65.41 Definitions.
- Control requirements.
- 65.43 Fixed roof with an internal floating roof (IFR).
- 65.44 External floating roof (EFR).
- 65.45 External floating roof converted into an internal floating roof.
- 65.46 Alternative means of emission limitation.
- 65.47 Recordkeeping provisions.
- 65.48 Reporting provisions.
- 65.49-65.59 [Reserved]

Subpart D—Process Vents

- 65.60 Applicability.
- 65.61 Definitions.
- Process vent group determination.
- 65.63 Performance and group status change requirements.
- 65.64 Group determination procedures.
- 65.65 Monitoring.
- 65.66 Recordkeeping provisions.
- 65.67 Reporting provisions.
- 65.68-65.79 [Reserved]
- TABLE 1 TO SUBPART D OF PART 65-CON-CENTRATION FOR GROUP DETERMINATION
- Table 2 to Subpart D of Part 65—TRE Pa-RAMETERS FOR NSPS REFERENCING SUB-PARTS
- TABLE 3 TO SUBPART D OF PART 65-TRE PA-RAMETERS FOR HON REFERENCING SUB-

Subpart E—Transfer Racks

- 65.80 Applicability.
- 65.81 Definitions.
- 65.82 Design requirements.
- 65.83 Performance requirements.
- Operating requirements. 65.84
- 65.85 Procedures.
- 65.86 Monitoring.
- 65.87 Recordkeeping provisions.
- 65.88-65.99 [Reserved]

§ 70.6

- (v) The compliance plan content requirements specified in this paragraph shall apply and be included in the acid rain portion of a compliance plan for an affected source, except as specifically superseded by regulations promulgated under title IV of the Act with regard to the schedule and method(s) the source will use to achieve compliance with the acid rain emissions limi-
- (9) Requirements for compliance certification, including the following:
- (i) A certification of compliance with all applicable requirements by a responsible official consistent with paragraph (d) of this section and section 114(a)(3) of the Act;
- (ii) A statement of methods used for determining compliance, including a description of monitoring, recordkeeping, and reporting requirements and test methods;
- (iii) A schedule for submission of compliance certifications during the permit term, to be submitted no less frequently than annually, or more frequently if specified by the underlying applicable requirement or by the permitting authority; and
- (iv) A statement indicating the source's compliance status with any applicable enhanced monitoring and compliance certification requirements of the Act.
- (10) The use of nationally-standardized forms for acid rain portions of permit applications and compliance plans, as required by regulations promulgated under title IV of the Act.
- (d) Any application form, report, or compliance certification submitted pursuant to these regulations shall contain certification by a responsible official of truth, accuracy, and completeness. This certification and any other certification required under this part shall state that, based on information and belief formed after reasonable inquiry, the statements and information in the document are true, accurate, and complete.

[57 FR 32295, July 21, 1992, as amended at 74 FR 51438, Oct. 6, 20091

§ 70.6 Permit content.

(a) Standard permit requirements. Each permit issued under this part shall include the following elements:

- (1) Emissions limitations and standards, including those operational requirements and limitations that assure compliance with all applicable requirements at the time of permit issuance. Such requirements and limitations may include ARMs identified by the source in its part 70 permit application as approved by the permitting authority, provided that no ARM shall contravene any terms needed to comply with any otherwise applicable requirement or requirement of this part or circumvent any applicable requirement that would apply as a result of implementing the ARM.
- (i) The permit shall specify and reference the origin of and authority for each term or condition, and identify any difference in form as compared to the applicable requirement upon which the term or condition is based.
- (ii) The permit shall state that, where an applicable requirement of the Act is more stringent than an applicable requirement of regulations promulgated under title IV of the Act, both provisions shall be incorporated into the permit and shall be enforceable by the Administrator.
- (iii) If an applicable implementation plan allows a determination of an alternative emission limit at a part 70 source, equivalent to that contained in the plan, to be made in the permit issuance, renewal, or significant modification process, and the State elects to use such process, any permit containing such equivalency determination shall contain provisions to ensure that any resulting emissions limit has been demonstrated to be quantifiable, accountable, enforceable, and based on replicable procedures.
- (2) Permit duration. The permitting authority shall issue permits for a fixed term of 5 years in the case of affected sources, and for a term not to exceed 5 years in the case of all other sources. Notwithstanding this requirement, the permitting authority shall issue permits for solid waste incineration units combusting municipal waste subject to standards under section 129(e) of the Act for a period not to exceed 12 years and shall review such permits at least every 5 years.
- (3) Monitoring and related recordkeeping and reporting requirements. (i)

40 C.F.R. § 70.5(d)

under the Act or under any of its applicable requirements. Terms and conditions so designated are not subject to the requirements of §§ 70.7, 70.8, or of this part, other than those contained in this paragraph (b) of this section.

- (c) Compliance requirements. All part 70 permits shall contain the following elements with respect to compliance:
- (1) Consistent with paragraph (a)(3) of this section, compliance certification, testing, monitoring, reporting, and recordkeeping requirements sufficient to assure compliance with the terms and conditions of the permit. Any document (including reports) required by a part 70 permit shall contain a certification by a responsible official that meets the requirements of §70.5(d) for this part.
- (2) Inspection and entry requirements that require that, upon presentation of credentials and other documents as may be required by law, the permittee shall allow the permitting authority or an authorized representative to perform the following:
- (i) Enter upon the permittee's premises where a part 70 source is located or emissions-related activity is ducted, or where records must be kept under the conditions of the permit;
- (ii) Have access to and copy, at reasonable times, any records that must be kept under the conditions of the permit;
- (iii) Inspect at reasonable times any facilities, equipment (including monitoring and air pollution control equipment), practices, or operations regulated or required under the permit; and
- (iv) As authorized by the Act, sample or monitor at reasonable times substances or parameters for the purpose of assuring compliance with the permit or applicable requirements.
- (3) A schedule of compliance consistent with §70.5(c)(8) of this part.
- (4) Progress reports consistent with an applicable schedule of compliance and §70.5(c)(8) of this part to be submitted at least semiannually, or at a more frequent period if specified in the applicable requirement or by the permitting authority. Such progress reports shall contain the following:
- (i) Dates for achieving the activities, milestones, or compliance required in the schedule of compliance, and dates

when such activities, milestones or compliance were achieved; and

- (ii) An explanation of why any dates in the schedule of compliance were not or will not be met, and any preventive or corrective measures adopted.
- (5) Requirements for compliance certification with terms and conditions contained in the permit, including emission limitations, standards, or work practices. Permits shall include each of the following:
- (i) The frequency (not less than annually or such more frequent periods as specified in the applicable requirement or by the permitting authority) of submissions of compliance certifications;
- (ii) In accordance with §70.6(a)(3) of this part, a means for monitoring the compliance of the source with its emissions limitations, standards, and work practices:
- (iii) A requirement that the compliance certification include all of the following (provided that the identification of applicable information may cross-reference the permit or previous reports, as applicable):
- (A) The identification of each term or condition of the permit that is the basis of the certification:
- (B) The identification of the method(s) or other means used by the owner or operator for determining the compliance status with each term and condition during the certification period. Such methods and other means shall include, at a minimum, the methods and means required under paragraph (a)(3) of this section;
- (C) The status of compliance with the terms and conditions of the permit for the period covered by the certification, including whether compliance during the period was continuous or intermittent. The certification shall be based on the method or means designated in paragraph (c)(5)(iii)(B) of this section. The certification shall identify each deviation and take it into account in the compliance certification. The certification shall also identify as possible exceptions to compliance any periods during which compliance is required and in which an excursion or exceedance as defined under part 64 of this chapter occurred; and

40 C.F.R. § 70.6(c)(5)

- (D) Such other facts as the permitting authority may require to determine the compliance status of the source.
- (iv) A requirement that all compliance certifications be submitted to the Administrator as well as to the permitting authority.
- (6) Such other provisions as the permitting authority may require.
- (d) General permits. (1) The permitting authority may, after notice and opportunity for public participation provided under §70.7(h) of this part, issue a general permit covering numerous similar sources. Any general permit shall comply with all requirements applicable to other part 70 permits and shall identify criteria by which sources may qualify for the general permit. To sources that qualify, the permitting authority shall grant the conditions and terms of the general permit. Notwithstanding the shield provisions of paragraph (f) of this section, the source shall be subject to enforcement action for operation without a part 70 permit if the source is later determined not to qualify for the conditions and terms of the general permit. General permits shall not be authorized for affected sources under the acid rain program unless otherwise provided in regulations promulgated under title IV of the Act.
- (2) Part 70 sources that would qualify for a general permit must apply to the permitting authority for coverage under the terms of the general permit or must apply for a part 70 permit consistent with §70.5 of this part. The permitting authority may, in the general permit, provide for applications which deviate from the requirements of §70.5 of this part, provided that such applications meet the requirements of title V of the Act, and include all information necessary to determine qualification for, and to assure compliance with, the general permit. Without repeating the public participation procedures required under §70.7(h) of this part, the permitting authority may grant a source's request for authorization to operate under a general permit, but such a grant shall not be a final permit action for purposes of judicial
- (e) Temporary sources. The permitting authority may issue a single permit

- authorizing emissions from similar operations by the same source owner or operator at multiple temporary locations. The operation must be temporary and involve at least one change of location during the term of the permit. No affected source shall be permitted as a temporary source. Permits for temporary sources shall include the following:
- (1) Conditions that will assure compliance with all applicable requirements at all authorized locations;
- (2) Requirements that the owner or operator notify the permitting authority at least 10 days in advance of each change in location; and
- (3) Conditions that assure compliance with all other provisions of this section.
- (f) Permit shield. (1) Except as provided in this part, the permitting authority may expressly include in a part 70 permit a provision stating that compliance with the conditions of the permit shall be deemed compliance with any applicable requirements as of the date of permit issuance, provided that:
- (i) Such applicable requirements are included and are specifically identified in the permit; or
- (ii) The permitting authority, in acting on the permit application or revision, determines in writing that other requirements specifically identified are not applicable to the source, and the permit includes the determination or a concise summary thereof.
- (2) A part 70 permit that does not expressly state that a permit shield exists shall be presumed not to provide such a shield.
- (3) Nothing in this paragraph or in any part 70 permit shall alter or affect the following:
- (i) The provisions of section 303 of the Act (emergency orders), including the authority of the Administrator under that section;
- (ii) The liability of an owner or operator of a source for any violation of applicable requirements prior to or at the time of permit issuance;
- (iii) The applicable requirements of the acid rain program, consistent with section 408(a) of the Act; or
- (iv) The ability of EPA to obtain information from a source pursuant to section 114 of the Act.

CERTIFICATE OF SERVICE

I hereby certify that, on this 21st day of February 2013, a copy of the Joint Brief of Industry Intervenors in Response to Environmental Petitioners was served electronically through the Court's CM/ECF system on all ECF-registered counsel.

/s/ Lauren E. Freeman